

Universitá degli studi di catania

DIPARTIMENTO DI INGEGNERIA ELETTRICA ELETTRONICA ED INFORMATICA

Dottorato di Ricerca in Ingegneria dei Sistemi, Energetica, Informatica e

delle Telecomunicazioni

Cycle XXXIII

Intentional and Unintentional Light Emission Diode in

Power Electronics applications

DOCTORAL THESIS

Author

Giovanni Susinni

Tutor

Prof. Ing. Santi Agatino Rizzo

Coordinator of the Ph.D. Program

Prof. Ing. Paolo Arena

Academic Year 2019-2020

vi

Acknowledgments

This dissertation is the result of almost three years of work.

I wish to express my sincere thanks to Prof. Angelo Raciti, for his support and guidance during my Ph.D.

I am truly grateful to Prof. Francesco Iannuzzo for supporting me and for having taken me in and made me feel so well during my research experience abroad at the Department of Energy Technology, University of Aalborg.

With a deep sense of gratefulness, I wish to express my sincere thanks to my supervisor Prof. Santi Agatino Rizzo, for his immense help in planning and executing my research work, and for the lessons and insights on the workings of scientific research.

Catania, 1st September 2020

Introduction

The phenomenon of light emission in a P-N junction is due to the recombination of electrons and holes in the junction. This light emission phenomenon has been widely used in power electronics applications. Among them, it is worth remembering the use of the light-emitting diode (LED) as a source of light in lighting system applications. In this perspective, LEDs technology has been widely studied as an "Intentional" source of light in Power Electronics.

In such a context, a LED light bulb can be assumed as a combination of LED semiconductor materials and a ballast circuit, where the LED driver circuit plays a fundamental role from a power quality point of view. In fact, the driver controls the input current of the LED light bulb that must be compliant with the standards and regulations concerning currents harmonic. Furthermore, the whole lighting system must meet the standards. In this perspective, a sufficient power factor (PF) must be guarantee thanks to a proper design of the lighting system, especially when a very huge number of lamps are used. Hence, the electrical model of each light bulb is fundamental to foresee the effects of different design configurations.

In this perspective, my research activity has been firstly focused on providing such a generic circuit model of LED light bulbs which is useful at the design stage of the lighting system to guarantee a high power factor. The proposed circuit model is able to emulate the current drawn of any LED light bulbs, whatever lamp driver is used. The procedure and equations to obtain the components of the equivalent circuit of the LED light bulbs have been provided. It is worth highlighting that such a procedure enables to obtain the model of any LED light bulbs available on the market without the knowledge of its inner components (driver, LED package, and so on). The components in the model of a generic lamp can be obtained by a few simple current measurements to be performed at the lamp terminals. These parameters are the coefficients of the functions obtained by linear or polynomial interpolation of the *rms* of the current harmonics. The model validation has confirmed the effectiveness and robustness of the proposed solution.

In the context of LED light bulbs, my research activity has also focused on the study of the LED drivers topologies. Among the various LED drivers available on the market for integrated lighting applications, the single-stage discontinuous current conduction mode flyback PFC converter has been revealed as the preferred choice in order to achieve a high PF with a good cost-performance ratio. Notwithstanding the use of a suitable control method that enables a high PF, different inherent causes of distortion in the input current, due to the power processing mechanism of the High PF QR flyback converter, have been found. Hence, the different issues due to the correlation of these causes, which are the ringing current, the crossover distortion due to transformer leakage inductance, and crossover distortion due to the input storage capacitor have been widely studied. It is worth noting that the problem of the input current distortion caused by the ringing current has been studied for the first time in a QR flyback. Moreover, the crossover distortion due to the input storage capacitor in the case of a Hi-PF flyback LED driver has never been treated in literature. Finally, some practical design guidelines for mitigating the inherent causes of distortion of the input current in a Hi-PF QR flyback driver for LED applications have been provided.

An emerging application of the light emission in a P-N junction is the use of light emission from the body diode of a SiC device for condition monitoring (CM). In this perspective, the LED is an "unintentional" source of light in Power Electronics.

More specifically, the reliability of power electronics devices is raising a great interest, especially, in some critical applications. In this perspective, several condition monitoring methods have been widely proposed so far. Among them, the light emission phenomenon from the body diode of a SiC device has been gained more interest in the last decade. In detail, the electroluminescence phenomenon refers to the photoemission from a SiC material and it is related to its intrinsic body diode. The light brightness in the SiC intrinsic body diode depends on the junction temperature, where the light intensity decrease at high temperature. Thus, the electroluminescence phenomenon can be a suitable indicator of the temperature of the SiC MOSFET chip during a general power converter application.

In this perspective, my research activity has been firstly focused on the description of several CM methods adopted for power devices. Moreover, the intrusiveness of these condition monitoring methods has been also deeply investigated for the first time. More specifically, three intrusiveness criteria have been considered at the device level, converter operation level, and conversion system level. Then, three figures of merit (FOM) have been accordingly considered and assigned to all the condition monitoring methods by means of a mark based rule. Finally, a unique single-value figure of merit has been obtained from the product of the three figures of merit. An application-based FOM that accounts for the importance of the different intrusiveness criteria in a given application has been also proposed. The use of a single value to compare the different CM methods is the main strength of these overall FOMs but they do not provide a fixed value due to the subjective of the marks. On the other hand, the rank of the CM methods according to the rank instead of the mark is proposed by borrowing some reasoning from the Pareto optimality. Such an approach has revealed that the CM

methods adopting, respectively, Integrated photodiode, IR Camera, Acoustic, and two thermo-sensitive electrical parameters (turn on-off delay time and peak gate current) are the best ones in terms of intrusiveness, and they are equivalent each other from the Pareto optimality point of view. Finally, the Integrated photodiode and the IR Camera CM methods are the less intrusive according to the overall FOM.

Therefore, these results have highlighted that the use of the Integrated photodiode is one of the best choices, and considering the SiC MOSFET intrinsic diode as an unintentional diode, a new CM method adopting the Integrated photodiode in a SiC MOSFET module has been devised. More specifically, a non-invasive temperature sensing method for high-voltage SiC MOSFET chips based on the measurement of light emission during reverse conduction has been proposed. The method is based on a compact sensing circuit that can be easily embedded in the package, allowing online temperature estimation. The effectiveness of the circuit has been confirmed by the analysis of a commercial SiC power module. It is worth noting that the proposed temperature sensing strategy is fast, inexpensive, accurate, and of course, non-invasive.

By summing up, this work has been focused on the application of "intentional" and "unintentional" light emission diode in Power Electronics. The first, second, and third chapters have been focused on the use of commercial LED light bulbs. More specifically, Chapter 1 describes the power quality issues and the drivers adopted to mitigate them. Chapter 2 reports the analysis of the intrinsic causes of current distortion in high power factor Quasi-Resonant flyback and proposes some mitigation guidelines. Then, in Chapter 3, a circuit model of a generic LED light bulb has been proposed with the aim to emulate the current drawn from a generic LED light bulb. Chapter 4 focuses on the advantages (in terms of intrusiveness) of using unintentional light emission diode for condition monitoring of power devices. Finally, Chapter 5 describes the proposed innovative condition monitoring method based on the integration of a photodiode to correlate the "unintentional" light emission of a SiC material to the temperature of the device.

х

Contents

| Acknowledgm | ents | . vii |
|-----------------|--|-------|
| Introduction | | .viii |
| List of Figures | | .xiii |
| List of Tables. | | xvii |
| Chapter 1 | Current harmonics due to LED light bulbs: mitigation and modeling | . 20 |
| 1.1 | Introduction | . 20 |
| 1.2 | Brief Overview of LED Driver Topologies | . 21 |
| | 1.2.1 Passive LED Drivers | . 21 |
| | 1.2.2 Switching LED driver | . 23 |
| 1.3 | LED light bulbs harmonics in lighting system design | . 29 |
| Chapter 2 | Analysis of the Input Current Distortion in High PF Quasi-Resonant Flyback LED Drivers . | . 33 |
| 2.1 | Introduction | . 33 |
| 2.2 | Generic Control Method Obtaining High Power Factor | . 34 |
| 2.3 | Experimental Verification of the Input Current Distortion in a Hi-PF QR Flyback LED Driver . | . 37 |
| 2.4 | Distortion Caused by the Ringing Current | . 39 |
| 2.5 | Crossover Distortion Due to the Input Capacitor | . 43 |
| 2.6 | Crossover Distortion Due to Transformer's Leakage Inductance | . 47 |
| 2.7 | Guidelines for Designing High Power Factor Quasi-Resonant Flyback LED Drivers | . 51 |
| Chapter 3 | Circuit model of LED light bulb | . 53 |
| 3.1 | Introduction | . 53 |
| 3.2 | Non-sinusoidal periodical electric quantities | . 54 |
| 3.3 | Linear model for emulating the current drawn from LED light bulb | . 57 |
| | 3.3.1 Description of the linear model | . 57 |
| | 3.3.2 Linear model validation | . 75 |
| 3.4 | Nonlinear model for emulating the current drawn from a LED light bulb | . 79 |
| | 3.4.1 Description of the nonlinear model | . 79 |
| | 3.4.2 Nonlinear model validation | . 83 |
| Chapter 4 | The superiority of the unintentional LED for temperature detection. | . 99 |
| 4.1 | Introduction | . 99 |

| 4.2 | Overv | iew of condition monitoring methods of power devices | 101 |
|------------|----------|--|------------|
| | 4.2.1 | Acoustic methods | 102 |
| | 4.2.2 | Optical Methods | 103 |
| | 4.2.3 | Physical Methods | 105 |
| | 4.2.4 | Electrical Methods | 106 |
| | 4.2.5 | Comparison of the CM methods | 120 |
| 4.3 | Intrus | veness criteria classification | 122 |
| | 4.3.1 | Description of the Intrusiveness criteria | 123 |
| | 4.3.2 | Classification of the CM methods based on the intrusiveness criteria | 124 |
| | 4.3.3 | Figures of Merit (FOMs) | 131 |
| Chapter 5 | Non-ii | nvasive SiC MOSFET Temperature Estimation Method based on the unir | itentional |
| Light | Emission | n from the Intrinsic Diode | 137 |
| 5.1 | Introd | uction | 137 |
| 5.2 | Electro | pluminescence and prior art | 138 |
| 5.3 | Experi | mental setup | 140 |
| 5.4 | Experi | mental results | 142 |
| References | | | 147 |

List of Figures

| Figure 1:1 Typical application of a passive "lossy" LED driver | 22 |
|--|----|
| Figure 1:2 Typical application of a passive "lossless" LED driver. | 23 |
| Figure 1:3 (a) SS switching LED driver topology. (b) TS switching LED driver topology | 24 |
| Figure 1:4 Switching -mode single-stage driver: depending on the position of the capacitor Cs the single dc-dc converter can provide both PFC and output current regulation | 24 |
| Figure 1:5 Simulation of the current drawn I_{ac} by the LED light bulb (a) and the output LED load current ILED (b). The blue one represents the case when the storage capacitor is directly connected on the low-frequency side and the red one on the high-frequency side. | 25 |
| Figure 1:6 Drain to source voltage waveforms for a DCM flyback (a) and a QR flyback (b) | 26 |
| Figure 1:7 QR resonant flyback valley-switching | 27 |
| Figure 1:8 Performance of three main SS LED driver topology. High step down ability (HSDA), Efficiency (EFF), reduced capacitor (CAP), Cost (COST), and Low power applications (LP). | 29 |
| Figure 1:9 Impact of a large LED lighting system on a distribution network complaint with the THD limit of IEEE-Std-519. | 31 |
| Figure 1:10 Simple equivalent circuit usually adopted in literature for modeling a CFL [76-83] | 31 |
| Figure 2:1 Primary and secondary currents in a QR-controlled Hi-PF flyback converter | 35 |
| Figure 2:2 Current waveforms of the circuit in Figure 1:7: line cycle time scale of the primary current I_p , the secondary current I_s , input rectified current I_{in} and current in the ac side I_{ac} | 37 |
| Figure 2:3 Steady-state waveforms at full-load: (a) Vin = 110 V at 60 Hz. (b) Vin = 230 V at 50 Hz. They have been overlapped by the harmonic contributions of the input current (green bars) with the Japanese norm JEITA_MITI Class-C (a), and European norm EN61000-3-2 Class-C (b) (red bars) for lighting equipment | 39 |
| Figure 2:4 (a) Simplified equivalent circuit during Tneg , (b) key waveforms in QR-mode, and DCM operations. | 41 |
| Figure 2:5 Detail of $V_{ac}(\theta)$ (red trace) and $V_{in}(\theta)$ (blue trace) near voltage zero- crossing (a), and zoomed view of the yellow area, the intersection of $V_{ac}(\theta)$ and $V_{in}(\theta)$ (b). Values are normalized to V_{PK} | 44 |

| Figure 2:6 Variation of $R_{eq}(\theta)$ along the sinusoid for the prototype converter caused by the ringing current. | 46 |
|---|----|
| Figure 2:7 (a) Simplified equivalent circuit during off time. (b) Current in the real primary windings (blue trace) and the secondary windings (black trace) | 48 |
| Figure 2:8 Ratio of the critical value of $I_{pkp}(\theta)$ for no input-to-output energy transfer in the dead zone, caused by leakage inductance, as in Equation (35), to that caused by ringing current, as in equation 38, as a function of Kv for $\sigma = 0.95$ | 51 |
| Figure 3:1 Measured amplitude of the fundamental current and greatest harmonics for a LED. The linear interpolation functions are reported on the right with the related current symbol. | 58 |
| Figure 3:2 Experimental setup. A variac, connected to the mains, feeds the LED light bulb through a power analyzer that acquires the measurements. The oscilloscope displays the waveforms. | 59 |
| Figure 3:3 Normalized current harmonics and THD error, when the LED light bulb is feed with 212V | 60 |
| Figure 3:4 Normalized current harmonics and THD error, when the LED light bulb is feed with 237V | 61 |
| Figure 3:5 Proposed linear equivalent circuit of a generic LED light bulb | 61 |
| Figure 3:6 Equivalent circuit when only vFk works while the other independent generators are turned off. It is apparent that this generator supplies only Lk. | 64 |
| Figure 3:7 Equivalent circuit when only v1 works while the other independent generators are turned off. It is apparent that a current at the fundamental frequency flows through Leq. Cc is thus accorded to nullify the fundamental current downstream from the "fundamental current circuit". | 67 |
| Figure 3:8 Resistance Rloop not belonging to the proposed model but necessary to enable PSpice simulation. | 67 |
| Figure 3:9 KVL for obtaining the voltage across a capacitor adopted in the k-th harmonic circuit. | 68 |
| Figure 3:10 VCCS that nullifies the effect of R_{loop} , thus obtaining a Pspice circuit that actually implements the proposed model. | 69 |
| Figure 3:11 A VCCS and a CCCS used to emulate, respectively, a short circuit and an open circuit when L_k is selected (that is $d=1$, then L_k is connected in the lamp model while C_k is disconnected). | 71 |
| Figure 3:12 Netlist of the overall lamp circuit | 73 |
| Figure 3:13 Netlist of the active current circuit. | 73 |
| Figure 3:14 Netlist of the reactive current circuit. | 74 |
| Figure 3:15 Netlist of the compensation circuit. | 74 |
| Figure 3:16 Netlist of the generic harmonic circuit | 75 |
| Figure 3:17 Measured voltage on the mains (blue waveform, <i>rms</i> 212V), measured current (green waveform), and simulated current (red waveform) for a LED. Per-unit system: base voltage 400V, base current 300mA. | 76 |

| Figure 3:18 Measured voltage on the mains (blue waveform, <i>rms</i> 237V), measured current (green waveform) and simulated current (red waveform) for a LED. Per-unit system: base voltage 400V, base current 300mA. | 76 |
|---|-----|
| Figure 3:19 LED light bulb: measured current (green waveform) and the simulated current involving a THD error of 1% (red waveform) and 10% (blue waveform) with a rms voltage of 212V. | 78 |
| Figure 3:20 LED light bulb: measured current (green waveform) and the simulated current involving a THD error of 1% (red waveform) and 10% (blue waveform) with a <i>rms</i> voltage of 237V. | 78 |
| Figure 3:21 Equivalent circuit for the active current. Only one element in the dashed box has to be used (it depends on the sign of b_{P1}) | 81 |
| Figure 3:22 Equivalent circuit for the fundamental nonactive current. Only one element in the dashed box has to be used (it depends on the sign of b_{Q1}) | 82 |
| Figure 3:23 Equivalent circuit of the k-th nonactive current harmonic. | 83 |
| Figure 3:24 Main steps to obtain the circuit model for any LED light bulb using only measurements at its terminal (i.e. black-box model). | 85 |
| Figure 3:25 Maximum and minimum percentage error: (a) before normalization - (b) after normalization. | 87 |
| Figure 3:26 Measured voltage (green) on the mains at V_1 =230V and currents (light blue) drawn by the LED 1 (a) and LED 2 (b). The waveforms are in per unit: 1p.u.= 400V for the voltage, 1p.u.=180 mA for LED1 current, 1p.u.=120 mA for LED2 current. | 88 |
| Figure 3:27 Fundamental and harmonics currents obtained by measurements (left) and simulation (right) at six voltage values for LED1 (a) and LED2(b). The sub-figure at the left corner presents a magnification to better observe higher-order harmonics amplitude | 92 |
| Figure 3:28 Total current drawn by the lamps. I_{mea} is the measured current (red trace) while I_{mod} is the simulated current (blue trace): (a) LED1 at V_1 =212V - (b) LED 1 at V_1 =237V - (c) LED2 at V_1 =212V - (d) LED2 at V_1 =237V | 94 |
| Figure 3:29. Parametric PSpice netlist of the circuit model of LED light bulbs | 95 |
| Figure 3:30. Coefficients (Table 3:1) and phase offset reported in the parametric PSpice netlist of LED1(a) and LED2 (b). | 96 |
| Figure 3:31. Circuit to simulate the current of the overall lighting system | 96 |
| Figure 4:1 Thermal map of a SiC power module in case of current injection in the body diode. | 104 |
| Figure 4:2 Electrical circuits for the static measurement of the voltage under a low current: (a) IGBT (b) MOSFET. | 110 |
| Figure 4:3. Electrical circuits of a DUT for the static measurement of the voltage under high current. | 111 |
| Figure 4:4 Electrical circuits for the measurement of the saturation current in an IGBT device. | 112 |
| Figure 4:5 Circuit for the calibration step of the threshold voltage method. | 113 |
| Figure 4:6 Simulated V_{ge} waveform during the turn off of an IGBT by varying the temperature working operations | 115 |

| Figure 4:7 Simplified $i_c - V_{ge}$ turn ON waveforms of an IGB at different temperature working operations. | 116 |
|--|-----|
| Figure 4:8 IGBT module equivalent circuit. | 118 |
| Figure 4:9 Gate driver RLC network. (a) Peak Detector Schematic to detect peak voltage over the external gate resistor. | 120 |
| Figure 4:10 FOM _D assigned to the CM methods | 126 |
| Figure 4:11 FOM _C assigned to the CM methods. | 128 |
| Figure 4:12 FOMs assigned to the CM methods | 130 |
| Figure 4:13 FOM _o assigned to the CM methods. | 131 |
| Figure 4:14 FOM_D Vs FOM_s represents the center of the circle, while the circle radius is set proportionally to FOM_c | 133 |
| Figure 5:1 Luminescence intensity under different forward currents at 120 °C. (spectrum adopted from [221]). | 138 |
| Figure 5:2 Experimental setup schematic. | 140 |
| Figure 5:3 Picture of the experimental setup. | 140 |
| Figure 5:4 Photodiode voltage output at different T_j and different diode forward current values: (a) I_L = 4,5A (b) I_L =7,5 A (c) I_L = 11,5 A | 141 |
| Figure 5:5 Photodiode voltage output at different T_j and different diode forward current values: (a) I_L =4,5A (b) I_L =7,5 A (c) I_L = 11,5 A | 141 |
| Figure 5:6 Experimental photodiode voltage output at T_j =40 °C with a I_L = 11,5 A at 3 kHz. Both the subinterval Δt_1 and Δt_2 represent the last 30% of the T_{on} and T_{off} time | 143 |
| Figure 5:7 Dependence of V_{OM} with T_j and I_L at 333 Hz. | 144 |
| Figure 5:8 Dependence of V_{OM} with T_j and I_L at 3 kHz | 144 |

List of Tables

| Table 2:1 Main characteristics of the Hi-PF QR flyback converter. | 38 |
|---|-----|
| Table 3:1 Coefficients of the polynomial functions for LED light bulbs with low (LED1) and high PF (LED2) | 90 |
| Table 4:1 Parameters dependence of different CM methods | 101 |
| Table 4:2 Summarization of different CM methods | 120 |
| Table 5:1 Main features of the Silicon PIN photodiode at 25°C | 141 |
| Table 5:2 Main features of the power module at 25°C | 141 |
| Table 5:3 Coefficients of the polynomial functions Equation (113) | 145 |
| Table 5:4 Coefficients of the polynomial functions Equation (114) | 145 |

Chapter 1 Current harmonics due to LED light bulbs: mitigation and modeling

1.1 Introduction

The lighting systems, both in-home and business environments, may take up over 15% of overall electrical consumption [1-3]. Hence, the lighting system designers always search for applying new technologies to obtain savings, thus reducing the waste of electrical energy and the bill. Nowadays, a huge quantity of lighting systems is widespread in the marketplace, where different types of lamps are used. Light bulbs may be divided in incandescent, compact fluorescent lamp (CFL), high-intensity discharge, halogen, LED, and so on [4-5]. Among them, the LED light bulbs technology is today in a good stage of trade-off for cost, lifetime, and efficiency in the case of lighting applications. Moreover, LED light bulbs are expected to provide in future more lumens per watt in comparison to today technology [6]. The LED technology presents the additional advantage of being free from the toxic substances contained in some CFL, which may cause pollution issues [7].

Generally speaking, a LED light bulb can be thought of as a combination of LED semiconductor materials and a driver circuit. The choice of the LED driver circuit plays a fundamental role from a power quality point of view. In fact, the driver controls its own input current and the LED output current. Consequently, it must guarantee a high power factor (PF), which means a displacement power factor next to one and an input current with low total harmonic distortion (THD) [8-9]. Actually, the LED light bulbs must comply with the national and international standards and regulations concerning harmonic currents, such as the standard IEEE-Std-519 and the IEC 61000-3-2.

The needed to ensure a high power factor is further significant in the case of the design of a lighting system, especially when a very huge number of lamps are used. In this perspective, the electrical model of each light bulb is fundamental to foresee the effects of different design configurations. An incandescent lamp typically consists of a glass enclosure containing a tungsten filament that produces light at a given temperature reached due to the flowing of the current and, consequently, it can be simply modeled by a resistor. Differently, the equivalent electrical model of a LED light bulb is not easy to define, due to the converters inside the lamp as well as due to the control technique used to obtain constant current on the diode light engine [10].

The following chapter has been organized into two main parts: firstly, various topologies of passive and switching LED drivers have been described and classified. More specifically, the passive drivers are simpler and more reliable and they operate at the line frequency. They do not provide any active control, so a significant current ripple can occur. The switching drivers can strongly reduce the current distortion and improve the PF with the additional advantage of being suitable for high-frequency operations, thus enabling reduced size. On the other hand, switching LED drivers are less reliable than passive ones. The second part of the chapter has been focused on the current state of the art of the harmonic pollution produced by a LED light bulb. More specifically, it has been widely discussed several key aspects of the optimal lighting system design, and the power quality degradation due to the employment of the LED light bulbs, and more in general, for different kinds of energy-saving lamps (ESLs). From the perspective of an optimal and efficient design, the distorted current drawn by a large number of LED light bulbs and the prediction of the amount of current harmonic that the lamps inject into the power grid have been discussed.

1.2 Brief Overview of LED Driver Topologies

In this section, various topologies of passive and switching LED drivers have been described and classified. The main drawback of passive LED drivers is the low PF that often cannot comply with the standard limit. Instead, switching LED drivers can realize a better output current regulation and higher power density, and they are more capable of satisfying the standard limits. Generally, there is a wide variety of LED driver topologies depending on the different power ranges. Furthermore, the topology of the driver can also be selected according to other requirements, including cost, galvanic isolation, and efficiency.

1.2.1 Passive LED Drivers

Passive LED drivers are characterized by the exclusive use of passive components (e.g., resistors, capacitors, magnetic components). The insertion of an impedance between the ac line and the LED light bulb load to limit the current is mandatory. One of the main drawbacks of these topologies is the low PF and high THD, sometimes not enough to meet the standards [11].

Passive LED drivers can be classified into lossy and lossless impedance driver. The strength and simplicity of the lossy impedance is due to the use of a resistor RL or a linear regulator on the dc side, as depicted in Figure 1:1. In many practical applications, it is widespread the use of a bulky step-down transformer from high to low voltage. It has a two-fold benefit: it reduces the voltage drop on the resistor with an increase of the overall system efficiency and, at the same time, guarantees galvanic isolation. A large electrolytic capacitor C_S is typically used in order to avoid flickering.

A passive driver with a lossless impedance (an inductor, capacitor, or their combinations), usually placed in the ac side to limit the LED output current, is shown in the example of Figure 1:2. An inductor L_{in} is usually adopted with the aim to replace the less efficient and low-frequency transformer. L_{in} usually behaves as an additional input filter (by means of C_{in}) that smooths the input current and leads to advantages such as reducing the input current distortion [12-14]. Instead of using a bulky electrolytic capacitor (E-CAP) that ensures a constant output current, it is often used as a non-E-CAP on the dc side, still achieving a higher PF with respect to the lossy counterpart. This solution can ensure a longer lifetime and small size of the overall system, although there is a small output current ripple in the LED load. Notwithstanding, lossless passive drivers can reach high efficiency, above 90% [13].

They can be easily employed in outdoor applications and they are very cost-effective especially for low-power applications. The main drawbacks of passive LED drivers are their lack of a proper output current control and, an input current THD that does not always meet the standards target.



Figure 1:1 Typical application of a passive "lossy" LED driver.



Figure 1:2 Typical application of a passive "lossless" LED driver.

1.2.2 Switching LED driver

Switching LED drivers bring various advantages arising from the use of the switching devices. Moreover, switching LED drivers can embed into a single electronic ballast various functionalities such as circuit fault protection, active and high PFC [15]. They are especially employed in indoor applications since the whole circuit control is compact, reliable, effective with a low and controllable ripple output current. Another strength point of switching LED drivers is their higher efficiency in comparison to the passive ones. Switching LED driver topologies can be classified into two main categories, SS and TS (see Figure 1:3). The SS drivers consist of a single power stage that acts as both a dc-dc regulator and PFC (Figure 1:3(a)) with a storage capacitor C_s . On the other hand, the TS drivers are used for high power applications. The aforementioned drivers comprise two power conversion stages that can perform different functions. In general, the first stage acts as a PFC and the second one as a dc-dc regulator and filter [16], as depicted in Figure 1:3(b).



(a)



Figure 1:3 (a) SS switching LED driver topology. (b) TS switching LED driver topology.



Figure 1:4 Switching -mode single-stage driver: depending on the position of the capacitor Cs the single dc-dc converter can provide both PFC and output current regulation.

In the following, a brief focus on the SS LED drivers with the main advantages of different topologies adopted in the literature.

SS drivers usually have low component count and among them a power switch that controls a power conversion stage. However, it is often difficult for a SS driver to simultaneously ensure good performance in many respects, such as high efficiency, high PF, constant current output, and so on. SS drivers are suitable for low and medium power class applications (below 70 W) where size and cost are usually more critical than PF and efficiency. The storage capacitor is usually placed downstream from the dc-dc converter, which is on the high-frequency side (red capacitor Figure 1:4) to obtain a high PF. Notwithstanding, another approach can be found in some embedded LED drivers where the capacitor is placed on the low-frequency side (blue capacitor Figure 1:4). More specifically, such an approach has been sometimes used in very low power applications (below 5 W) [17]. Indeed, nowadays, in this power range, passive LED drivers are adopted to guarantee an inexpensive cost. Figure 1:5 shows the waveforms of the current drawn by LED light bulbs when the capacitor is placed on the low and high-frequency side. The waveforms confirm that only when the storage capacitor is placed downstream from the dc-dc converter it guarantees PFC. Although the input current drawn by the LED light bulbs has a better THD when the capacitor is placed at the high-frequency side, a higher output current ripple occurs in comparison with the previous solution (as shown in Figure 1:5(b)). However, the required storage capacitance is not always able to handle both the low and the high-frequency ripple [18].



Figure 1:5 Simulation of the current drawn *I_{oc}* by the LED light bulb (a) and the output LED load current ILED (b). The blue one represents the case when the storage capacitor is directly connected on the low-frequency side and the red one on the high-frequency side.

The SS LED drivers can be roughly classified on the bandwidth of the feedback control systems. In case of a narrow bandwidth (e.g. boost, buck-boost, flyback) the storage capacitor must be located at the output of the converter. Instead, in case of a wide bandwidth systems (e.g. quadratic or S4ICS topologies), the storage capacitor is placed between the two semi-stages. In the following, it has been analyzed the narrow bandwidth systems. In detail, a huge number of conventional LED drivers are widely discussed in literature, such as buck [19-21], buck-boost [22], SEPIC [23-24], flyback [25-26], half-bridge [27-29], push-pull converters [30-31]. Furthermore, PFC can be achieved with valley-fill circuits [32]. Other approaches, as coupled-inductor modified converters [33-36] and valley-fill modified converters [37] show a simple solution to the step-down ratio requirement without compromising the efficiency and system complexity.

The flyback converter has been widely adopted in LED driver lamps because of its simple structure and high PF. One of the strengths is its efficiency that can be improved by using the leakage energy or using soft-switching techniques [38]. In spite of the flyback LED driver has various advantages, in many practical applications, the QR mode operation has become one of the most familiar methods in LED driver applications. It is noteworthy the decrease of switching losses with respect to a flyback converter operated with a fixed frequency. Moreover, the QR driver has an enhanced transient response in DCM operation [39-40] and it may have a smaller EMI filter [41]. In fact, in applications operated from the mains, the switching frequency is modulated at twice the mains frequency due to the voltage ripple appearing across the input capacitance. The switching frequency span depends on the amplitude of this input voltage ripple. This causes the spectrum to be spread over some frequency bands, rather than being concentrated on single frequency values.

In addition, the QR driver has a higher safety degree under short circuit conditions: since the switch is not enabled until the primary windings are fully demagnetized, both the flux runaway and the transformer saturation are not possible. On the other hand, the QR flyback LED driver may have a high ripple output current and high conduction losses in comparison to the fixed frequency one.

It is important to point out that the difference between a DCM flyback and a QR flyback is in the turn-on mechanism. In a DCM flyback converter, the gate driver provides a constant switching frequency, while in a QR flyback it is used a variable frequency where the off-time depends on the resonant valley detection of the drain to source voltage ringing that follows transformer's demagnetization. Figure 1:6 depicts the drain to source voltage waveforms in case of a DCM flyback (Figure 1:6(a)) and a QR flyback (Figure 1:6(b)), thus highlighting the benefit in terms of reduced switching losses.



Figure 1:6 Drain to source voltage waveforms for a DCM flyback (a) and a QR flyback (b).



Figure 1:7 QR resonant flyback valley-switching.

A simplified schematic of a QR flyback LED driver (not for Hi-PF applications) is shown in Figure 1:7. The primary current I_p starts to flow into L_p when the power switch is turned on. The voltage at the secondary is such that the diode is reverse biased, hence the capacitance C_{out} supplies the LED string. Once the switch is turned off, the current I_p goes to zero and the voltage across both windings reverses, so that the output diode is forward-biased, the current starts flowing in the secondary winding and C_{out} can be charged by the energy stored in the transformer. It is worth noting that while the current flows on the secondary side, the drain-source voltage is equal to the rectified input $V_{in}(t)$ plus the reflected output voltage V_R at the primary windings. The transformer takes a time T_{FW} to demagnetize, and as soon as the energy transfer is completed, the drain-source voltage starts ringing. The main reason for this energy exchange phenomenon is due to a resonant tank between the inductance L_{pp} and the capacitance C_{pp} . The inductance L_{pp} is the sum between the leakage inductance of the copper paths and the primary winding L_p . The latter is the prevalent contribution, hence L_{pp} can be simply approximated as L_p . The capacitance C_{pp} can be expressed as the sum of the parasitic capacitances on the primary circuit and the capacitances of the secondary circuit referred to as the primary one. The latter is due to the parasitic capacitances of the secondary circuit and the output capacitance Cout. The primary circuit capacitance includes various contributions as the output parasitic capacitance (C_{oss}) of the MOSFET, the junction capacitance of the diode, and the package capacitance of the intrawinding capacitance of the transformer plus other stray contributors together and so on [42-45]. Coss is a strongly nonlinear capacitance and, especially in the latest MOSFET generations, it increases dramatically (a hundred times or more) when the drain-source voltage, v_{DS} , falls below few ten volts, i.e. C_{pp} is a function of v_{DS} : C_{pp} (v_{DS}). In the following, this capacitance will be considered constant or, at least, not significantly impacting the overall C_{DS} . Hence, the resonant frequency f_r can be represented as:

$$f_r = \frac{1}{2\pi\sqrt{L_{pp}C_{pp}(v_{DS})}} \approx \frac{1}{2\pi\sqrt{L_pC_{DS}}}$$
(1)

In applications, the inductance L_p is measured with an impedance meter or by measuring the di/dt when a voltage square wave is applied. While C_{DS} is estimated by (1) from the measurement of the drain-source voltage ringing.

Bearing in mind that in a conventional QR topology is used a feedback loop voltage that controls the average value of the secondary current I_s , the switching frequency is continuously adjusted depending on the output current load. In this way, the switching device turns on when necessary provided that the inner controller is able to detect a valley in the ringing drain-source voltage. The switching period *T* can be expressed as:

$$\begin{cases} T = T_{on} + T_{FW} + T_{ring} \\ T_{ring} = \frac{1}{2} [1 + 2(k - 1)]T_r \end{cases}$$
(2)

where T_{on} is the ON time of the power switch, T_{FW} is the time interval where the current flows on the secondary side and T_{ring} is the time interval during which the drain-source voltage rings. T_{on} is reduced as the load reduces, hence at very light load the frequency is high. Therefore, the maximum switching frequency imposes a minimum T_{on} . In this case, a further load reduction involves an increment of T_{ring} . In detail, Tring strictly depends on the load level and it is imposed by keeping off the switch for some valley points in the drain-source voltage. In other terms, it depends on the number of the valley points k "skipped" by the controller. More specifically, k increases as the load level decreases. Until T_{on} is greater its minimum value k=1, that is the switch turned on at the first valley on the drain to source voltage waveform (one-half of the resonant period). In the following, the QR mode operation of the converter is defined when k=1.

The reason behind the turn-on during a valley point of the drain-source voltage is the achievement of a lower capacitive switching losses:

$$E_{s} = \frac{1}{2} C_{DS} V_{DS}^{2}$$
(3)

This mechanism is able to minimize in a significant way the switching losses. It is worth noting that the bill of material of a QR driver contains only a few simple components which implies an inexpensive solution [45-46].

Finally, Figure 1:8 compares the performance between a QR flyback, a buck-boost, and a hybrid solution [31-43] in terms of some key factors. The high step down ability (HSDA)[31-33], efficiency (EFF) [33-36], compact capacitor volume (CAP) [37], cost (COST)[38 -41], and low power range applications (LP) [42] have been taken into account

for the three converters. The comparison highlights the superior performance of the QR flyback converter, which makes it the preferred choice for LED driver applications.



Figure 1:8 Performance of three main SS LED driver topology. High step down ability (HSDA), Efficiency (EFF), reduced capacitor (CAP), Cost (COST), and Low power applications (LP).

1.3 LED light bulbs harmonics in lighting system design

Different analyses of LED light bulbs related to their behavior and efficiency as well as to the driver design have been widely treated [47-49]. The lighting design considers several variables for both indoor and outdoor applications [50]. More specifically the design accounts for the light level and the uniformity of the light pattern, the aesthetic appearance, the economic benefit, the safety, and appropriate equipment [51-52]. Nowadays, with the large use of the energy-saving lamps, that are non-linear loads, the amount of distorted current drawn by the lighting system has to be considered looking to the power quality issues [53]. Such issues have been not considered for many years as each lamp injection is very small but the problem will increase considerably in the next future with the massive use of LED light bulbs. In the past, only the distortion due to the single large harmonic source has been of concern, and the solution was the design of the dedicated filter, or any other harmonic mitigation option, able at meeting the system requirements, and installed across the device terminal [54-55]. The widespread adoption of LED light bulbs can be more deleterious than such a source since a large lighting system with LED light bulbs acts as a large distortion source but it is composed of many distributed harmonic sources with a difficult compensation [56]. The general adoption of the power factor corrector inside the lamps mitigates the problem but it leads to increasing the lamp cost [57]. In literature, the impact on the distribution network of the current harmonic distortion due to the LED light bulbs has already been explored [58-60]. Some works have analyzed the harmonic emissions before and after replacing the existing lamps with the LED ones [61-62]. In [63] the harmonic current from the domestic appliances with the use at the same time of LED light bulbs has been investigated. A comparison of different commercial LED light bulbs, in terms of harmonics and their negative impact on the distribution network, has been presented in [64]. Advanced studies regarding the injection of the harmonics have been performed in [65] by means of a simulator that emulates different LED light bulbs, where each lamp has been represented as a simple independent current generator.

Moreover, the harmonic content of a LED light bulbs changes with the variation of the voltage supply. Such a variation is very frequently due to many factors, such as the load profile variation in the nodes of the network, the presence of distributed generators based on intermittent or variable renewable energy sources, the wider use of electric vehicles, the network reconfigurations, the voltage regulation and so on [66-71].

Therefore, a proper design of the lighting system needs to foresee the distortion level due to the whole lighting system for different network voltage. For this purpose, it is necessary to estimate the harmonics due to different LED light bulbs configurations. Conventionally, the fixed current injection method sums the harmonic current magnitudes of each load, but it may significantly overestimate the cumulative harmonic currents produced by the LED light bulbs [72]. It is more appropriate to consider the diversity and attenuation factors because they can predict the behavior of distributed single-phase loads more accurately [73-76].

In this perspective, the PSpice implementation of the LED light bulb model can be adopted to predict the distortion levels at the design stage of the lighting system. Considering that the current harmonic usually significantly increases when LED light bulbs are used, the effect of different configurations can be easily forecasted in view of an optimal design of the lighting system. Figure 1:9 shows the voltage THD level before and after a large LED lighting system turn-on. Before the turn-on, it is expected the voltage THD is under the 8% limit imposed by IEEE-Std-519. More specifically, it is expected that any countermeasures have been realized at the load/generation points along the distribution network to avoid exceeding the limit. The impact of a few lamps on the voltage THD is negligible even when their current is highly distorted, this is due to the low current draw. When a large number of LED light bulbs are adopted, the whole current is not negligible and, consequently, the current drawn from the LED light system may negatively affect the voltage THD. In this context, the optimal design task is finding the combinations of LED light bulbs which led to the minimum overall distorted current even when the single LED light bulb drawn a very distorted current. Such results can be obtained by optimally exploiting the operating principle of the diversity and attenuation factors [72].



Figure 1:9 Impact of a large LED lighting system on a distribution network complaint with the THD limit of IEEE-Std-519.

Some models devised for CFL could be adopted for LED. More specifically, some researchers have been worked out the behavior of the CFL using an equivalent electrical circuit [77-83]. A common and general model is depicted in Figure 1:10, where the key circuit components are: a diode bridge that rectifies the main voltage, an ac equivalent resistor (R_{ac}) , and a DC smoothing electrolytic capacitor (C) supplying the downstream inverter that, in turn, feeds the fluorescent tube. Both the inverter and the fluorescent tube can be modeled as a unique equivalent resistance, R_D , since they behave like a constant load for the DC busbar. In [76-77] an admittance model that depends on some internal parameters (firing and extinction angles) that, in turn, depend on the voltage waveform has been used instead of the general model. A method based on the measurements to obtain harmonic models of power electronic-based home appliances, among which ESLs, has been presented in [76]. In [77] a model based on harmonically coupled admittance matrix used to study harmonics in static converters of ESLs has been proposed. However, the admittance model is valid only in a specific condition of the supply voltage. In [78] a simplified version of the general model is adopted where the AC resistor has been neglected since it presents a small resistance. However, this approximation may lead to not realizable infinite slopes of the ac current rising edge.



Figure 1:10 Simple equivalent circuit usually adopted in literature for modeling a CFL [76-83]

Similarly, the CFL equivalent circuit considering the AC resistance has been proposed in [79]. The model considers that the behavior of the CFL electrical circuit is similar to the one shown in Figure 1:10. The supply voltage has been modeled with the series of the fundamental and inter-harmonic voltage generators with their network equivalent impedance. In [80] the CFL parameter estimation has been obtained as a detailed analysis of the electrical model developed in [79], where the former used non-linear least-square procedures based on actual measurements. The resolution is based on the Newton method calculating the terms of the Jacobian matrix by finite difference approach. The study of CFL impact and the related model has involved the determination of the CFL equivalent circuit parameters R_{ac} , C, and R_D described previously. It is worth noting that in the literature several procedures are developed to determine the parameters from the supply voltage and ac current measurements [81-82]. Other studies deal with the estimation of other non-linear loads using least-square algorithms [82]. In [83] the parameter estimation of single-phase rectifiers by analyzing several non-linear sets of equations has been performed. More specifically, the former has proposed the two methods to esteem the electrical components in the input rectifier that there are in many electronic equipment available in the market. The latter has presented an estimation algorithm based on a rectifier model and actual measurements.

The procedures described before offers several rules that enable to obtain the circuit model of CFLs available on the market. Such an approach could be adopted for LED light bulbs, but these procedures require the knowledge of the ballast internal circuit components and not negligible computational resources, which make them unsuitable tools for optimal lighting system design. Moreover, the circuit model includes non-linear components. On the other hand, as said before, although several objectives and constraints have been considered so far in the optimal lighting system design, the harmonic mitigation target has been totally neglected. Therefore, a parametric Pspice circuit overcoming these limitations has been proposed in chapter 2. The knowledge of the ballast circuit, driver, package, and so on, is not necessary to obtain the parameters to be used in the proposed Pspice circuit. Indeed, any LED light bulb can be treated as a black box, and the components of the circuit are simply obtained with some electrical measurements, which is the amplitude and phase shift of the fundamental current and harmonics drawn by the LED light bulb. An additional advantage of the proposed circuit is that it contains only linear components and the circuit simulation is very fast and it requires few computational efforts.

Chapter 2 Analysis of the Input Current Distortion in High PF Quasi-Resonant Flyback LED Drivers

2.1 Introduction

As said before, the driver regulates its input current as well as the output current towards the LED load, thus it must guarantee a high power factor and a suitable current for the LED load. Among the various LED drivers available on the market, the quasi-resonant (QR) flyback topology shows interesting benefits. The flyback converter is the most used SS LED driver because it obtains high step-down ability and it provides a good tradeoff among the power quality, the cost, the capacitor size, and the efficiency. In integrated lighting applications, a SS discontinuous current conduction mode (DCM) flyback PFC converter is commonly used to drive the LED light bulbs in order to achieve a high PF. The task is performed by means of a simple circuit configuration used to regulate the lamp current. A flyback LED driver operating in DCM where the PFC proprieties can be easily obtained has been frequently adopted. In this context, the quasi-resonant (QR) flyback LED driver can effectively reduce the transformer size and weight thanks to the high switching frequency, which is not fixed since it increases as the load decreases. Furthermore, the switching losses can be strongly reduced by adopting zero voltage switching (ZVS).

A suitable control method enables to obtain a high power factor (Hi-PF) QR flyback LED driver. On the other hand, there are different inherent causes of distortion in the input current to be faced. It is worth noting that, these causes are due to the power processing mechanism of the Hi-PF QR flyback converter and they are not ascribable to the specific control method (although the control method may mitigate them). The first part of the proposed work aims at investigating and analyzing the different issues due to the correlation of several causes such as the ringing current, crossover distortion due to transformer leakage inductance, and crossover distortion due to the input storage capacitor. The input current distortion caused by the ringing current has been widely studied for the boost converter, while it is the first time the problem has been studied for the QR flyback. Moreover, the

crossover distortion due to the input storage capacitor in the case of a Hi-PF flyback LED driver has never been treated in literature. Furthermore, an accurate analysis has been performed in this work by considering the linear approximation of the input voltage and the related comparison with a sinusoidal waveform. Finally, some practical design guidelines for mitigating them in a Hi-PF QR flyback driver for LED applications are provided.

2.2 Generic Control Method Obtaining High Power Factor

Although the QR flyback converter topology is extremely popular since it is a very costeffective solution with high performance, the converter features inherent distortion of the input current. Normally, this distortion is not a concern for compliance with the IEC61000-3-2, however, some input current THD targets (e.g., <10% at full power) are becoming market requirements that are very difficult to achieve, especially when working with lighting equipment over 25 W [84].

As shown in Figure 1:7, the storage capacitor of the QR flyback is placed on the lowfrequency side of the dc-dc converter. With the aim to reduce the input current THD, the capacitor should be placed downstream from the converter, according to the waveform simulations in Figure 1:5 (red trace) related to the capacitor position of Figure 1:4. In this perspective, the converter can be considered as a Hi-PF QR flyback, which has a rectified voltage in input and the storage capacitor downstream of the power stage. The aforementioned solution strongly reduces the harmonics distortions of the input current drawn by the LED light bulb.

The control gear in a Hi-PF QR flyback converter has a twofold task: firstly, it is responsible for regulating the output voltage or current and, simultaneously, it has to maintain a low THD of the input current. For the sake of completeness, the characteristics and proprieties of the control method that have to be considered to achieve a Hi-PF QR flyback converter will be briefly discussed.

The control method, shown in Figure 2:1, is responsible for the turn on and turn off of the switch.

Henceforth, the quantities depending on the instantaneous line voltage will be considered as a function of the term $\theta = 2\pi f_{line}t$.

The target of the control method is to obtain an input current very similar to a sinusoid inphase with the input voltage to achieve high-PF. Regardless of the specific control method, the previous target can be partially converted into obtaining a primary current, as shown by I_p in Figure 1:7, whose peak envelope, detailed by the green elements in Figure 2:1, leads to a sinusoid in-phase with the input voltage. To obtain such an envelope, the control method must turn off the switch when the current on the primary side reaches one of the (green) peaks in Figure 2:1. To obtain a similar result for the secondary current, shown by the red elements in Figure 2:1, a similar control must be adopted for the switch turn on. The turn-on must occur when the transformer is fully demagnetized, thanks to a zero-current detector. In other terms, the combination of these switching rules, which result in a variable switching frequency, shown in Figure 2:1, ensures that the peaks of the primary current, in a half period of the rectified voltage $V_{in}(\theta)$, can be enveloped by a rectified sinusoid. The primary current $I_p(\theta)$ in a switching cycle is triangular shaped and flows only during the switch on-time, as sketched by the green triangles shown in Figure 2:1. During the off-time, the secondary current $I_s(\theta)$ flows and it is represented by the red triangles. Therefore, the switching frequency SW is variable, where the $T_{ON}(\theta)$ and $T_{OFF}(\theta)$ are modulated to achieve a sinusoidal current envelope.



Figure 2:1 Primary and secondary currents in a QR-controlled Hi-PF flyback converter.

However, the primary current $I_{\rho}(\theta)$ flows only during the on-time of the power switch, and this means that the average value of the primary current deviates significantly from an ideal sinusoid. To ensure good performance and low THD input current for a Hi-PF driver LED, the quantity $D^{2}T$ (*D* is the duty ratio) must be constant along each line half-cycle, thus a unity PF is obtained [85].

In the following, it is assumed that the generic control method implements the previous features. For the sake of simplicity to obtain a quantitative expression of the input current I_{in} , the following assumptions have been considered:

1. The line voltage is sinusoidal, and the input bridge rectifier is ideal, thus the voltage at the bridge output terminal is a rectified sinusoid.

- 2. The voltage drop across the power switch in the on-state is negligible and there is negligible energy accumulation on the dc side of the bridge.
- 3. The transformer windings are perfectly coupled (i.e., no leakage inductance).
- 4. The turn-off transient of the power switch has a negligible duration so that T_{FW} immediately follows T_{ON} .
- 5. The converter is operated so the power switch is turned on in each cycle after the secondary current becomes zero, therefore in either QR-mode (i.e., on the first valley of the ringing in the drain-source voltage) or DCM.
- 6. The output voltage is constant along a line half-cycle.
- 7. During the time interval elapsing from the instant when the transformer demagnetizes to the instant when the power switch is turned on, the transformer current is zero; consequently, the initial current during the on-time is zero too. This time interval is equal to $T_r/2$ in the case of the converter being used in QR mode.

It is worth generalizing the relation to D^2T when a variable switching frequency is used, in particular:

$$D^{2}T = \left[\frac{T_{ON}(\theta)}{T(\theta)}\right]^{2} T(\theta) = \frac{T_{ON}(\theta)^{2}}{T(\theta)} = \text{constant}$$
(4)

where the dependencies of T_{ON} and T on θ point out that they are a function of the instantaneous line voltage V_{ac} . Assuming that the rectified voltage V_{in} is sinusoidal in $0 \le \theta \le \pi$, according to assumption 1, it can be written as:

$$V_{in}(\theta) = V_{in,pk}\sin(\theta)$$
(5)

where V_{in,pk} is the amplitude of the rectified voltage.

By considering the current flowing in the primary windings during the time interval T_{ON} , and bearing in mind the inductance current-voltage differential relation, the peak value of primary current $I_{pkp}(\theta)$ can be expressed as:

$$I_{pkp}(\theta) = \frac{1}{L_p} \left(V_{in,pk} \sin(\theta) \right) T_{ON}(\theta)$$
(6)

The generic control method must ensure that the height of the triangles depicted in Figure 2:1 varies along a line cycle as expressed by (6). To reach this target, the control method

must properly vary the width of $T_{ON}(\theta)$ as well as $T_{OFF}(\theta)$. The input current $I_{in}(\theta)$ is the average value of each triangle over a switching cycle, hence, taking (6) into account:

$$I_{in}(\theta) = \frac{1}{2} I_{pkp}(\theta) \frac{T_{ON}(\theta)}{T(\theta)} = \frac{1}{2L_p} \left(V_{in,pk} \sin(\theta) \right) \frac{T_{ON}(\theta)^2}{T(\theta)}$$
(7)

It is worth noting that, if the ratio $T_{on}(\theta)^2/T(\theta)$ is maintained constant by the controller, the input current $I_{in}(\theta)$ can be assumed sinusoidal. Figure 2:2 summarizes the key current waveform in the Hi-PF driver LED.



Figure 2:2 Current waveforms of the circuit in Figure 1:7: line cycle time scale of the primary current I_{p} , the secondary current I_{s} , input rectified current I_{in} and current in the ac side I_{ac} .

Despite the fact that a suitable control method leads to a Hi-PF QR flyback driver according to the previous features, there are different inherent causes of distortion in the input current to be faced. The distortion due to these causes is reported in the next section where the THD has been experimentally evaluated. After that, in the following subsections, these causes, due to the power processing mechanism of the QR flyback converter that are not ascribable to the specific control method (although the control method may mitigate them), are described.

2.3 Experimental Verification of the Input Current Distortion in a Hi-PF QR Flyback LED Driver

A prototype of a Hi-PF QR flyback LED driver has been set up to highlight the distortion occurring, even when a control method implementing the previous features is adopted. The main parameters of the converter are summarized in Table 2:1. The prototype has been built and its performance has been evaluated on the bench. The PF has been result greater than 0.98 over the input voltage range at full load. At 50% load, at a low line, it was nearly equal to that at 100% load, but at the high line, it dropped to about 0.97 at 230 V_{ac} . Figure 2:3 depicts the experimental waveforms of the input current $I_{ac}(\theta)$ (blue trace), the output voltage (red trace), the drain current of the MOSFET (green trace), and the current sense voltage reference for the controller (purple trace). The measurements have been carried out at full load both 110 V_{ac} (60 Hz) and 230 V_{ac} (50 Hz). From the experimental evidence, the shape of the measured $I_{ac}(\theta)$ results very close to an ideal sinusoid waveform. It is worth noting that the harmonic contribution the prototype boards met the European norm EN61000-3-2 Class-C and Japanese norm JEITA_MITI Class-C, both of which are relevant to lighting equipment, at full load and nominal input voltage mains, as depicted in Figure 2:3.

A small distortion in the input current is apparent in Figure 2:3 (blue trace) by looking at the zero crossing of the waveform. Therefore, as previously mentioned, notwithstanding the enhanced performance of a Hi-PF QR flyback LED driver, there are still different inherent causes of distortion in the input current that are not ascribable to the specific control method. The causes are the ringing current, the crossover distortion due to transformer leakage inductance, and crossover distortion due to the input storage capacitor. Generally speaking, being the converter in a nonlinear system, the overall THD of the input current cannot be the sum of each distortion contribution. Due to the number and the complexity of the distortion causes and, above all, due to the complexity of their mutual multi-interactions, the only way to have a sensible estimate of the overall result in terms of input current THD is to resort to simulations.

It is worth noting that the distortion of the input current $I_{ac}(\theta)$ caused by the ringing effect has already been studied theoretically for boost PFC [86-88], while in the case of a Hi-PF QR flyback it has not yet been treated. Furthermore, the crossover distortion due to the input storage capacitor in the case of a Hi-PF flyback LED driver has never been studied. Finally, the crossover distortion due to transformer leakage inductance has also been theoretically analyzed in depth.

| Parameter | Value |
|---|----------|
| Input voltage range [V _{ac}] | 90–265 V |
| Line frequency range [f _l] | 47–63 Hz |
| Rated output voltage [V _{out}] | 48 V |
| Regulated dc output current [<i>I</i> _{out}] | 700 mA |
| Expected full-load efficiency $[\eta]$ | 86% |
| Transformer primary inductance $[L_p]$ | 500 µH |
| Reflected voltage $[V_R]$ | 120 V |
| Drain–Source capacitance [C _{DS}] | 150 pF |

Table 2:1 Main characteristics of the Hi-PF QR flyback converter.



Figure 2:3 Steady-state waveforms at full-load: (a) Vin = 110 V at 60 Hz. (b) Vin = 230 V at 50 Hz. They have been overlapped by the harmonic contributions of the input current (green bars) with the Japanese norm JEITA_MITI Class-C (a), and European norm EN61000-3-2 Class-C (b) (red bars) for lighting equipment.

2.4 Distortion Caused by the Ringing Current

In a flyback, the drain-to-source voltage rings as soon as the secondary winding is fully demagnetized. The energy is exchanged between the total capacitance C_{DS} of the drain node and the primary inductance of the flyback transformer L_p . For the sake of simplicity, assumption 3 has been assumed, and consequentially it has been carried out an equivalent circuit of the QR flyback during the time interval T_{neg} , which is the duration of the negative portion of the primary current, I_p . The duration of the positive portion, T_{pos} , of the primary current is equal to T_{ON} when the current in the turn-on instant of the power switch is zero. The simplified circuit model and the key waveforms are depicted in Figure 2:4.

In Figure 2:4(a), the voltage across L_p is equal to the reflected output voltage V_R , which is almost constant during T_{FW} , according to the previous assumption 6. After that, the voltage

across L_p and C_{DS} and the current through them start to oscillate. More specifically, the analytical expressions of $V_{DS}(t)$ and $I_p(t)$ in the time interval T_{neg} , can be written as:

$$V_{DS}(t) = \begin{cases} V_{in}(t) + V_R \cos\left(2\pi \frac{t}{T_r}\right) & 0 < t \le T_z \\ 0 & T_z < t \le T_{neg} \end{cases}$$
(8)

$$I_{p}(t) = \begin{cases} -Y_{L}V_{R}\sin\left(2\pi\frac{t}{T_{r}}\right) & 0 < t \le T_{z} \\ I_{p}(T_{z}) + \frac{V_{in}}{L_{p}}t & T_{z} < t \le T_{neg} \end{cases}$$
(9)

$$Y_L = \sqrt{\frac{C_{DS}}{L_p}} \tag{10}$$

where Y_L is the characteristic admittance of the C_{DS} - L_p tank circuit and T_z is the time interval needed for the V_{DS} to fall zero when $V_{in} < V_R$; that is:

$$V_R \cos\left(2\pi \frac{T_z}{T_r}\right) = -V_{in}\left(T_z\right) \tag{11}$$

When QR-mode is adopted, the device can be turned on, but according to assumption 5, the current must reach zero. T_{zz} is the time interval needed for the primary current I_p to ramp linearly until zero from the current value $I_p(T_z)$. In DCM operation, considering that I_p oscillates around zero, and that ringing is damped, after a few ringing cycles, assumption 7 can be considered exactly true; with QR operation, the negative current just after demagnetization is not compensated by subsequent positive contributions as in DCM. Considering zero the average value of I_p , as per assumption 7, is already a better approximation as compared to totally neglecting T_R (i.e., assuming the operation is exactly at the boundary between DCM and CCM). However, in this context, assumption 7 is just a simplification, whose impact on the shape of the input current, quantitatively expressed by its THD, needs to be assessed. Being the ringing current at the turn-on instant of the power switch equal to the initial current during the on-time interval, this current may or may not be zero, which has an impact on the shape of the input current and, consequently, on its THD too.

When $V_{in} > V_R$, in QR-mode, the turn-on occurs at the first valley of the ringing in the drain voltage that follows the transformer demagnetization. Therefore, in this case, T_{neg} is equal to $T_r/2$.


Figure 2:4 (a) Simplified equivalent circuit during Tneg, (b) key waveforms in QR-mode, and DCM operations.

By neglecting the input capacitor C_s from the QR flyback schematic in Figure 1:7, the following considerations are valid: during T_{on} a charge Q_{pos} is provided from the input source and stored in the transformer; during T_{FW} , the energy is mostly delivered to the output; finally, in T_{neg} , a negative charge Q_{neg} is returned to the input source. The average input current during a switching cycle can, therefore, be expressed as:

$$\left\langle I_{p}\right\rangle = \frac{Q_{pos} - Q_{neg}}{T} \tag{12}$$

As shown in Figure 2:4, the positive charge Q_{pos} is clearly given by:

$$Q_{pos} = \frac{1}{2} I_{pkp} T_{on} \tag{13}$$

To evaluate the absolute value of the negative charge Q_{neg} , it is necessary to consider Equations (8)-(10)that describe the $V_{DS}(t)$ and $I_p(t)$ variations in the time interval T_{neg} (when QR operation is only considered) and distinguish two cases.

1. $V_{in} > V_R$. In the time interval (0, T_{neg}) $V_{DS}(t)$ is always greater than zero and the current $I_p(t)$ is sinusoidal; T_{neg} equals half the ringing period. The average value of $I_p(t)$ during T_{neg} is $2/\pi$ times the negative peak value $|I_{vyp}| = Y_L V_R$, therefore:

$$Q_{neg} = T_{neg} \frac{2}{\pi} Y_L V_R = \frac{T_r}{\pi} Y_L V_R = 2V_R C_{DS}$$
(14)

2. $V_{in} \leq V_R$. The current $I_p(t)$ is sinusoidal in the subinterval (0, T_z). T_z can be expressed as:

$$T_z = \frac{T_r}{2} \left(1 - \frac{1}{\pi} \cos^{-1} \left(\frac{V_{in}}{V_R} \right) \right)$$
(15)

and the current Ip(t) evaluated when t = Tz is:

$$I_{p}\left(T_{z}\right) = -Y_{L}V_{R}\sqrt{1 - \left(\frac{V_{in}}{V_{R}}\right)^{2}}$$
(16)

As depicted in Figure 2:4 (b), in the time interval T_{zz} , $I_p(t)$ ramps up linearly to zero. Hence, T_{zz} can be expressed as:

$$T_{zz} = \frac{L_p}{V_{in}} \left| I_p \left(T_z \right) \right| \tag{17}$$

Since T_{neg} is the sum of T_z and T_{zz} , it can be written:

$$T_{neg} = T_z + T_{zz} = \frac{T_r}{2} \left[1 + \frac{V_R}{V_{in}} \sqrt{1 - \left(\frac{V_{in}}{V_R}\right)^2} - \cos^{-1}\left(\frac{V_{in}}{V_R}\right) \right]$$
(18)

which is always greater than $T_r/2$, except when $V_{in} = V_R$. Finally, Q_{neg} is given by the sum of the two contributions, Q_{neg1} during the subinterval (0, T_z) and Q_{neg2} during the subinterval (T_z , T_{neg}). After some mathematical steps, Q_{neg} can be evaluated as:

$$Q_{neg} = Q_{neg1} + Q_{neg2} = \int_0^{T_z} Y_L V_R \cos\left(2\pi \frac{t}{T_r}\right) dt + \int_{T_z}^{T_{neg}} I_p\left(T_z\right) + \frac{V_{in}}{L_p} t \, dt = \frac{1}{2} C_{DS} \frac{\left(V_{in} + V_R\right)^2}{V_{in}}$$
(19)

Considering equation (12), the overall input current $I_{in}(\theta)$ can be found by adding the contributions obtained in Equations (7)and (13), which also takes into account the ringing oscillations along each line half-cycle. Hence it can be written as:

$$I_{in}(\theta) = \begin{cases} \frac{1}{2} I_{pkp}(\theta) \frac{T_{on}(\theta)}{T(\theta)} - \frac{2}{T(\theta)} V_R C_{DS} & V_{in} > V_R \\ \frac{1}{2} I_{pkp}(\theta) \frac{T_{on}(\theta)}{T(\theta)} - \frac{1}{2T(\theta)} \frac{(V_{in} + V_R)^2}{V_{in}} C_{DS} & V_{in} \le V_R \end{cases}$$
(20)

It is evident that the positive term in Equation (20) does not introduce any distortion, provided that the generic control method provides $I_{pkp}(\theta)$, as in Equation (6), and satisfies equation (4). The contribution of the ringing current, related exclusively to the negative terms, takes into account the ringing contributions derived in Equations (14)-(19). They represent a twofold effect: they downwards offset the input current waveform, which produces crossover distortion, and, considering that the offset is a function of the instantaneous line voltage, they distort its shape as well. It is worth noting that the distortion contribution is not ascribable to the control method; that is, it is not due to the control but is inherent in the power processing mechanism of any Hi-PF QR flyback converter. On the other hand, the control method may compensate or mitigate it. First of all, the contribution of the ringing current can be reduced by lowering the switching frequency (i.e., a longer $T(\theta)$, obtained with a larger L_p value) and using a low reflected voltage V_R . However, a larger L_p value implies a bigger flyback transformer, and a lower V_R increases the primary rms current and, consequently, the conduction losses. A trade-off is therefore required. A smaller C_{DS} also helps to reduce the ringing contribution, however, it is important to underline that a tradeoff between switching losses and EMI should be found. In fact, the lower the C_{DS} , the faster the V_{DS} transient at turn-off, and this faster transient may adversely affect both the efficiency of the power switch, as well as the EMI.

LED driver applications are typically specified to accommodate a certain range of output voltages V_{out} to power different types and lengths of LED string. Thus the contribution of the ringing current is expected to be maximum at the upper end of the V_{out} range and minimum at the lower end of the V_{out} range. In fact, lowering V_{out} will simultaneously reduce V_R and increase $T(\theta)$.

2.5 Crossover Distortion Due to the Input Capacitor

The storage capacitor *Cs* placed downstream of the input bridge, as depicted in Figure 1:7, is part of the EMI filter, which is always needed in an SMPS connected to the power line to restrict the conducted emission within the limits envisaged by the relevant EMI regulations. The capacitor has a twofold effect: it contributes to the voltage-current phase-shift and worsens the THD by maintaining a residual voltage on the dc side of the input bridge rectifier. The latter causes a non-conduction zone as the line voltage approaches zero, therefore, assumption 2 is no longer valid for the following analysis. This phenomenon also occurs for systems without a PFC and is an additional source of crossover itself, which interacts with the other distortion mechanisms. A quantitative analysis of this crossover distortion can be carried out independently from the topology employed in a PFC.

In these terms, any PFC and, consequently, the Hi-PF QR flyback converter can be modeled with an equivalent resistor (R_{eq}) that can be expressed as:

$$R_{eq} = \frac{V_{PK}^{2}}{2P_{in}}$$
(21)

where V_{PK} is the peak of the rectified line voltage $V_{in}(\theta)$ and P_{in} is the input power of the converter.

Generally speaking, the dead zone in $I_{ac}(\theta)$ increases at large C_s values, high line voltage, and low load (large R_{eq}). In quantitative terms, the dead zone starts, near the zero-crossing, when the rate of fall in the line voltage $V_{ac}(\theta)$ exceeds the rate of the voltage $V_{in}(\theta)$ across C_s , limited by the time constant $R_{eq} C_s$, so that from that instant on it is $V_{ac}(\theta) < V_{in}(\theta)$, as shown in Figure 2:5.

Focusing on the $V_{ac}(\theta)$ and $V_{in}(\theta)$ near the zero-crossing zone, shown in Figure 2:5(a), and neglecting the voltage drop across the input bridge rectifier, the phase angle α at which the slope of the two voltages are equal can be found with the following relation:

$$\left|\frac{dv_{ac}(\theta)}{d\theta}\right| = \left|\frac{dv_{in}(\theta)}{d\theta}\right| \implies 2\pi f_L V_{PK} \cos\alpha = \frac{1}{R_{eq}C_s} V_{PK} \sin\alpha$$
(22)

Solving for α :

$$\tan \alpha = 2\pi f_L R_{eq} C_s \tag{23}$$



Figure 2:5 Detail of $V_{ac}(\theta)$ (red trace) and $V_{in}(\theta)$ (blue trace) near voltage zero-crossing (a), and zoomed view of the yellow area, the intersection of $V_{ac}(\theta)$ and $V_{in}(\theta)$ (b). Values are normalized to V_{PK} .

From π - α on, $V_{in}(\theta)$ follows an exponential decay until the phase θ is equal to $\pi + \beta$, where it is again $|V_{ac}(\theta)| \ge V_{in}(\theta)$, which marks the end of the dead zone. In this interval the expression of $V_{in}(\theta)$ can be assumed as:

$$V_{in}(\theta) = \sin \alpha e^{-\frac{\theta - (\pi - \alpha)}{\tan \alpha}}$$
(24)

Beyond the angle $\pi + \beta$, $V_{ac}(\theta)$ and $V_{in}(\theta)$ are again both sinusoidal and overlapping until they have a phase shift equal to α away from the next zero-crossing. The duration of the dead zone is clearly $\alpha + \beta$.

The phase angle β can be found at the intersection of an exponential curve with a sinusoidal one, which results in a transcendental equation with no closed-form solution. Since β is small, it is possible to find an approximate solution β_a , substituting the last part of the exponential function (from $\theta = \pi$ to $\theta = \pi + \beta$) with its expansion in Taylor series to the first order $(e^{-\theta} \approx 1-\theta)$. In a similar way, it can be done for the rectified sinusoid ($sin \theta \approx \theta$). Therefore, the value Λ of the input voltage, evaluated when the phase angle is π , can be computed as follows:

$$\Lambda = V_{in}(\pi) = \sin \alpha e^{-\frac{\alpha}{\tan \alpha}} \approx \frac{\sin \alpha}{e}$$
(25)

As shown in Figure 2:5(b), at the intersection of the two straight lines, the phase angle is π + β_{a} .

$$\begin{cases} V_{in1}(\theta) = \Lambda \left(1 - \frac{\theta - \pi}{\tan \alpha} \right) \\ V_{in2}(\theta) = \theta - \pi \end{cases}$$

$$\beta_a = \frac{\Lambda \tan \alpha}{\Lambda + \tan \alpha}$$
(26)

The truncation of the exponential Equation (24) to the first order introduces an underestimation, as depicted in Figure 2:5(b), hence Equation (26) provides an approximated value β_a < β . On the other hand, if Equation (24) is approximated to the second-order, which provides a better approximation of equation (24), an overestimation will result, and the approximated value $\beta_a > \beta$ is less than 4% larger than the value provided by Equation (26), which proves that the accuracy of the first assumption is valid anyway.

The previous analysis shows that the storage capacitor C_s can be assumed as a source of crossover distortion, even if R_{eq} is a real resistor. The key point is the inability of $V_{in}(\theta)$ to keep pace with $V_{ac}(\theta)$ on the falling edge of the sinusoid, due to the maximum discharge rate of C_s through R_{eq} . However, its net impact is the result of the interaction with the other sources of distortion, previously analyzed.

In fact, it has been assumed that a resistor R_{eq} that represents the whole converter, which implicitly means that the ratio of $V_{in}(\theta)$ to $I_{in}(\theta)$ is constant in $(0, \pi)$; i.e., $I_{in}(\theta)$ is pure sinusoi-

dal in $(0, \pi)$. Actually, the distortion of the current shape, caused by all the previously considered sources, tends to reduce $I_{in}(\theta)$ with respect to the undistorted case. Instead, considering the same model approach, if it can be assumed that the R_{eq} changes along the sinusoid, then $R_{eq} = R_{eq}(\theta) = V_{in}(\theta)/I_{in}(\theta)$. Figure 2:6 shows $R_{eq}(\theta)$ obtained by dividing a sinusoidal input voltage and the current drawn that takes into account the distortion caused by the ringing current in Equation (20).



Figure 2:6 Variation of $R_{eq}(\theta)$ along the sinusoid for the prototype converter caused by the ringing current.

 $R_{eq}(\theta)$ is almost flat for a wide range of the sinusoid except when the current is approaching the zero crossings. The increase in $R_{eq}(\theta)$ produces an early dead zone with respect to the case with a fixed R_{eq} . The corresponding value of α would be the solution of the equation:

$$\tan \alpha = 2\pi f_L R_{eq}(\alpha) C_s \tag{27}$$

The solution of Equation (27) must be defined in the interval $(\pi/2, \pi)$, where $tan \alpha$ has a finite value and consequentially also $R_{eq}(\alpha)$. As a result, the dead zone caused by the input capacitor C_s starts before $R_{eq}(\theta)$ diverges, and $I_{ac}(\theta)$ must cross zero before $I_{in}(\theta)$. In other words, as long as the bridge rectifier conducts, $I_{ac}(\theta)$ is the sum of $I_{in}(\theta)$ and the current through C_s , which is essentially a sinusoid leading by 90 degrees. This causes $I_{ac}(\theta)$ to lead $I_{in}(\theta)$, whereas without C_s they would be essentially coincident.

Within the dead zone, the relation $R_{eq}(\theta) = V_{in}(\theta)/I_{in}(\theta)$ is no longer the one shown in the diagram of Figure 2:6 because of both the voltage $V_{in}(\theta)$ and the $I_{in}(\theta)$ are not sinusoidal. Besides, the voltage retained by C_s , which makes $V_{in}(\theta) > |V_{ac}(\theta)|$, and $T_{ON}(\theta)$ and $T(\theta)$ will be much smaller than those predicted during the control method design, which it has been carried out assuming a sinusoidal input voltage, as per Equation (5). The resulting switching frequency may be higher. As long as some energy is delivered to the output, C_s (which is the only source of energy, since no current comes from the reverse-biased bridge rectifier),

keeps on discharging and as long as $I_{in}(\theta) > 0$. If the peak current becomes lower than the critical peak primary current for no energy transfer, the energy is no longer transferred to the output and bounces back and forth between C_s and L_p , except for the energy dissipated during this process. As a consequence, C_s is discharged at a lower rate. Instead, if the peak current does not go below that critical value, the discharge rate of C_s has no change.

When $|V_{ac}(\theta)| \ge V_{in}(\theta)$ and the dead zone ends at $\theta = \beta$, it is possible to observe an abrupt variation in $I_{ac}(\theta)$. This is caused by the bridge rectifier that is forward-biased, as $I_{ac}(\theta)$ has to transition from zero to $I_{ac}(\beta)$, which is already greater than zero because of the leading phase of $I_{ac}(\theta)$.

It is worth mentioning that, near the zero-crossing interval, the input voltage is far from constant in a switching cycle and the switching frequency may come close to the frequency related to the resonance between C_s and L_p . So, the $V_{in}(\theta)$ has sinusoidal behavior, rather than constant. Additionally, the resonance of the inductors and capacitors in the EMI filter can also be stimulated, so that different resonance phenomena may coexist. This means that the modeling approach used throughout the present discussion is not always valid in describing the practical behavior.

2.6 Crossover Distortion Due to Transformer's Leakage Inductance

In practical cases, the real transformer's windings are not perfectly coupled. This phenomenon causes a crossover distortion that affects the input current shape and degrades the THD. Therefore, in the following subsection, the analysis of the distortion no longer takes into account assumptions 3 and 4.

In a real transformer, a portion of the energy stored in the primary winding cannot be transferred to the secondary winding because of imperfect magnetic coupling. In other words, this can be modeled by considering the primary inductance L_p split into two distinct elements: the magnetizing inductance L_M perfectly coupled to the secondary winding and the leakage inductance L_{lk} (uncoupled). In these terms, it is useful to define the coupling coefficient σ such that $L_M = \sigma L_p$ and $L_{lk} = (1 - \sigma) L_p$. Typical values of σ range from 0.95 to 0.99.

The energy stored in the leakage inductance is not transferred to the output and this energy would overcharge C_{DS} well over $V_{in}(\theta) + V_R$, in most cases exceeding the voltage rating of the power switch. A typical solution is to use a clamp circuit that limits $V_{DS}(t)$ at a well-defined and properly selected value V_{CL} (> V_R) above $V_{in}(\theta)$. Figure 2:7 shows the equivalent circuit of the Hi-PF QR flyback converter during the off time and the current waveforms underlining the effect of L_{lk} .

When the power switch is turned off, the primary current firstly charges the capacitance C_{DS} , thus the $V_{DS}(t)$ evolves until the voltage of the magnetizing inductance L_M , V_M , equals $-V_R$. During this time interval T_{ps} , the primary current I_p is flowing through L_M and L_{lk} , so there is an inductive voltage divider. Therefore:

$$V_{M} = -\frac{L_{M}}{L_{p}}V_{p} = \frac{L_{M}}{L_{p}}\left[V_{in}\left(\theta\right) - V_{DS}\left(t\right)\right]$$
(28)



Figure 2:7 (a) Simplified equivalent circuit during off time. (b) Current in the real primary windings (blue trace) and the secondary windings (black trace).

In the instant when the voltage V_M equals $-V_R$, the voltage V_p across the primary winding can be expressed as:

$$V_p = \frac{L_p}{L_M} V_R = \frac{1}{\sigma} V_R \tag{29}$$

Immediately after the time interval T_{ps} , the current starts flowing through D1, and the energy starts being transferred to the output. $V_{DS}(t)$ keeps on ramping up until it reaches $V_{in}(\theta) + V_{CL}$ in a time T_{LK} . After that, $V_{DS}(t)$ is clamped, L_{lk} starts being demagnetized with a rate equal to $(V_R - V_{CL})_{Llk}$, and the current through D1 reduces at the same rate. The portion of the primary current $I_p(t)$ of the leakage inductance varies from the peak value, $I_{pkp}(\theta)$, until zero in a time interval equal to $T_{LK}(\theta)$. Hence:

$$I_{p}(t) = I_{pkp}(\theta) - \frac{V_{CL} - V_{R}}{(1 - \sigma)L_{p}}t$$
(30)

As soon as $I_p(t)$ is equal to zero, it can be written in the analytical expression of T_{LK} :

$$T_{LK}(\theta) = \frac{(1-\sigma)L_p}{V_{CL} - V_R} I_{pkp}(\theta)$$
(31)

Bearing in mind the current that flows in L_M can be written as:

$$I_M(t) = I_{pkp}(\theta) - \frac{V_R}{\sigma L_p}t$$
(32)

Meanwhile, the secondary current reaches its peak value $I_{pks}(\theta)$:

$$I_{pks}(\theta) = nI_M(T_{LK}) = n \left[1 - \frac{V_R}{V_{CL} - V_R} \frac{1 - \sigma}{\sigma} \right] I_{pkp}(\theta)$$
(33)

After that, the magnetizing inductance on the secondary side starts being demagnetized at a rate V_{R/L_M} until it zeroes in a time T_{DEM} , as shown in Figure 2:7. The presence of L_{lk} splits the time interval T_{FW} in a first subinterval T_{LK} needed to demagnetize the leakage inductance (and magnetize the secondary winding) and a second subinterval T_{DEM} needed to demagnetize the secondary winding. It is possible to prove that the resulting T_{FW} can be expressed as:

$$T_{FW}(\theta) = \frac{L_M}{V_R} I_{pkp}(\theta) = \sigma \frac{L_p}{V_R} I_{pkp}(\theta)$$
(34)

(i.e., the one calculated neglecting the leakage inductance multiplied by the coupling coefficient σ).

The presence of L_{lk} clearly compromises the input-to-output transfer energy, thus the dead zone where there is no energy transfer is expected to occur over a wider phase angle range of the line voltage. In fact, the "no energy transfer" condition concerns only the portion of the primary voltage V_p across the magnetizing inductance L_M . Combining Equations (8) and (9) and (28), it is possible to find that the condition for no energy transfer, solved for $I_{pkp}(\Theta)$, yields:

$$I_{pkp}\left(\theta\right) \le Y_L \sqrt{\left(\frac{V_R}{\sigma}\right)^2 - {V_{in}}^2\left(\theta\right)}$$
(35)

The critical peak primary current for no energy transfer, expressed by Equation (35), is larger than the ideal case ($\sigma = 1$), thus confirming the forecast of a wider dead zone related to this phenomenon.

It is noteworthy to compare the critical value of $I_{pkp}(\theta)$ in Equation (35) to that determined by the ringing current after demagnetization and the relative positions of the dead zones they generate. Bearing in mind that the on-time can be written as:

$$T_{ON} = \frac{L_p}{V_{in}} I_{pkp}$$
(36)

The region around zero crossings where there is no input-to-output energy transfer ($I_{in}(\theta) = 0$), can be written by considering the combination of the input current in Equation (20) in the case of $V_{in} < V_R$, and Equation (36):

$$I_{pkp} = \frac{(V_{in} + V_R)^2}{V_{in}} \frac{C_{DS}}{L_p} \frac{V_{in}}{I_{pkp}}$$
(37)

Combining Equations (10)-(37), the ringing current after demagnetization related to the dead zone can be expressed as:

$$I_{pkp} = \frac{(V_{in} + V_R)^2}{V_{in}} \frac{C_{DS}}{L_p} \frac{V_{in}}{I_{pkp}}$$
(38)

Figure 2:8 shows the ratio of the critical value of $I_{pkp}(\theta)$ for no input-to-output energy transfer, as in Equation (35), to that caused by the ringing current, as in Equation (38). The ratio is expressed as a function of the parameter K_v :

$$K_{v} = \frac{V_{PK}}{V_{R}}$$
(39)

where V_{PK} is the peak value of the rectified line voltage $V_{in}(\theta)$. The dashed blue line represents the ideal case. As is evident from Figure 2:8, the dead zone becomes even larger near the zero crossings. Therefore, the lack of input-to-output energy transfer may override the dead zone caused by the ringing current needed for the demagnetization of the secondary winding in a time T_{DEM} .



Figure 2:8 Ratio of the critical value of $I_{\rho k \rho}(\theta)$ for no input-to-output energy transfer in the dead zone, caused by leakage inductance, as in Equation (35), to that caused by ringing current, as in equation 38, as a function of Kv for σ = 0.95.

2.7 Guidelines for Designing High Power Factor Quasi-Resonant Flyback LED Drivers

The QR flyback shows the highest figure of merit (FOM) among the different LED driver solutions analyzed in the literature. With this aim, in the previous sections have been analyzed the different inherent causes of distortion in the input current and the power processing mechanism of the proposed QR flyback converter that are not ascribable to the control method.

Several effects, such as the distortion caused by the ringing current, crossover distortion due to transformer leakage inductance, and crossover distortion due to the input storage capacitor, have been analyzed in depth. It is important to point out that the converter is a nonlinear system and, consequentially, the overall THD of the input current cannot be evaluated as a simple sum of the THD associated to each individual contribution, where the aforementioned issues are mutually interacting or partly overlapping, such as those creating a dead zone in the line current.

Some useful design hints have been extracted and discussed in order to provide some suggestions for an optimized design of a QR flyback LED driver converter. In detail, the following list summarizes all the possible precautions to bear in mind.

1. The impact of the ringing current after transformer demagnetization can be mitigated by lowering the switching frequency, using a low reflected voltage VR or choosing a power MOSFET with a RDS(on) with an optimized RDS(on)/ C_{oss} FOM. These criteria also help to

reduce the phenomenon of the lack of input-to-output energy transfer near the zero crossings of the line voltage.

2. The leakage inductance of the transformer should be kept as low as practically possible. This choice essentially optimizes the converter efficiency but does not impact the reduction in the dead zones near the zero crossings of the line voltage caused by other phenomena (essentially, the input capacitor C_s).

3. The input storage capacitor C_s should be minimized to reduce the dead zone near the line voltage zero crossings and the current leap occurring in the proximity of the dead zone. However, particular attention should be paid to the following points.

a. The diodes of the input bridge rectifier are usually slow-recovery ones, so the primary current at the switching frequency may require an enhanced filter on the ac side of the bridge and may cause the diodes of the bridge to overheat.

b. Close to the zero crossings, the switching frequency can be very low. If the ringing frequency related to C_s and L_p is comparable with the switching one, it may generate current spikes that would degrade the current THD.

4. Class-X capacitors are generally used along with inductors for EMI filtering, necessary for the certification of the final product. Class-X capacitors can degrade the PF, although they do not contribute to the THD. From this perspective, on the one hand, the design of the filter must make the device compliant with the standards. On the other hand, there is a degree of freedom that can be exploited to minimize PF lowering at the high line and light load. The filters should be designed with the largest inductance and the smallest capacitance practically possible.

Chapter 3 Circuit model of LED light bulb

3.1 Introduction

Nowadays, the investigation of the current harmonics prediction in the case of a lighting system generates a lot of interest. With a large use of LED light bulbs, additional issues will involve in the future smart grids that use network control and management techniques based on measurements that may be affected by the harmonics. The problem is exacerbated by the LEDs since dealing with the large distorted current due to a very great number of dispersed LEDs is more complex than in the case of a single large harmonics source. Therefore, there is a noteworthy need for mitigation of the harmonics due to LED-based lighting systems. The addition of filters, power factor corrector circuits, and other devices in the ESL increases the lamp cost and consequently the economic investment for both retrofitting and new lighting systems installation. On the other hand, although a single compensation circuit for the whole lighting system may be less expensive, it has to be revised each time a new lighting system retrofit is performed.

The best choice from cost and flexibility point of view is the harmonic mitigation at the lighting system design stage. The optimal lighting system design aiming at mitigation of the distorted current drawn from the whole lighting system requires a tool for estimating such a current when different lamps configurations are adopted (number, type, nominal parameter, and so on). In such a case the best configuration is the one with the greatest harmonic cancellation and lowest THD.

In this perspective, the proposed parametric Pspice circuit is suitable to perform this task. This chapter describes the proposed equivalent circuit which is able to model the current drawn of any LED light bulb, whatever is the lamp driver, provided that the voltage THD does not exceed the limit of IEEE-Std-519. The measurements of the current drawn from several LED light bulbs from different manufacturers have confirmed the presence of odd harmonics while even harmonics, subharmonics, and inter-harmonics are almost negligible. Moreover, the measurements have also confirmed a low voltage total-harmonic-distortion,

which is less than 4%. Therefore, in the following, firstly, the key equations of non-sinusoidal periodical, instantaneous electric quantities in the absence of subharmonics and interharmonics have been discussed.

After that, the rules and equations to obtain the components of the equivalent circuit of the LED light bulbs have been provided. The procedure to obtain the quantities to be used in the equations have been also described. Such a procedure with the aforesaid rules and equations enable to obtain the model of any LED light bulbs available on the market without the knowledge of its inner components (driver, LED package, and so on). Therefore, any researcher and practitioner can treat any LED light bulbs in the market as a black-box, and the components of the equivalent circuit model have been simply obtained by some electrical measurements. It is noting that the parameters of a generic LED light bulb can be obtained by a few simple current measurements to be performed at the lamp terminals (such as the active and reactive power, and some current harmonics that are measured on the line at different values of the voltage on the mains). These parameters are the coefficients of the functions obtained by linear or polynomial interpolation of the *rms* of the current harmonics.

Finally, the proposed parametric Pspice circuit that is able in emulating the current drawn by the lamps under variable voltage on the mains has been widely discussed. Then, the validation has been performed by comparing the waveforms and harmonics of the measured currents with the simulated ones for several LED light bulbs. The validation has confirmed the effectiveness and robustness of the proposed model.

3.2 Non-sinusoidal periodical electric quantities

Considering a circuit operating at steady-state conditions, the non-sinusoidal periodical instantaneous electric quantities, that is the voltage v(t) and the current i(t), can be represented by means of the Fourier series in the absence of subharmonics and inter-harmonics [11]:

$$v(t) = V_0 + \sqrt{2} \sum_{k \in \mathbb{N}} V_h \sin(h\omega t + \alpha_k)$$

$$i(t) = I_0 + \sqrt{2} \sum_{k \in \mathbb{N}} I_h \sin(h\omega t + \beta_k)$$
(40)

Key terms are the power system fundamental frequency of the voltage, $v_1(t)$, and current, $i_1(t)$ [11]:

$$v_1(t) = \sqrt{2}V_1 \sin(\omega t + \alpha_1)$$

$$i_1(t) = \sqrt{2}I_1 \sin(\omega t + \beta_1)$$
(41)

Grouping the constant term with the harmonics [11]:

$$v_{H}(t) = V_{0} + \sqrt{2} \sum_{k \in \mathbb{N}} V_{k} \sin(k\omega t + \alpha_{k})$$

$$i_{H}(t) = I_{0} + \sqrt{2} \sum_{k \in \mathbb{N}} I_{k} \sin(k\omega t + \beta_{k})$$
(42)

any non-sinusoidal periodical instantaneous electric quantity can be summarized according to the IEEE Std-1459 [11]:

$$v(t) = v_1(t) + v_H(t)$$

 $i(t) = i_1(t) + i_H(t)$
(43)

Without loss of generality, it can be set:

$$\alpha_k = \delta_k + \frac{\pi}{2}$$

$$\beta_k = \gamma_k + \frac{\pi}{2}$$
(44)

and also $\delta_1 = 0$. Therefore:

$$v_{1}(t) = \sqrt{2}V_{1}\cos(\omega t)$$

$$v_{k}(t) = \sqrt{2}V_{k}\cos(k\omega t + \delta_{k})$$

$$i_{1}(t) = \sqrt{2}I_{1}\cos(\omega t + \gamma_{1})$$

$$i_{k}(t) = \sqrt{2}I_{k}\cos(k\omega t + \gamma_{k})$$
(45)

The instantaneous power related to the fundamental frequency deriving from these equations is:

$$p_{1}(t) = v_{1}(t)i_{1}(t) = V_{1}I_{1}\cos(\gamma_{1})\left[1 + \cos(2\omega t)\right] - V_{1}I_{1}\sin(\gamma_{1})\sin(2\omega t) =$$

$$= P_{1}\left[1 + \cos(2\omega t)\right] - Q_{1}\sin(2\omega t)$$
(46)

and the fundamental apparent power, S_1 , is then obtained from the fundamental active power, P_1 , and the fundamental reactive power, Q_1 , through the following equation [11]:

$$S_1^2 = P_1^2 + Q_1^2 \tag{47}$$

that, in turn, is part of the overall apparent power, *S*, that includes the harmonics contribution (that is the non-fundamental apparent power) [11]:

$$S^2 = S_1^2 + D_I^2 + D_V^2 + S_H^2$$
(48)

where, the current distortion power, D_I , the voltage distortion power, D_V , and the harmonic apparent power, S_H , are obtained from the evaluation of the *rms* value of $v_H(t)$, that is V_H , and $i_H(t)$, that is I_H [11]:

$$D_{I}^{2} = V_{1}^{2} I_{H}^{2}$$

$$D_{V}^{2} = I_{1}^{2} V_{H}^{2}$$

$$S_{H}^{2} = V_{H}^{2} I_{H}^{2}$$
(49)

The harmonics related to the voltage on the mains are negligible in comparison with the current harmonics, it is also confirmed by the measurements. Consequently, the term $v_H(t)$ in (43) can be neglected, then the apparent power is approximated to:

$$S^2 = P_1^2 + Q_1^2 + D_I^2$$
(50)

Therefore, the modeling of the current harmonics enables to estimate the non-active powers Q1 and D_1 that involve undesired power losses in the line. It is worth noting that, while the current harmonics pertain to the undesired term D_1 , the fundamental current affects the active power as well as the useless power Q_1 . Therefore, it is useful to consider the fundamental current divided into two components [89-90]:

$$i_{1}(t) = i_{P1}(t) + i_{Q1}(t)$$

$$i_{P1}(t) = \sqrt{2}I_{P1}\cos(\omega t) = \sqrt{2}I_{1}\cos(\gamma_{1})\cos(\omega t)$$

$$i_{Q1}(t) = \sqrt{2}I_{Q1}\cos\left(\omega t + \frac{\pi}{2}\right) = \sqrt{2}I_{1}\sin(\gamma_{1})\cos\left(\omega t + \frac{\pi}{2}\right)$$
(51)

It is easy to prove that:

$$v_{1}(t)i_{P1}(t) = V_{1}I_{1}\cos(\gamma_{1}) \left[1 + \cos(2\omega t)\right] = P_{1}[1 + \cos(2\omega t)]$$

$$v_{1}(t)i_{Q1}(t) = V_{1}I_{1}\sin(\gamma_{1}) \left[-\sin(2\omega t)\right] = -Q_{1}\sin(2\omega t)$$
(52)

Therefore, the term $i_{P1}(t)$ affects only the active power, then in the following, it is called "active current". The term $i_{Q1}(t)$, affects only the reactive power, that it is called "reactive current". Finally, as expected, in all performed tests for all the lamps, the phase shift of the

fundamental frequency current drawn has been always $-\pi/2 <\gamma_1 < \pi/2$: (with respect to the voltage on the mains), which implies a positive value of $\cos(\gamma_1)$ in (51).

3.3 Linear model for emulating the current drawn from LED light bulb

3.3.1 Description of the linear model

The measurements of the amplitude and phase shift of the fundamental and harmonic currents have to be performed in order to obtain the components of the proposed Pspice circuit. For a given LED light bulb, these measurements have to be performed by setting five voltage levels at the lamp terminals: $0.9V_{nom}$, $0.95V_{nom}$, V_{nom} , $1.05V_{nom}$, and $1.1V_{nom}$. For each current harmonic, a polynomial function has to be obtained by measurement interpolation. Similarly, polynomial functions have to be obtained for, respectively, the active and reactive current. The coefficients of these polynomial functions have to be used to select the components of the Pspice circuit according to the rules that are described in the next section. Moreover, the quantities related to these components are also related to these coefficients as described in the following. With this in mind, this section describes the test rig used to set the desired voltage levels across the lamp terminals and to perform the aforesaid measurements.

Figure 3:1 depicts the measurements of the fundamental current and the odd current harmonic amplitudes up to the 13th for a LED light bulbs. The measurements highlighted that a linear interpolation of the current harmonics turns out to be enough accurate:

$$I_{P1} \stackrel{\text{def}}{=} I_{P1}(V_1) = a_{P1} + b_{P1}V_1$$

$$I_{Q1} \stackrel{\text{def}}{=} I_{Q1}(V_1) = a_{Q1} + b_{Q1}V_1$$

$$I_k \stackrel{\text{def}}{=} I_k(V_1) = a_k + b_kV_1$$

$$k = 2, 3, \dots K$$
(53)

with K the number of harmonics considered for emulating the waveform of the current drawn from the lamp.

Remark 1 - I_{P1} is always positive regardless of its trend in (53) because it is obtained by multiplying I_1 with $cos(\gamma_1)$, where the former is positive by definition while the latter has been previously proved to be positive since $-\pi/2 < \gamma_1 < \pi/2$. Similar considerations are valid for I_k as well as for the terms in the related interpolation functions. On the other hand, although I_1 is positive by definition, I_{O1} may be positive or negative, it depends on γ_1 . Remark 2 - Considering the previous remark and that V_1 is positive by definition, at least one between a_{P1} and b_{P1} in the interpolation function have to be positive. Similar considerations are always valid for a_k and b_k , while are valid for a_{Q1} and b_{Q1} only when I_{Q1} is positive. On the other hand, when I_{Q1} is negative, a dual behavior occurs, that is at least one between a_{O1} and b_{O1} in the interpolation function have to be negative.



Figure 3:1 Measured amplitude of the fundamental current and greatest harmonics for a LED. The linear interpolation functions are reported on the right with the related current symbol.

In the test rig (Figure 3:2), a transformer with a continuous variable transform ratio (variac) has been connected to the mains voltage with the aim of emulating the variable voltage on the mains. In other words, each time a measurement has to be performed at a given voltage level, the variac has been properly tuned to emulate such a voltage level across the lamp terminals. The variac feeds the LED light bulb through a power analyzer used to measure the voltage across the lamp terminals as well as to measure the fundamental and harmonic currents (in terms of amplitude and phase shift) drawn. Moreover, to simultaneously observe the current harmonics amplitude and the waveforms of the voltage and current at LED light bulb through the oscilloscope. The power analyzer has a 1 A shunt probe that offers a high resolution and accuracy for testing currents as low as 80 μ A. This enables the meter to measure standby power as low as 20 mW at 240 V. The maximum voltage peak can reach 2kV, with an accuracy of 20mV. The power analyzer has a bandwidth of 1 MHz.



Figure 3:2 Experimental setup. A variac, connected to the mains, feeds the LED light bulb through a power analyzer that acquires the measurements. The oscilloscope displays the waveforms.

The measurements have been performed in the range of $\pm 10\%$ of the rated voltage due to the voltage variation that is allowed for the utility [91]. Throughout the day, the rated voltage could suffer from some variations due to the distributed renewable generators into the network, the load variations, the network configurations, and so on, without the possibility for the users to do anything about it [92].

Figure 3:3 shows the amplitude of the current harmonics normalized with respect to the fundamental one, in the case of a LED light bulb with 212V (*rms*) at its terminals. The normalized harmonic amplitudes, in blue, are sorted in ascending order and the abscissa reports the related harmonic. The figure also reports, in red, the THD error in percentage when some current harmonics are neglected, assuming as THD reference, *THD*_{1,TOT}, the one computed considering the harmonics until the 50th. More specifically, for a given harmonic, *n*, the THD error, *Error*_{1,n}, is obtained when a set of *m* harmonics are neglected. Harmonic *n* and those on its left in the figure belong to this set. The current THD reference is:

$$THD_{I,TOT} \stackrel{\text{def}}{=} \sqrt{\frac{\sum_{k=2}^{K} {I_k}^2}{{I_1}^2}}$$
(54)

while the THD referred to a given current harmonic n, $THD_{l,n}$, is computed subtracting the aforesaid m current harmonic amplitudes:

$$THD_{I,n} = \sqrt{\frac{\sum_{p=m+1}^{K} \hat{I_p}^2}{{I_1}^2}}$$
(55)

with \hat{I} an array where the amplitudes of the current harmonics have been ascending sorted and p indicates the position in the sequence. Consequentially the percentage error referred to the $THD_{l,n}$ is equal to:

$$Error_{I,n}(\%) = 100 \frac{THD_{I,TOT} - THD_{I,n}}{THD_{I,TOT}}$$
(56)

In the THD error diagram, the first red point on the left has been calculated by removing from the total current the harmonic that has the smallest amplitude (that is the 50th), the second point is calculated neglecting the two smallest current amplitude (that is the 50th and 48th) and so on until it is removed the 3-rd current harmonic.

Figure 3:3 depicts the normalized current harmonics and the THD percentage error for a generic LED light bulb. It is less to 1% by removing the smallest m=41 (that is n=15) current harmonics (that is considering only the 3rd, 5th, 7th, 9th, 11th, 13th, 17th, and 19th harmonic). On the other hand, by removing m=46 harmonics, n=9 (considering only the 3rd, 5th, and 7th), the percentage error is less than 10%. Figure 3:4 reports the quantities for the LED when the voltage is 237V.



Figure 3:3 Normalized current harmonics and THD error, when the LED light bulb is feed with 212V.



Figure 3:4 Normalized current harmonics and THD error, when the LED light bulb is feed with 237V.

The proposed circuit model accounts for the change of the current drawn from a LED light bulb when the *rms* voltage, V_1 , on the mains varies within the range allowed by the regulation (±10% of V_{nom}). More specifically, the model emulates the variation of the value of the *rms*, I_1 , and the change in the phase shift, γ_1 , of the fundamental frequency current, $i_1(t)$. The model also emulates the variation of the *rms* of any other current harmonic, I_k , while it neglects any variation of the phase offset, γ_k (that is the phase offset at nominal voltage is considered). A graphical representation of the proposed circuit that accounts for the previous interpolation functions is reported in Figure 3:5.



Figure 3:5 Proposed linear equivalent circuit of a generic LED light bulb.

The components in the subcircuits called "Active", "Reactive" and "k-th harmonic" emulate the behavior of, respectively, I_{P1} , I_{Q1} and I_k in (53). According to the circuit model in the figure and the equations in (53), the following equations are valid:

$$\underbrace{interpolation}_{i_{aP1}(t) + i_{bP1}(t)} = i_{P1}(t) = \sqrt{2}I_{P1}\cos(\omega t) = \sqrt{2}\underbrace{[a_{P1} + b_{P1}V_{1}]}_{[a_{P1} + b_{P1}V_{1}]}\cos(\omega t)$$

$$i_{aQ1}(t) + i_{bQ1}(t) = i_{Q1}(t) = \sqrt{2}I_{Q1}\cos\left(\omega t + \frac{\pi}{2}\right) = \sqrt{2}[a_{Q1} + b_{Q1}V_{1}]\cos\left(\omega t + \frac{\pi}{2}\right) \quad (57)$$

$$i_{ak}(t) + i_{bk}(t) = i_{k}(t) = \sqrt{2}I_{k}\cos(k\omega t + \gamma_{k}) = \sqrt{2}[a_{k} + b_{k}V_{1}]\cos(k\omega t + \gamma_{k})$$

$$k = 2, 3, ... K$$

Then $i_{aP1}(t)$ accounts for the constant term a_{P1} , $i_{bP1}(t)$ accounts for the linear term b_{P1} and so on:

$$i_{aP1}(t) = \sqrt{2}a_{P1}\cos(\omega t)$$

$$i_{bP1}(t) = \sqrt{2}b_{P1}V_1\cos(\omega t)$$

$$i_{aQ1}(t) = \sqrt{2}a_{Q1}\cos\left(\omega t + \frac{\pi}{2}\right)$$

$$i_{bQ1}(t) = \sqrt{2}b_{Q1}V_1\cos\left(\omega t + \frac{\pi}{2}\right)$$

$$i_{ak}(t) = \sqrt{2}a_k\cos(k\omega t + \gamma_k)$$

$$i_{bk}(t) = \sqrt{2}b_kV_1\cos(k\omega t + \gamma_k)$$

In the previous equations, the currents $i_{aP1}(t)$, $i_{aQ1}(t)$ and $i_{ak}(t)$ are independent from V_1 since they arise from the constant terms in (53). Therefore, their waveforms are emulated by means of independent current generators in the equivalent circuit (Figure 3:5).

It is useful to recall that when a constant term is negative (for example $a_k < 0$) the related generator ($i_{ak}(t)$ in such an example) is in antiphase with the overall current it belongs to ($i_k(t)$ in such an example). Consequently, according to *Remark 2*, the other term (b_k in such an example) is definitively positive because the related current ($i_{bk}(t)$ in such an example) has to be in phase with the overall current. Moreover, the amplitude of the "in-phase" current ($i_{bk}(t)$) is greater than the amplitude of the "antiphase" current ($i_{ak}(t)$) according to *Remark 1*.

The voltage-independent generator, v_{Fk} , with an element in the series, is responsible for emulating the current $i_{bk}(t)$. The voltage-independent generator has an angular frequency k times greater than the mains voltage and the same amplitude:

$$v_{Fk}(t) = \sqrt{2}V_1 \cos\left(k\omega t + \gamma_k + sign(b_k)d\frac{\pi}{2}\right)$$
(59)

where γ_k is the phase offset of the *k*-th harmonic current according to the equations in (45); *d* can assume only two values +1 or -1, it depends on the component selected by the switches.

When the circuit component adopted is an inductor then d=1, otherwise, when it is chosen a capacitor d=-1. It is worth noting that this voltage generator does not emulate the k-th voltage harmonic on the mains but it is a fictitious generator belonging to the lamp model.

By using the superposition theorem, it can be noted that such a fictitious voltageindependent generator supplies only the aforesaid component, e.g. the inductor when d=1(see Figure 3:6). Therefore, at steady-state, when an inductor L_k is adopted, the steady-state current through it due to v_{Fk} is:

$$i_{Lk}(t) = \frac{\sqrt{2}V_1}{k\omega L_k} \cos\left(k\omega t + \gamma_k + sign(b_k)\frac{\pi}{2} - \frac{\pi}{2}\right)$$
(60)

When b_k is positive, the phase of $i_{bk}(t)$ is $k\omega t + \gamma_k$ and the previous equation becomes:

$$i_{Lk}(t) = \frac{\sqrt{2}V_1}{k\omega L_k} \cos(k\omega t + \gamma_k)$$
(61)

that is $i_{Lk}(t)$ presents the same phase of $i_{bk}(t)$. These currents can present also the same amplitude by properly choosing L_k :

$$L_k = \frac{1}{k\omega b_k} \xrightarrow{\text{yields}} i_{Lk}(t) = i_{bk}(t) \text{ when } b_k > 0$$
(62)

When b_k is negative, the phase of $i_{bk}(t)$ is $k\omega t + \gamma_k - \pi$ since this current is in antiphase with $i_k(t)$, moreover (60) becomes:

$$i_{Lk}(t) = \frac{\sqrt{2}V_1}{k\omega L_k} \cos(k\omega t + \gamma_k - \pi)$$
(63)

Once again $i_{Lk}(t)$ presents the same phase offset of $i_{bk}(t)$ thanks to the adopted waveform of the fictitious voltage generator. These currents can present also the same amplitude by properly choosing L_k :

$$L_k = \frac{1}{k\omega(-b_k)} \xrightarrow{\text{yields}} i_{Lk}(t) = i_{bk}(t) \text{ when } b_k < 0 \tag{64}$$

Therefore, by using a fictitious voltage generator with the waveform reported in (59) and an inductor is always possible to emulate the linear term b_k :

$$L_{k} = \frac{1}{k\omega|b_{k}|} \xrightarrow{\text{yields}} i_{Lk}(t) = i_{bk}(t) \forall b_{k} \neq 0$$
(65)

When b_k is equal to 0, the amplitude of I_k is independent from the voltage on the mains, then only the independent current generator is considered while the fictitious independent voltage generator and the inductor are removed.



Figure 3:6 Equivalent circuit when only vFk works while the other independent generators are turned off. It is apparent that this generator supplies only Lk.

When a capacitor C_k is adopted (then d=-1) the steady-state current through it due to v_{Fk} is:

$$i_{Ck}(t) = k\omega C_k \sqrt{2} V_1 \cos\left(k\omega t + \gamma_k - sign(b_k)\frac{\pi}{2} + \frac{\pi}{2}\right)$$
(66)

When b_k is positive the previous equation becomes:

$$i_{Ck}(t) = k\omega C_k \sqrt{2} V_1 \cos(k\omega t + \gamma_k)$$
(67)

that is $i_{Ck}(t)$ presents the same phase of $i_{bk}(t)$ like the previous case where an inductor has been considered. These currents can present also the same amplitude by properly choosing C_k :

$$C_k = \frac{b_k}{k\omega} \xrightarrow{\text{yields}} i_{Ck}(t) = i_{bk}(t) \text{ when } b_k > 0$$
(68)

and more in general, also using the fictitious voltage (59) and a capacitor is always possible to emulate the linear term b_k :

$$C_{k} = \frac{|b_{k}|}{k\omega} \xrightarrow{\text{yields}} i_{Ck}(t) = i_{bk}(t) \forall b_{k} \neq 0$$
(69)

In [89] and [90] have been presented, respectively, a circuit model (called "fundamental current circuit", Figure 3:5) of the active and reactive current as well as the related Pspice model. The model of the fundamental current has been slightly modified in this work and for the sake of completeness, the main considerations and relations are reported in the following.

The active current has to be in phase with the mains voltage since the lamp is a load. Notwithstanding, when the constant term, a_{P1} , is negative the independent current generator is in antiphase with the mains voltage. On the other hand, in such a case, the term b_{P1} has to be positive according to Remark 2, that is the current i_{bP1} must be in phase with the mains voltage. Moreover, its amplitude has to exceed the other in the voltage range of application (±10% of V_{nom}). Similarly, when the linear term, b_{P1} , is negative the current i_{bP1} is in antiphase with the mains voltage, but the term a_{P1} is positive (according to Remark 2) and enough to ensure that the overall active current is in phase with the mains voltage in the range of application of the model. It is worth to note that negative values of b_{P1} highlights a reduction of the active current drawn from the lamp as the mains voltage increases. Therefore, in the Pspice circuit a resistor is adopted when i_{bP1} is positive (to emulate an increasing current in phase with the mains voltage), otherwise, a voltage-controlled current source (VCCS), G1, is considered:

$$R_{1} = \frac{1}{b_{P1}} \quad when \quad b_{P1} > 0 \tag{70}$$

$$VCCS \ gain = b_{P1} \quad when \quad b_{P1} < 0$$

Finally, no issues arise when both terms are positive. In such a case the independent generator is in phase with the mains voltage and a resistor is adopted to emulate the increasing active current drawn by the lamp as the voltage on the mains increases.

When I_{Q1} is positive, the reactive current leads to the mains voltage. In such a case, if the constant term, a_{Q1} , is negative then the related independent current generator lags the mains voltage. Therefore, the term b_{Q1} has to be positive (according to Remark 2), that is the current i_{bQ1} must lead the mains voltage. Moreover, its amplitude has to exceed the other in the voltage range of application (±10% of V_{nom}). Similarly, when the linear term, b_{Q1} , is negative, the current i_{bQ1} lags the mains voltage, but the term a_{P1} is positive (according to Remark 2) and enough to ensure that the overall reactive current leads the mains voltage in the range of application of the model. It is worth to note that, in this specific case, negative values of b_{Q1} means a reduction of the reactive current drawn from the lamp as the mains voltage increases.

Dually, when I_{Q1} is negative, the reactive current lags the mains voltage. In such a case, if the constant term, a_{Q1} , is positive then the related independent current generator leads the

mains voltage. Therefore, the term b_{Q1} has to be negative (according to Remark 2), that is the current i_{bQ1} must lag the mains voltage. Moreover, its amplitude has to exceed the other in the voltage range of application (±10% of V_{nom}). Similarly, when the linear term, b_{Q1} , is positive, the current i_{bQ1} leads the mains voltage, but the term a_{P1} is negative (according to Remark 2) and enough to ensure that the overall reactive current lags the mains voltage in the range of application of the model. It is worth noting that, in this specific case, a positive value of b_{Q1} means a reduction of the reactive current drawn from the lamp as the mains voltage increases. Figure 3:1 represents graphically the previous cases.

Whatever I_{Q1} , a capacitor may be adopted in the model when i_{bQ1} leads the mains voltage and an inductor when i_{bQ1} lags the mains voltage:

$$C_{1} = \frac{b_{Q1}}{\omega} \quad \text{when} \quad b_{Q1} > 0$$

$$L_{1} = \frac{1}{\omega(-b_{Q1})} \quad \text{when} \quad b_{Q1} < 0$$
(71)

It is worth to note that there is a crucial difference between these components and the others described before (L_k and C_k). More specifically, when b_{QI} is positive a capacitor must be used in the Pspice circuit, while when b_k is positive L_k and C_k can be equally used in the Pspice model. On the other hand, once the component (L_k or C_k) is chosen the waveform of the fictitious independent voltage generator cannot be chosen arbitrarily since the value of *d* is fixed by the component choice.

By using the superposition theorem, it can be noted that the voltage-independent generator representing the mains voltage supplies the "fundamental current circuit" and also the components (L_k , C_k) adopted for emulating the current harmonics. Figure 3:7 easily makes evident such a consideration. Therefore, undesired reactive currents flow through these components. These currents have to be eliminated to ensure that the reactive current is given only by the "fundamental current circuit". This goal is reached by adding a compensation component (L_c or C_c) which is resonant at the fundamental frequency with the equivalent component (C_{EQ} or L_{EQ}) downstream form it:

$$L_{EQ} = \frac{1}{\sum_{k=2}^{K} k\omega |b_k|} \Rightarrow C_C = \frac{1}{\omega^2 L_{EQ}} = \frac{\sum_{k=2}^{K} k|b_k|}{\omega}$$

$$C_{EQ} = \sum_{k=2}^{K} \frac{|b_k|}{k\omega} \Rightarrow L_C = \frac{1}{\omega^2 C_{EQ}} = \frac{1}{\omega \sum_{k=2}^{K} \frac{|b_k|}{k}}$$
(72)



Figure 3:7 Equivalent circuit when only v1 works while the other independent generators are turned off. It is apparent that a current at the fundamental frequency flows through Leq. Cc is thus accorded to nullify the fundamental current downstream from the "fundamental current circuit".

Starting from the previous considerations and equations, the netlist (that is the circuit description file .CIR) can be obtained and simulated in Pspice. Indeed, Pspice impedes the simulation of the considered circuit "as is" when L_k is considered due to the presence of loop with only voltage sources and inductors. This obstacle can be easily overcome by adding a resistor, R_{loop} , in each loop (Figure 3:8). A small value of resistance has to be used since the resistor negatively affects the ability of the Pspice circuit in emulating the measured current.



Figure 3:8 Resistance Rloop not belonging to the proposed model but necessary to enable PSpice simulation.

It is worth to underline that the proposed model enables to emulate the steady-state current drawn by the lamp while it is not able in emulating the transient (inrush current and so on [93]) at the turn on. On the other hand, the waveforms in the Pspice circuit are obtained by using a transient analysis, then the simulation period has to be enough to ensure the steady-state condition is reached. When only capacitors are used for emulating the current harmonics, the voltage across C_k is imposed by the related loop (Figure 3:9):

$$v_{Ck}(t) = v_1(t) + v_{Fk}(t)$$
(73)

According to Kirchhoff's voltage law (KVL), the voltage across the capacitors is imposed, and then it is already at steady-state when the Pspice transient analysis starts.



Figure 3:9 KVL for obtaining the voltage across a capacitor adopted in the k-th harmonic circuit.

The use of these capacitors asks for inserting a compensation inductor, L_c , resonant at the fundamental frequency with the equivalent capacitor C_{EQ} . As said before a resistor has to be added in series with the inductor to avoid the loop $(L_c - v_1)$ which impedes simulation starts. Unfortunately, the current across this inductor is not yet at steady-state when the Pspice transient analysis starts and recalling the standard expression of the current in a resistor–inductor circuit:

$$i_{L_{c}}(t) = \left[i_{L_{c}}(0) - i_{L_{c}SS}(0)\right]e^{-\frac{R}{L_{c}}t} + i_{L_{c}SS}(t)$$
(74)

it is evident that the smaller the resistance the greater the time required to reach a steadystate condition. In other words, the use of small resistance provides more accurate results but at the cost of long simulation time. To solve this problem, the initial condition value has to nullify the transient current across the compensation inductor, L_c , that is:

$$i_{L_c}(0) = i_{L_cSS}(0) \implies i_{L_c}(t) = i_{L_cSS}(t)$$
 (75)

Given the expression of the current across the inductor at steady-state when $R \rightarrow 0$, that implies $v_{Lc}(t) \rightarrow v_1(t)$:

$$i_{L_{c}}(t) = i_{L_{c}SS}(t) \approx \frac{\sqrt{2}V_{1}}{\omega L_{c}} \cos\left(\omega t - \frac{\pi}{2}\right)$$
(76)

the initial condition to be set is easily obtained:

$$i_{L_c}(0) = 0$$
 (77)

To nullify the effect of R_{loop} (which is not present in the proposed model), a voltagecontrolled voltage source (VCVS) is placed in series with the resistor. More specifically, it is controlled by the voltage across the resistor with a gain equal to -1, thus obtaining that the following relation is valid for any resistance value Figure 3:10:



Figure 3:10 VCCS that nullifies the effect of R_{loop}, thus obtaining a Pspice circuit that actually implements the proposed model.

Therefore, setting the initial condition equal to zero enables a steady-state current through the inductor when the Pspice transient analysis starts, regardless of the value of R_{loop} . Moreover, nullifying the effect of the resistor enables to obtain a Pspice circuit that actually implements the proposed model. It is worth noting that the use of the VCVS without properly setting the initial condition enables to start the simulation but it does not address the underlining issues that lead Pspice to impede the simulation of the circuit where a loop with only voltage sources and inductors is present. Finally, it is worth recalling that the steadystate current across the inductor has the same amplitude of the fundamental current flowing through the equivalent capacitor but these currents are in the antiphase between them. Then the resulting fundamental current is zero which means, in turn, that the fundamental current flows only in the "fundamental current circuit".

As said before, whether b_{q1} is positive a capacitor is used in the "fundamental current circuit", then the voltage across it is equal to v_1 according to the related KVL, then the voltage is also already at steady-state when the Pspice transient analysis starts. On the other hand, when b_{q1} is negative an inductor has to be adopted and, consequently, a resistor R_{loop} with

(78)

the related VCVS has to be adopted and the initial condition has to be properly set (once again it has to be set equal to zero).

The mechanism adopted for these inductors can be readapted when L_k is used instead of C_k . At steady-state the voltage across the inductor is (Figure 3:9):

$$v_{Lk}(t) = v_1(t) + v_{Fk}(t) = \sqrt{2}V_1\left[\cos(\omega t) + \cos\left(k\omega t + \gamma_k + sign(b_k)\frac{\pi}{2}\right)\right]$$
(79)

Then the steady-state current is:

$$i_{LkSS}(t) = \sqrt{2}V_1 \left[\frac{\cos\left(\omega t - \frac{\pi}{2}\right)}{\omega L_k} + \frac{\cos\left(k\omega t + \gamma_k + sign(b_k)\frac{\pi}{2} - \frac{\pi}{2}\right)}{k\omega L_k} \right]$$
(80)

Consequently, the initial condition to set in order to obtain a zero transient time for the current through the inductor is:

$$i_{Lk}(0) = i_{LkSS}(0) = \frac{\sqrt{2}V_1}{k\omega L_k} \cos\left(\gamma_k + sign(b_k)\frac{\pi}{2} - \frac{\pi}{2}\right)$$
(81)

The switches in the model (Figure 3:5) have been emulated in Pspice with "Voltage-Controlled Switch" (VCS) components. On the other hand, the VCS is not an ideal switch as the ones considered in the model. More specifically, a VCS is a resistor with a high resistance, R_{OFF} , when it emulates the open-status while the closed-status is obtained by considering a low resistance, R_{ON} . The VCS resistance is set according to the voltage value of the controlling nodes. It is worth noting that, the switch SR1 and SG1 (Figure 3:5) have to be controlled by the same voltage but with opposite logic since when the former is open the latter is closed and vice versa. Therefore, to drive these switches, a fictitious constant voltage generator has been considered:

$$V_{bp1} = sign(b_{P1}) \tag{82}$$

and

$$R_{SR1} = \begin{cases} R_{ON} & when \, V_{bp1} = 1 \\ R_{OFF} & when \, V_{bp1} = -1 \end{cases}$$
(83)

$$R_{SG1} = \begin{cases} R_{ON} & \text{when } V_{bp1} = -1 \\ R_{OFF} & \text{when } V_{bp1} = 1 \end{cases}$$
(84)

A similar mechanism can be adopted for the couple of switches *SL1-SC1*, *SCc-SLc*, and *SLk-SCk*.

Regardless of the status of a switch, the related resistance does not belong to the proposed model. Hence, similarly to R_{loop} , their effect should be nullified. It is worth to note, that the switch should ideally be a short-circuit when it is closed (that is R_{ON} should be ideally zero). Therefore, the insertion of a VCVS that nullifies the effect of R_{ON} indeed enables to emulate a short circuit (as shown in Figure 3:11), thus, once again, the Pspice circuit actually implements the proposed model. On the other hand, R_{OFF} emulates an open-circuit and, consequently, the addition of the VCVS is undesired in such a case. Notwithstanding, as the VCVS nullifies the effect of R_{ON} , dually a current-controlled current source (CCCS) placed in parallel with R_{OFF} enables to nullify its effect, that is it enables to obtain the desired open circuit. More specifically, it is controlled by the current through the resistor with a gain equal to -1, thus obtaining that the circuit element downstream from the switch have not any effect on the current drawn from the lamp circuit model.



Figure 3:11 A VCCS and a CCCS used to emulate, respectively, a short circuit and an open circuit when L_k is selected (that is d=1, then L_k is connected in the lamp model while C_k is disconnected).

The parametric Pspice circuits of the generic model of an ESL are reported in Figure 3:12. It is worth noting that all voltage and current waveforms have been previously defined in terms of the "cosine" function while Pspice adopts the "sine" function; hence $\pi/2$ has been added in all the independent generators used in the Pspice netlist. The phase shift of the harmonics has been accordingly modified accounting for the related frequency.

Figure 3:12 is the main Pspice circuit of a generic ESL, where "freq" is the fundamental frequency, V1 is the rms of the voltage on the main assumed to be purely sinusoidal and d is the aforesaid parameter used for considering an inductor (d=1) or a capacitor (d=-1) in each harmonic subcircuit. The switches models "swpos" and "swneg" are used to emulate the complementary operated switches. They are used in combination with the fictitious independent generators Vbp1, Vbq1, and Vd to drive the couple of switches SR1-SG1, SL1-SC1 and SL-SC. Finally, Xp1, Xq1, and Xksim are the instances of, respectively, the active, reactive, and harmonics currents, while X1 is the instance of the compensation subcircuit. The parameters *ap1*, *bp1*, *aq1*, *bq1*, *ls*, *cs*, *gamma2*, *a2*, *b2*, and so on until *gamma50*, *a50*, *b50* can be reported in an external file for each lamp and then used with a ".INC" statement. The parameters *ls* and *cs* refer to the series in equations (72).

Figure 3:13 reports the netlist related to the active current, where the nodes 1 and 2 are the local numbering of the lamp terminals (which are node 1 and 0 of the Pspice lamp model as evident from the instance Xp1 in Figure 3:12); the nodes 3 and 4 are the local numbering of the terminals (which are node 2 and 0 of the Pspice lamp model) of the voltage controlling the switches status; "freq" is the fundamental frequency (50 Hz is the default value); *a* and *b* are a_{P1} and b_{P1} . Then, the actual nodes numbering as well as the parameter values depend on the calling instance Xp1. *SR1* and *SG1* are the switches complementary operated to ensure that only one between the resistor, *R1*, and the VCCS are connected. The following explanation refers to the local numbering.

When b_{p1} is a positive number the voltage across terminals 3-4 is 1 V, consequently *SR1* and *SG1* are equivalent to resistors with resistance equal to, respectively, $1\mu\Omega$ (that is *ron*) and 1G Ω (that is *roff*). Moreover, the expression in the curly brackets of the VCVS *ER1* is equal to -1, while is 0 for the CCCS *FR1* (it is an open circuit). Consequently, the voltage between nodes 1-6 is equal to 0, which implies a voltage across the resistor equal to the voltage on the mains (nodes 1-2) being also 0 the voltage between nodes 6-7. At the same time, the expression in the curly brackets of the VCCS *G1* is equal to 0.

The behavior of *SR1* and *SG1* is dual to the previous one when b_{q1} is a negative number because, in such a case, the voltage across the terminals 3-4 is -1 V. Moreover, the expression in the curly brackets of the CCCS *FR1* is equal to -1, while is 0 for the VCVS *ER1*. Consequently, no current flows through *R1* according to Kirchhoff's Current Law (KCL) at node 7. At the same time, the expression in the curly brackets of the VCCS *G1* is equal to the value of b_{p1} . Therefore, only *G1* is actually connected to the lamp circuit.

Similar considerations are valid for the other sub-circuits. When a switch is closed, the VCVS in series with it presents a gain -1 that ensures the switch is equivalent to a short circuit (in this case the CCCS has a gain equal to 0, then it has not any effect). The complementary switch is open, hence its VCVS presents a gain equal to 0, having no effect, while the CCCS has a gain equal to -1 to ensure the switch behaves like an open circuit.

| .param freq | 50 | | |
|-----------------------------|---|--|--|
| .param V1 | 212 | | |
| .param d | 1 | | |
| .param ron | lu | | |
| .param roff | lg | | |
| .model swpos | VSWITCH (RON={ron} ROFF={roff} VON= 1 VOFF=-1) | | |
| .model swneg | VSWITCH (RON={ron} ROFF={roff} VON=-1 VOFF=1) | | |
| .tran 0 40m | 1n 10u | | |
| .probe | | | |
| Vsim 10 | sin(0 {sqrt(2)*V1} {freq} 0 0 90) | | |
| Xp1 1020 | Ip1 PARAMS: freq={freq} a={ap1} b={bp1} | | |
| Xq1 1030 | Iq1 PARAMS: freq={freq} a={aq1} b={bq1} | | |
| X1 1040 | Comp PARAMS: freq={freq} lp={ls} cp={cs} d={d} | | |
| X2sim 1040 | Ik PARAMS: freq={freq} V1={V1} k={2} gammak={gamma2} ak={a2} bk={b2} d={d} | | |
| * other current harmonics | | | |
| X50sim 1 <mark>0 4 0</mark> | Ik PARAMS: freq={freq} V1={V1} k={50} gammak={gamma50} ak={a50} bk={b50} d={d} d={d} bk={b50} d={b50} d={b | | |
| Vbp1 20 | {sgn(bp1)} | | |
| Vbq1 30 | sgn(bq1) | | |
| Vd 40 | $\{d\}$ | | |

Figure 3:12 Netlist of the overall lamp circuit.

| .SUBCI | T Ip1 1 2 | 2 3 4 PARAMS: freq=50 a=0 b=0 |
|------------|-----------|--------------------------------------|
| Iap1 | 12 | sin(0 {sqrt(2)*a} {freq} 0 0 {0+90}) |
| SR1 | 1534 | swpos |
| ER1 | 5615 | $\{-(sgn(b)+1)/2\}$ |
| VR0 | 67 | 0 |
| FR1 | 17 | $VR0\{-(-sgn(b)+1)/2\}$ |
| R 1 | 7 2 | {1/abs(b)} |
| SG1 | 1834 | swneg |
| G1 | 8212 | ${b*(-sgn(b)+1)/2}$ |
| .ENDS | | |

Figure 3:13 Netlist of the active current circuit.

| .SUBCKT Iq1 1 2 3 4 PARAMS: freq=50 a=0 b=0 | | | | |
|---|------|---------------------------------------|--|--|
| Iaq1 | 12 | sin(0 {sqrt(2)*a} {freq} 0 0 {90+90}) | | |
| SL1 | 1534 | swneg | | |
| EL1 | 5615 | $\{-(-sgn(b)+1)/2\}$ | | |
| VL0 | 67 | 0 | | |
| FL1 | 17 | VL0 {-(sgn(b)+1)/2} | | |
| L1 | 72 | {1/(2*pi*freq*abs(b))} IC=0 | | |
| SC1 | 1834 | swpos | | |
| EC1 | 8918 | $\{-(sgn(b)+1)/2\}$ | | |
| VC0 | 9 10 | 0 | | |
| FC1 | 1 10 | VC0 {-(-sgn(b)+1)/2} | | |
| C1 | 10 2 | abs(b)/(2*pi*freq) | | |
| .ENDS | | | | |

Figure 3:14 Netlist of the reactive current circuit.

| .SUBC | KT Comp | 1 2 3 4 PARAMS: freq=50 lp=0 cp=0 d=0 |
|-------|---------|---------------------------------------|
| SL | 1534 | swneg |
| EL | 5615 | $\{-(-d+1)/2\}$ |
| VL0 | 67 | 0 |
| FL | 17 | VL0 {-(d+1)/2} |
| Lc | 72 | {1/(2*pi*freq*lp)} IC=0 |
| SC | 1834 | swpos |
| EC | 8918 | $\{-(d+1)/2\}$ |
| VC0 | 9 10 | 0 |
| FC | 1 10 | VC0 {-(-d+1)/2} |
| Cc | 10 2 | cp/(2*pi*freq) |
| .ENDS | | |

Figure 3:15 Netlist of the compensation circuit.

| .SUBCKT Ik 1 2 3 4 PARAMS: freq=50 V1=230 k=0 gammak=0 ak=0 bk=0 d=0 | | | | |
|---|----------|---|--|--|
| Iak | 12 | sin(0 {sqrt(2)*ak} {k*freq} 0 0 {gammak+90}) | | |
| VFk | 5 1 | $sin(0 {sqrt(2)*V1} {k*freq} 0 0 {gammak+sgn(bk)*d*90+90})$ | | |
| SL | 5634 | swpos | | |
| ERL | 6756 | $\{-(d+1)/2\}$ | | |
| VL0 | 78 | 0 | | |
| FRL | 58 | VL0 {-(-d+1)/2} | | |
| Lk | 8 2 | $\{1/(2*pi*k*freq*abs(bk))\}$ | | |
| $+IC = \{sqrt(2)*V1*abs(bk)*cos(\{(gammak+sgn(bk)*d*90-90)*pi/180\})\}$ | | | | |
| SC | 5934 | swneg | | |
| ERC | 9 10 5 9 | {-(-d+1)/2} | | |
| VC0 | 10 11 | 0 | | |
| FRC | 5 11 | VC0 {-(d+1)/2} | | |
| Ck | 11 2 | abs(bk)/(2*pi*k*freq) | | |
| .ENDS | | | | |

Figure 3:16 Netlist of the generic harmonic circuit.

3.3.2 Linear model validation

In the following, the comparison between the measured and simulated current waveforms of different LED light bulbs have been carried out. The results have highlighted the high fidelity level of the proposed linear model in emulating the current drawn by the LED light bulbs under variable voltage on the mains.

Figure 3:17 and Figure 3:18 report the measured voltage on the mains (blue waveform), the measured current (green waveform), and the simulated current (red waveform) for the two LED light bulbs when the *rms* voltage on the mains is, respectively, 212V and 237V. These two voltage levels have not been used for obtaining the parameters of the interpolation function, since these measurements have been performed only for model validation. They have been set by means of the variac similarly to the voltage level used for the interpolation functions as described before. The simulated currents that are reported in the figures have been obtained using the inductors in the harmonic subcircuits (i.e. d=1). The results confirm that the current is already at steady-state when the transient analysis starts, then the method adopted for the Pspice implementation of the current lamp model is effective. By analyzing the figures, it is also evident the goodness of the proposed Pspice model. It is worth noting that the predicted current is accurate for the LED light bulb in case of voltage on the mains exceeding the nominal value. As expected, there has been not any difference be-

tween the voltage THD during the current measurement (lamp on) and without any load (lamp off). The THD of the voltage on the mains has been measured in the interval 2%-4% during the measurements.



Figure 3:17 Measured voltage on the mains (blue waveform, *rms* 212V), measured current (green waveform), and simulated current (red waveform) for a LED. Per-unit system: base voltage 400V, base current 300mA.



Figure 3:18 Measured voltage on the mains (blue waveform, *rms* 237V), measured current (green waveform) and simulated current (red waveform) for a LED. Per-unit system: base voltage 400V, base current 300mA.
The total number of circuit components in the active and reactive current circuit is, respectively, 8 and 11 (Figure 3:13 and Figure 3:14). The compensation circuit includes 10 components (Figure 3:15), while there are 12 components in each harmonic subcircuit (Figure 3:16). Therefore, the total number of components of the lamp Pspice circuit is 617 when the current harmonics until the 50th are used. On the other hand, a reduced circuit can be obtained using a less general circuit where each switch with the related zero-voltage generator, VCVS, and CCCS are removed together with the elements to be disconnected. In such a case the number components in the active current circuit is 2. The number of components in the reactive current circuit is also 2 when the capacitor has to be considered (bq1 is positive), while is 4 when the inductor has to be considered since a resistor, R_{loop} and the related VCVS has to be considered. The adoption of only capacitors or only inductors in the harmonic subcircuit reduces its number of components to 3 in the first case, while 5 components in the second one due to the need of the resistor and VCVS. Finally, only the capacitor is adopted in the compensation circuit when the inductors are used in the harmonic subcircuits, while 3 components are necessary when an inductor has to compensate the equivalent capacitance of the harmonic subcircuits. Therefore, the smallest reduced lamp circuit has 154 components while 252 components are present in the greatest reduced circuit. It is worth noting that the reduced circuits provide the same current waveform of the one with 617 (full circuit) but each reduced circuit is related only to a given lamp.

Neglecting a current harmonic enables to remove 12 components from the full circuit (3 or 5 in the reduced ones) at cost of worsening the model accuracy. The error on the current THD described before has been used as a criterion for choosing the harmonic to be considered. Two levels of maximum THD error have been considered: 1% and 10%. It is worth noticing that each time a group of harmonics is neglected and the related components removed from the Pspice circuit the value of L_{EQ} (or C_{EQ}) changes, then the resonant (at the fundamental frequency) component in the compensation circuit have to be properly accorded. Figure 3:19 and Figure 3:20 report the measured current (green waveform) and the simulated current involving a THD error of 1% (red waveform) and 10% (blue waveform) for a LED light bulb at the two aforementioned voltage levels adopted for model validation. The harmonics neglected for achieving a given THD error refer to Figure 3:3 and Figure 3:4. It is worth noting that the reduced circuit performing a THD error of 1% contains less than 40 components.



Figure 3:19 LED light bulb: measured current (green waveform) and the simulated current involving a THD error of 1% (red waveform) and 10% (blue waveform) with a rms voltage of 212V.



Figure 3:20 LED light bulb: measured current (green waveform) and the simulated current involving a THD error of 1% (red waveform) and 10% (blue waveform) with a *rms* voltage of 237V.

3.4 Nonlinear model for emulating the current drawn from a LED light bulb

3.4.1 Description of the nonlinear model

The next stage of the research proposal has the aim to increase the effectiveness of the previous model. In this perspective, nonlinear interpolation functions have been taken into account. More specifically, bearing in mind the relations in (53), it has been considered a polynomial function of degree m for each component of the fundamental current as well as for each current harmonic. All the derived components have been expressed as a function of V_1 by interpolation of the measured rms currents. In detail:

$$I_{P1} = a_{P1} + b_{P1}V_1 + \sum_{m=2}^{np1} c_{P,m}V_1^{\ m}$$
(85)

$$I_{Q1} = a_{Q1} + b_{Q1}V_1 + \sum_{m=2}^{nq1} c_{Q,m}V_1^{\ m}$$
(86)

$$I_{k} = a_{k} + \sum_{m=2}^{nqk} c_{k,m} V_{1}^{m}$$
(87)

where n_{p1} , n_{q1} , and n_{qk} are numbers representing the maximum exponents of the polynomial expressions adopted for modeling, respectively, the active current, the fundamental nonactive current, and the *k*-th nonactive current.

The components of the proposed equivalent circuit model are obtained directly by the coefficient of the polynomial functions that, in turn, are obtained by the current measurements performed at the lamp terminals for different values of the voltage supply.

In the nonlinear model, both the active and nonactive fundamental currents are represented by means of three-term contributions:

$$i_{P1}(t) = i_{aP1}(t) + i_{bP1}(t) + i_{cP1}(t)$$

$$i_{Q1}(t) = i_{aQ1}(t) + i_{bQ1}(t) + i_{cQ1}(t)$$
(88)

(89)

In both equations, the magnitude of the first term is constant to emulate the constant coefficient in the related polynomial function, the second one represents the linear dependence on the voltage V_{I} , and the last one refers to the other terms of the interpolation function.

For the active current, the sum of the three contributions must present a phase offset equal to zero regardless of their amplitudes since i_{P1} is in phase with the network voltage. Considering that i_{aP1} has a constant amplitude and the amplitude of the other two currents changes as V_1 changes, the phase offset of each current must be equal to zero. At the most, some contributions could be in antiphase (i.e. phase offset equal to π) with i_{P1} , but the sum of all the contributions must present phase offset equal to zero. The contributions in phase with the active fundamental currents are associated with the positive coefficients in the related polynomial function, while the negative coefficients involve antiphase currents. In other words, a negative coefficient emulates an amplitude reduction that involves an antiphase current. Similar considerations are valid for the other polynomial functions related to the nonactive currents.

For the three current terms in (89), the phase offset has to be equal to $\pi/2$ (or $-\pi/2$), which is the phase offset of the nonactive fundamental current. Such an assumption is necessary to ensure that the phase offset of the sum of the three currents does not change when their amplitude changes.

Finally, the *k*-th current harmonic is represented as the sum of two different contributions:

$$i_{Qk}(t) = i_{ak}(t) + i_{ck}(t)$$
 (90)

The first term emulates the constant value of the related polynomial function while the second one represents the dependence on the voltage V_1 . Once again, the phase offset of these currents should be equal to φ_k although, at most, one could be in antiphase.

For the active current (see Figure 3:21), the constant and linear terms of the polynomial function are emulated by means of an independent current generator and a resistor o VCCS respectively. It is worth noting that the value of the magnitude of the independent current generator and the choice of the Resistor or the VCCS in the model circuit are both evaluated by using the (58) and (70) (See section 3.2).



Figure 3:21 Equivalent circuit for the active current. Only one element in the dashed box has to be used (it depends on the sign of b_{PI}).

The remaining terms in (91) are emulated in Figure 3:21 by means of a VCCS as represented in (92):

$$i_{cP1}(t) = \left\{ \sum_{m=2}^{np1} c_{P,m}[pV_1]^m \right\} v_1(t) = \left\{ \sqrt{2} \sum_{m=2}^{np1} c_{P,m}[pV_1]^m \right\} \sin(\omega t)$$
(91)

Where *p* is a parameter used to account for the variation of the network voltage amplitude.

Figure 3:22 depicts the components related to the nonactive fundamental current. The constant term in the polynomial function related to the amplitude of the nonactive fundamental current is emulated by means of an independent current generator to keep a constant amplitude. While the linear term in the polynomial function is emulated by means of an inductor or capacitor depending on the value of the coefficient b_{Q1} . Also in this case, the value of the independent current generator and the inductor or capacitor can be derived by using the relations (58)-(65)-(69) (See section 3.2).

The second and higher-order terms in the polynomial function in (89) are emulated by means of a current-controlled current source (CCCS) i_{cQ1} . It is worth noting that a CCCS enables to obtain current with variable amplitude while their phase offset is equal to the control current when the gain of the CCCS is a positive number, otherwise the current is in antiphase with the control one. Therefore, such a control current has to present a phase offset equal to i_{Q1} to obtain currents in phase or antiphase with that current. The independent current generator, i_{aQ1} , is chosen as control current because a waveform in phase with i_{Q1} is easily obtained by dividing the current of the generator:

$$i_{cQ1}(t) = \left\{ \frac{1}{a_{q1}} \sum_{m=2}^{nq1} c_{P,m} [pV_1]^m \right\} i_{aQ1}(t) = \left\{ \sqrt{2} \sum_{m=2}^{np1} c_{Q,m} [pV_1]^m \right\} \sin(\omega t + \varphi q)$$
(92)

The previous considerations and rules are also adopted for modeling the nonactive currents referred to the harmonics. The circuit model is depicted in Figure 3:23, where the k-th current harmonic, i_{qk} , is represented as the sum of two different contributions as reported in (90).

The first current in (92) is referred to the constant value of the related polynomial function while the second one represents the dependence on the voltage V_1 . It is worth mentioning that such a current also includes the linear term of the polynomial function (differently from the previous cases). On the other hand, the constant term of the polynomial function is modeled by means of an independent current generator (See section 3.2 and Equation (58)).

A CCCS is adopted to emulate the other terms in the polynomial function. The angular frequency of the control current waveform has to be $k\omega$, with a phase offset equal to φ_k . Therefore, i_{ck} is the control current that may be adopted (similarly to the previous cases):

$$i_{ck}(t) = \left\{ \frac{1}{a_k} \sum_{m=1}^{nqk} c_{P,m} [pV_N]^m \right\} i_{ak}(t) = \left\{ \sqrt{2} \sum_{m=1}^{nqk} c_{k,m} [pV_N]^m \right\} \sin(\omega t + \varphi_K)$$
(93)



Figure 3:22 Equivalent circuit for the fundamental nonactive current. Only one element in the dashed box has to be used (it depends on the sign of b_{QI}).



Figure 3:23 Equivalent circuit of the k-th nonactive current harmonic.

3.4.2 Nonlinear model validation

In the following, the comparison between the measured and simulated current waveforms of different LED light bulbs have been carried out. The results have highlighted the high fidelity level of the proposed nonlinear model in emulating the current drawn by the LED light bulbs under variable voltage on the mains. Furthermore, the PSpice netlist has been partially modified according to the aforementioned model.

A. Measurements

The quantities necessary for obtaining the components of the circuit model according to the rules and equations discussed before are obtained by using the coefficients of the polynomial functions that, in turn, have been obtained by the simple procedure described before.

The circuit model described in the previous section presents several components that are obtained through the acquisition of the active and nonactive power referred to the fundamental frequency and the amplitude and phase offset of 49 current harmonics, i.e. up to 2.5 kHz. The active and nonactive power and the current harmonics have been detected at the eight aforementioned voltage values from the measurements on the mains by a power analyzer. The experimental setup is the same as Figure 3:2, where the lamp in the box represents any commercial LED light bulbs. The experimental measurements and the model validation have been performed using several LED light bulbs by different manufacturers in the range 6W-12W. Each lamp type has been experienced by a five sample stock in order to verify the spread of the performances. The phase offset of each simulated current harmonic has been assumed equal to the one measured when the network voltage has been set at its

nominal value. In other words, the equivalent circuit is able to simulate only the variation of the harmonic amplitude, while the variation of both amplitude and phase offset of the fundamental current are modeled. Figure 3:24 summarizes the main steps to be executed in order to obtain the circuit model starting from the measurements performed at the terminals of LED light bulbs with the related data flow (represented by the dashed arrows). More specifically, the first box refers to the measurement procedure and the red storage silos represent the saved data. A limited number of measurements (six in the case study) is necessary to obtain the coefficients, thus this step requires low time. The second box refers to a simple modification of the data. The third box highlights the use of the data to evaluate the polynomial coefficients, which are saved in the blue silos. Finally, the last box reports the implementation (in a PSpice netlist) of the circuit model by using both previous data. Figure 3:24 highlights that the parameters extraction is performed off-line. Moreover, the computational effort to obtain the coefficients is very low because only the previous few values are necessary for each quantity to be modeled (e.g. the amplitude of the k-th current harmonic). According to these facts, the computational time for the extraction of the model parameters is very low.



Figure 3:24 Main steps to obtain the circuit model for any LED light bulb using only measurements at its terminal (i.e. black-box model).

Similar considerations hold true for the degree of the polynomial function. Indeed, for a given lamp, there is not an optimal value for all the current harmonics (i.e. the optimal value

of *np1*, *nq1*, and *nqk* is not the same). On the other hand, the value of each exponent in (85)-(87) depends on two conflicting targets: the improvement of the model accuracy of the related current asks for higher exponent values, while the reduction of the computational burden of the circuit model asks for lower exponent values. A second-order polynomial function has been always adopted (i.e. *np1*, *nq1* and *nqk* are set to 2 for each lamp) since the tests have revealed that it has been enough to obtain good results.

B. Simulation of the equivalent circuit

The data acquired from the instruments have been processed using an interface tool used to implement the circuit model and to create a PSpice netlist. As mentioned before, the circuit model integrates the polynomial functions of each current harmonic to emulate its amplitude variation according to the variability of the network voltage, but the model does not account for the phase offset variation. Therefore, for each harmonic, a fixed phase offset, equal to the one measured at V_1 , has been used in the netlist. On the other hand, the interpolation of both the active and nonactive fundamental current amplitude enables to account in the model for the phase offset variation of the fundamental frequency current. Finally, the netlist contains the parameter p to emulate the network voltage variation. Therefore, only one netlist file is obtained for each LED light bulb.

To evaluate the effectiveness of the proposed model, the amplitude of the fundamental (k=1) and harmonic (k=2...h) currents obtained by simulations have been compared with the measured ones. The comparison has been performed at eight voltage values (V_1 =0.85V₁ and so on) for each lamp under test and the percentage error between the current simulated, $I_{ksim}(V_1)$, and measured, $I_{kmeas}(V_1)$, has been computed:

$$E_{k}(V_{1}) = \left| \frac{I_{k,sim}(V_{1}) - I_{k,meas}(V_{1})}{I_{k,meas}(V_{1})} \right| \%$$
(94)

The maximum and minimum percentage error obtained at the various frequencies (fundamental and harmonics) is depicted in Figure 3:25(a).

It is worth noting that a considerable error occurs for many even harmonics and some odd harmonics at high frequency. However, this does not affect the effectiveness of the model since the amplitude of these harmonics is very small in comparison with the fundamental. To support such a consideration each error has been weighted by the normalized harmonic to evaluate the factual effect of each error in the suitability of the proposed model:

$$EN_{k}(V_{1}) = \left| \frac{I_{k,sim}(V_{1}) - I_{k,meas}(V_{1})}{I_{k,meas}(V_{1})} \right| \left| \frac{I_{k,meas}(V_{1})}{I_{1,meas}(V_{1})} \right| = \left| \frac{I_{k,sim}(V_{1}) - I_{k,meas}(V_{1})}{I_{1,meas}(V_{1})} \right|$$
(95)

Figure 3:25(b) shows the range of normalized error, *EN*, once again considering all the lamps. Figure 3:25(b) weights the error by the normalized amplitude of each harmonic to

highlight the low overall impact of the error on the current prediction ability of the model. The low impact of errors on the ability of the proposed circuit model in emulating the true current waveform (when using a second-order polynomial interpolation) has been also confirmed by visual inspection of the measured and simulated currents. In other words, a small estimation error occurs for the fundamental current and the relevant harmonics while a higher error occurs for the smallest ones (i.e. many even harmonics and some odd harmonics with k>22). Moreover, as it will be highlighted in the following, whether all the even harmonics and the odd harmonics at higher frequency (k>19) are neglected, the error on the power factor is less than 1% also when the lamp presents a low power factor. It is worth noting that, the voltage THD has been monitored during the acquisition of the currents to be modeled. The values of the voltage THD have been measured in the range 2.2%-4.7% during the measurements, which is typical THD meeting the limit of IEEE-Std-519. Therefore, the previous error has been computed comparing the measured currents (obtained with a distorted voltage within the limit) with the simulated ones obtained considering a pure sinusoidal voltage supply. In other words, a good accuracy of the model has been experienced when the voltage THD satisfies the limit.



Figure 3:25 Maximum and minimum percentage error: (a) before normalization - (b) after normalization.



Figure 3:26 Measured voltage (green) on the mains at V_1 =230V and currents (light blue) drawn by the LED 1 (a) and LED 2 (b). The waveforms are in per unit: 1p.u.= 400V for the voltage, 1p.u.=180 mA for LED1 current, 1p.u.=120 mA for LED2 current.

The main data and results for the LED light bulbs, named LED1, with greater error in terms of relation (32) are given in the following. LED 1 is a lamp with a low power factor (PF): 0.55. The data and results related to a high PF LED light bulbs, LED 2, have been also reported.

Figure 3:26 shows the waveforms measured for, respectively, the voltage on the mains and for the measured current drawn by the two LED light bulbs under the nominal condition in case of a voltage THD=2.6%. Figure 3:27 reports the amplitude of the fundamental current and the harmonics until the 50th for both lamps. More specifically, the amplitude of each current measured at the six voltage values adopted for obtaining the interpolation functions are reported. Moreover, the related amplitude obtained by using these interpolation functions are also reported to reveal the estimation error due to the interpolation. A magnification of higher-order harmonics has been also reported in order to better observe the measurements and the estimated values. Finally, Table 3:1 reports the coefficients of the poly-

nomial functions related to the active and nonactive fundamental currents and for the odd harmonics until the 45th. Figure 3:28 shows the comparison between the measured and simulated currents when two generic voltage values, different from the ones used to identify the model parameters, are used. The results confirm the effectiveness and robustness of the proposed model in emulating the lamp current drawn in case of variability of the network voltage for the two different LED light bulbs. Figure 3:28(a) and Figure 3:28 (b) report the comparison for LED1. In both cases, the current waveforms have been obtained considering until the 50th harmonic, thus obtaining a power factor (PF) equal to 0.5498 in good agreement with the one provided by the manufacturer. They are indicated as "original" waveforms in the figures. In the figure are also reported two further cases where a reduced number of harmonics are considered. More specifically, the dotted waveforms are obtained when the fundamental current and the 3rd, 5^{th,} and 7th harmonic are considered for the measured and simulated waveforms, thus obtaining a PF equal to 0.5864 that implies an estimation error less than 10%. Similarly, the dashed waveforms are obtained considering other current harmonics (odd harmonics until the 19th) so that the error on the PF estimation is less than 1% (estimated PF equal to 0.5547).

| coeff. | LED 1 | LED 2 | coeff. | LED 1 | LED 2 | coeff. | LED 1 | LED 2 |
|-------------------|-----------|-----------|------------------------|-----------|-----------|------------------------|-----------|-----------|
| a _{P1} | 8.93E-02 | 2.04E-01 | a _{13,0} | 1.99E-02 | 2.66E-03 | a ₂₇ | -2.68E-02 | 5.22E-04 |
| b _{P1} | -4.36E-04 | -1.18E-03 | C13,1 | -1.31E-04 | -1.39E-05 | C _{27,1} | 2.52E-04 | -3.08E-06 |
| Cp,2 | 6.94E-07 | 2.10E-06 | C13,2 | 2.93E-07 | 2.30E-08 | C _{27,2} | -5.42E-07 | 5.52E-09 |
| a _{Q1} | 7.11E-02 | 1.78E-02 | a 15 | -8.49E-03 | 9.58E-04 | a ₂₉ | -9.60E-03 | 3.31E-04 |
| bq1 | -4.95E-04 | -1.01E-04 | C15,1 | 1.28E-04 | -4.97E-06 | a ₃₁ | 1.76E-02 | 4.18E-04 |
| C _{Q,2} | 9.25E-07 | 1.81E-07 | C15,2 | -2.97E-07 | 8.22E-09 | C _{31,1} | -1.54E-04 | -2.49E-06 |
| a₃ | 8.20E-02 | 1.12E-01 | a ₁₇ | -3.24E-02 | 1.60E-03 | C31,2 | 3.63E-07 | 4.54E-09 |
| C _{3,1} | -4.00E-04 | -6.71E-04 | C17,1 | 3.26E-04 | -9.75E-06 | a ₃₃ | 2.25E-02 | 3.39E-04 |
| C _{3,2} | 6.47E-07 | 1.22E-06 | C17,2 | -7.08E-07 | 1.80E-08 | C33,1 | -1.92E-04 | -1.84E-06 |
| a₅ | 3.99E-02 | 2.57E-02 | a 19 | -3.25E-02 | 1.18E-03 | C33,2 | 4.27E-07 | 3.12E-09 |
| C _{5,1} | -1.16E-04 | -1.35E-04 | C19,1 | 3.09E-04 | -6.43E-06 | a ₃₅ | 2.60E-03 | 2.08E-04 |
| C _{5,2} | 1.31E-07 | 2.27E-07 | C19,2 | -6.48E-07 | 1.10E-08 | C _{35,1} | -1.11E-05 | -9.12E-07 |
| a ₇ | 1.51E-03 | 1.33E-02 | a ₂₁ | -1.29E-02 | 7.56E-04 | C _{35,2} | 1.80E-08 | 1.29E-09 |
| C _{7,1} | 1.23E-04 | -7.86E-05 | C _{21,1} | 1.33E-04 | -4.29E-06 | a ₃₇ | -1.59E-02 | 3.06E-04 |
| C7,2 | -2.78E-07 | 1.43E-07 | C _{21,2} | -2.70E-07 | 7.53E-09 | C37,1 | 1.49E-04 | -1.80E-06 |
| a9 | -1.24E-02 | 3.08E-03 | a ₂₃ | -4.27E-03 | 1.26E-03 | C _{37,2} | -3.30E-07 | 3.22E-09 |
| C9,1 | 1.64E-04 | -1.45E-05 | C _{23,1} | 6.40E-05 | -7.31E-06 | a ₃₉ | -1.10E-02 | 3.40E-04 |
| C _{9,2} | -2.89E-07 | 2.18E-08 | C _{23,2} | -1.46E-07 | 1.31E-08 | C39,1 | 1.01E-04 | -2.15E-06 |
| a ₁₁ | 3.62E-03 | 6.87E-03 | a ₂₅ | -1.91E-02 | 7.08E-04 | C39,2 | -2.13E-07 | 4.06E-09 |
| C _{11,1} | -6.00E-06 | -4.41E-05 | C _{25,1} | 1.95E-04 | -4.15E-06 | a ₄₁ | 3.45E-03 | 8.82E-05 |
| C _{11,2} | 8.15E-08 | 8.41E-08 | C _{25,2} | -4.39E-07 | 7.46E-09 | C41,1 | -2.94E-05 | -2.88E-07 |

Table 3:1 Coefficients of the polynomial functions for LED light bulbs with low (LED1) and high PF (LED2)

Similarly, Figure 3:28(c) and Figure 3:28(d) report the comparison for LED2. In such a case, it has been sufficient considering only the 3rd harmonic to obtain an estimation error of less than 10%. Moreover, to reach an error less than 1% on the estimation of PF the 3rd and 5th harmonics are enough. Such a strongly reduced number of harmonics to be considered is due to the higher PF with respect to LED1.

Figure 3:29 shows a screenshot of the PSpice netlist adopted to perform the simulation with the references to the equations provided in the model. The circuit model is parametric and for each LED light bulbs the coefficients are obtained by interpolation (Table 3:1), while the phase offset of each current harmonics measured at nominal voltage is adopted. Figure 3:30 reports a subpart of both lamps netlists which is included in the aforesaid main parametric circuit for obtaining the simulated waveforms. Therefore, the users can treat the lamp as a black-box where its behavior is emulated by the PSpice netlist which may be also modeled within a PSpice subcircuit (.SUBCKT). The subcircuits of different LED light bulbs can be used together with resistors and inductors that models the cable to obtain the current drawn from the overall lighting system (Figure 3:31). Such an approach is practicable because the PSpice subcircuit and, consequently, its underlying model is valid for typical THD voltage on the mains. This is due to the fact that the subcircuit output (i.e. the current drawn by the lamp) is not directly dependent on the other components, but it only depends on its input (the voltage at its terminals) and the other components affect the voltage without directly influencing its behavior.

According to the IEEE-Std-519, at the point of common coupling between a network operator and a user, the voltage distortion, in terms of THD, has to be limited to 8%. The value of the THD has been measured in the range of 2.2%-4.7% during the voltage and current measurements. As highlighted by the results (Figure 3:25, Figure 3:27, and Figure 3:28), the current waveform obtained by simulation neglecting the voltage distortion well approximates the current waveforms measured when the voltage on the mains has been distorted. Therefore, it is apparent that the model is enough robust at the use of a pure sinusoidal waveform for emulating the voltage supply at the simulation stage. In other words, although the model may fail when the voltage distortion exceeds the THD limits, it is plausible that it performs well until the limits are respected



Figure 3:27 Fundamental and harmonics currents obtained by measurements (left) and simulation (right) at six voltage values for LED1 (a) and LED2(b). The sub-figure at the left corner presents a magnification to better observe higher-order harmonics amplitude.





Figure 3:28 Total current drawn by the lamps. I_{mea} is the measured current (red trace) while I_{mod} is the simulated current (blue trace): (a) LED1 at V₁ =212V - (b) LED1 at V₁ =237V - (c) LED2 at V₁ =212V - (d) LED2 at V₁ =237V.

since the currents waveforms obtained by simulation considering a sinusoidal voltage on the mains well approximate the current measured in presence of a distorted voltage on the mains with a THD less than 8%.

It is worth noting that the normal situation is the one in which the THD limit is met since usually any countermeasures have been realized at the load/generation points along the distribution network to avoid exceeding the limit, for example by adopting filters in case of distorted load or generation units. In such a case, few lamps do not negatively affect the voltage THD due to the low current drawn. Therefore, the proposed model is suitable for estimating the current waveform of the whole (small) lighting system (Figure 1:9(a)). In the case of a large number of LED light bulbs, the model may fail at foreseeing the large current drawn from the whole light system when the current distortion level is so high that it lets the voltage THD at exceeding the limit (up case of Figure 1:9(b)). On the other hand, when the whole current is large but few distorted so that the voltage THD is left under the limit (down the case of Figure 1:9(b)), then using the model of each lamp is again sufficient to properly estimating the whole current. In this regard, the optimal

| GENERAL CIRCUIT MODEL OF LED LIGHT BULB SUBCKT Isi 1 2 PARAMS: freq=50 Vn=230 p=1 npl=2 alpha 10=0 alpha 10=0 alpha 1,2=0 : alpha 1,3=0 Isi_0 1 2 Isi_0 0 (apr(12)=sink=1,0) (freq) 0 0) : eq.16 Remember the line to be wonterrily eliminated, that is: come Isi 2 (alpha,1,1) I 1 2 (alpha,1,2=p*Vn) SUBCKT I 1 2 (alpha,1,2=p*Vn) SUBCK SUBCKT |
|---|
| <pre>SUBCKT Iq1 1 2 PARAMS: freq=50 Vn=230 p=1 phi=90 nql=2 beta_l_0=0 beta_l_1=0 beta_l_2=0 : beta_l_3=0 Iq_l_0 1 s sin(0 (sqrt(2)=beta_l_0) (freq) 0 0 (pi)) : eq.23 c consent the line to be noenterilly eliminated, that is consent Cl when beta_l_l=phi(0 othervise consent Ll and Rl1 =Cl 1 2 (beta_l_1/(2=pi=treq)): I 1 4 (-1/(2=pi=treq)): I (c (sqrt(2)=p=Vn=beta_l_1) : eq.27 Rl1 4 2 in.NOT part of the model. It breaks the loop. It effects the accuracy of the lamp current, then it has to be as small as possible. Fn 1 5 VOG ((L)=beta_l_0)=beta_l_2=(p=Vn)=*2) Rnn 5 2 1 (Rn is NOT part of the model. It breaks the loop. It has not any effect since it does not affect the current. ENDS</pre> |
| SUBCKT 1gk 1 2 PARAMES freq-50 %n+220 pr1 k-3 phi-0 ng-2 geama_k_2*0*0 geama_k_2*0 ; geama_k_3*0 1gk_0 1 3 sin(0 (sqrt(2)*geama_k_0) (k*treg) 0 0 (phi)) : eq.25 VG0 3 2 0 Fn 1 4 VG0 ((1/geama_k_0)*(geama_k_1*p*Wn+geama_k_2*(p*Wn)**2)); CCCS gain in eq.30+geama_k_3*(p*Wn)**3+ Rhn 4 2 1 ; Rhn is not part of the model. It breaks the loop. It has not any effect since it does not affect the current. ENDS |
| <pre>.INC LED_1.cir : insert here the lasp to be sizulated .paras freq 50 .paras Yn 220 .paras Yn 220 .paras Yn 220 .tran 0 20% 0 100 .tran 0 20% 0 100 .tran 0 20% 0 100 .tran 0 20% 0 100 .tran 1 0 size (2)*p*Vn) (freq) 0 0 0) Xa 1 0 Iai PARMMS: freq*(freq) Yn*(Yn) p*(p) pnl*(pnl) alpha_1_0*(a0) alpha_1_1*(a1) alpha_1_2*(a2) : alpha_1_3*(a2). Xa 1 0 Iai PARMMS: freq*(freq) Yn*(Yn) p*(p) phi*(phing) alpha_1_0*(a0) alpha_1_1*(a1) alpha_1_2*(a2) : beta_1_3*(a2). Xa 1 0 Iai PARMMS: freq*(freq) Yn*(Yn) p*(p) phi*(phing) alpha_1_0*(a0) beta_1_0*(b0) beta_1_2*(b1) beta_1_2*(b2) : beta_1_3*(a2). X2 1 0 Iak PARAMS: freq*(freq) Yn*(Yn) p*(p) k*(2) phi*(phin2) mq*(nq2) gaama_k_0*(g2_1) gaama_k_1*(g2_1) gaama_k_2*(g2_2). X3 1 0 Iak PARAMS: freq*(freq) Yn*(Yn) p*(p) k*(2) phi*(phin2) mq*(nq2) gaama_k_0*(g2_1) gaama_k_2*(g2_2).gaama_2*(g2_3)</pre> |
| X49 1 0 Iqk PARAMS: freq*(freq) Vn*(Vn) p*(p) k*(49) phi*(phi49) nq*(nq49) gaama_k_0*(g49_0) gaama_k_1*(g49_1) gaama_k_2*(g49_2):gaama_k_3*(g49_3) X50 1 0 Iqk PARAMS: freq*(freq) Vn*(Vn) p*(p) k*(50) phi*(phi50) nq*(nq50) gaama_k_0*(g50_0) gaama_k_1*(g50_1) gaama_k_2*(g50_2):gaama_k_3*(g50_3) |

Figure 3:29. Parametric PSpice netlist of the circuit model of LED light bulbs.



Figure 3:30. Coefficients (Table 3:1) and phase offset reported in the parametric PSpice netlist of LED1(a) and LED2 (b).



Figure 3:31. Circuit to simulate the current of the overall lighting system.

design of the large lighting system aims at reaching such a low-distorted overall current. Therefore, the proposed method is useful at the design stage because when a combination of LED light bulbs involves a:

- high-distorted current that involves voltage THD over the limit, a high-distorted current that does not well estimates the actual one is obtained by using the proposed model; but being the target of the optimization the combination with the lowest distorted current, such a combination leading a high-distorted current is however discarded regard-less the accuracy of the current.

 low-distorted current that does not involve voltage THD over the limit, a low-distorted current that well estimates the actual one is obtained by using the proposed model; and the estimation accuracy is necessary to effectively find the combination with the lowest distorted current. It is worth highlighting that, in the latter case, the model provides an accurate estimation of each lamp current and such a current may be high-distorted regardless the overall current is low-distorted. Indeed, all the currents may be high-distorted.

Chapter 4 The superiority of the unintentional LED for temperature detection.

4.1 Introduction

Nowadays, power electronics is widespread into a huge number of daily applications that improve services for the collective [94]. Furthermore, power electronic has a key role in renewable energy systems [95-96], lighting [97], electric mobility [98-99], and other systems that are enabling sustainable development [100].

A crucial aspect is the reliability and lifetime prediction of the whole power conversion system. The warrant of the highest robustness level while minimizing the product and maintenance cost is extremely mandatory. For example, the devices used in avionic and automotive applications must have a fault rate close to zero that imposes stringent requirements during the system design. In the same way, the wind farms must be guaranteed the normal operations without interruption, but it is extremely difficult due to the expensive access to the farm for easier maintenance. In this context, many approaches to forecast the lifetime of power electronic systems and the single power device have been intensively studied.

With the aim to guarantee a high level of reliability, it is important to comply with several standards [101-103], and furthermore, different strategies are usually performed such as the use of the fault-tolerant topologies with redundant components [104-105], and the advanced reliable design of power electronic devices using innovative materials [106-107].

Unfortunately, the enhanced system's robustness does not prevent failure and, indeed, it is never completely foreseeable. Therefore, a maintenance operation before a failure is necessary. Considering the costs of maintenance operations, lifetime prediction combined with condition monitoring approaches [108-109] are very useful tools to choose when a maintenance operation has to be carried out.

Several studies have shown that capacitors are fragile with a failure of 30% and the PCB and connectors failure are around 36% [110]. The remaining part is related to the semi-

conductor and soldering failures in device modules that consist of the most important area of concern for converter system failures. In this context, countless condition monitoring (CM) methods for the evaluation of the semiconductor state of health have been widely explored in the literature.

It is worth underlining that many papers have already studied and classified the CM methods based on the physical or electrical proprieties, where each method has a specific degree of intrusiveness with respect to the measurement to perform. It is worth noting that the intrusiveness level of each CM method has not been defined uniquely yet.

Hence, the definition of a unique figure of merit (FOM) able to represent the level of intrusiveness of the various CM methods is extremely interesting. In this perspective, in section 4.3 different criteria to establish the level of intrusiveness have been defined. They are related to the impact of a CM method in terms of the modification in the device, change in the converter operations, and system modification.

Then, three FOMs have been accordingly considered and assigned to all the CM methods by means of a mark based rule. Finally, a unique single-value FOM has been obtained from the product of the three FOMs. An application-based FOM that accounts for the importance of the different intrusiveness criteria in a given application has been also proposed. The use of a single value to compare the different CM methods is the main strength of these overall FOMs but they do not provide a fixed value due to the subjective of the marks. On the other hand, the rank of the CM methods according to an intrusiveness criterion is almost unbiased. Therefore, a strategy to compare the CM methods according to the rank instead of the mark is proposed by borrowing some reasoning from the Pareto optimality. Such an approach has revealed that the CM methods adopting, respectively, Integrated photodiode, IR Camera, Acoustic, and two thermo-sensitive electrical parameters (turn on-off delay time and peak gate current) are the best ones in terms of intrusiveness, and they are equivalent each other from the Pareto optimality point of view. Finally, the Integrated photodiode and the IR Camera CM methods are the less intrusive according to the overall FOM.

Among the various CM methods, the exploitation of the light emission phenomenon from the body diode of a SiC device is gaining more interest in the last decade. The electroluminescence phenomenon refers to the photoemission from a SiC material and it is related to its intrinsic body diode that operates in the third quadrant. The light brightness in the SiC intrinsic body diode depends on the junction temperature Tj and on the forward current that flows into the diode. The light intensity decrease at high Tj being equal to the current injected. Thus, the electroluminescence phenomenon can be a suitable indicator of the temperature of the SiC MOSFET chip during a general power converter application. Therefore, considering the results of the comparison among the CM methods performed in this chapter, which has highlighted that the use of the Integrated photodiode is one of the best choices, and considering the SiC MOSFET intrinsic diode as an unintentional diode, in Chapter 5 has been proposed a new CM method adopting the Integrated photodiode in a SiC MOSFET module.

4.2 Overview of condition monitoring methods of power devices

Nowadays, various CM methods are gaining more interest since the various maintenance strategies allow to increase the lifetime of the power device, and hence, the reliability of the overall system can be enhanced. In many applications, it is becoming crucial to monitor the state of the health (SOH) of the power devices to prevent a failure, that is the possibility for the operators to obtain a lifetime estimation, thus properly scheduling any maintenance operations.

| Parameter Dependence | | | |
|---|--|--|--|
| Acoustic waves | | | |
| Light wavelength | | | |
| Light intensity | | | |
| Light wavelength | | | |
| Resistance | | | |
| Resistance | | | |
| Voltage | | | |
| V _{ce} , V _{ds} | | | |
| Device current, V_{ce} , V_{ds} | | | |
| V _{ce} - V _{ds} | | | |
| V _{ge} - V _{gs} | | | |
| V_{ge} - V_{ce} , V_{gs} - V_{ds} , Gate resistance | | | |
| Device current, V_{ce} - V_{ds} | | | |
| Device current, V_{ce} - V_{ds} | | | |
| Vge-Vgs | | | |
| | | | |

Table 4:1 Parameters dependence of different CM methods

The thermal stresses like the increasing of the means temperature and abrupt temperature fluctuations are the main failure mechanisms. Consequentially, the temperature is an index of the power device SOH. Although the purpose of a continuous thermal control improves the reliability of the power applications, the level of intrusiveness should be also taken into account. Other CM methods focus on other quantities. Table 4:1 summarizes the physical or electrical quantities measured for each CM method, and, in the following subsections, a brief overview has been presented. It is worth noting that, among the various CM methods in the literature, the acoustic one is used to detect the state of the aging of the power device without any estimation of the working temperature, while all the other strategies have been focused on the estimation of the junction temperature (JT). More specifically, some CM methods perform such estimation by directly measuring the temperature, other methods provide an indirect estimation by mapping electrical quantities in a temperature value.

4.2.1 Acoustic methods

Acoustic emission has been widely investigated in literature as a CM quantity useful in different applications fields such as pumps, industrial electrical machines, and so on. Moreover, in the field of power electronics, acoustic monitoring has been extensively used to detect any defects or damage in transformers and capacitors [111-113]. Only in the last decade, a few works have been focused on the acoustic phenomenon such as a measurement method for monitoring the SOH of the power semiconductor devices [114-119].

The acoustic emission has been used with the aim to detect any physical damage in a power module packaging by using an acoustic microscope. Furthermore, from the experimental evidence, it has been proved that acoustic emissions are related to the switching operations of the power devices. In the case of fast switching operation (tens of nanoseconds), a certain amount of current is switched and it causes large *di/dt* that involves magnetic interaction within the module packaging. This means that the magnetic force could be the source of the acoustic emission, such as the mechanical breaking of the structure inside the component package. However, the physical phenomena causing the acoustic emission are not definitively understood.

Thereby, the device under test (DUT) is monitored contactless with an acoustic sensor that is usually placed in the proximity of the package. It intrinsically eliminates the issues related to contact directly with the voltage probes.

A correlation between the SOH of the power module and the analysis of its acoustic emission during the switching process has been analyzed [114-116]. It has been demonstrated that the acoustic peak in an aged device is smaller in comparison to a new one. However, in these works, only the acoustic emission of an IGBT connected into short-circuit has been investigated. While [117] presents an experimental setup to detect the

acoustic emission during the normal operation of the converter. In [118] an acoustic emission variation of a power module, when it is subjected to solder delamination, has been measured. Furthermore, the authors in [119] have proved that this method can be used for non-invasively state of the aging of a power semiconductor module due to power cycling during a real application. The aging process has been observed with a simple acoustic sensor during the switching operation by using a resistive-inductive load. The experimental results have also proved a strong correlation between the acoustic emission and the on-state voltage drop that in turn, is used to understand the bond wires degradation of the module. More specifically, the on-state voltage drops increases as the bond wires degrade.

As for disadvantages, the acoustic method needs an expensive and complex sensing circuit to decode correctly the acoustic emission. Furthermore, the system has to be shielded against EMI and superposition of noise contributions.

4.2.2 Optical Methods

Temperature variation, especially the sudden increase of the JT, plays a significant role in terms of power device reliability [120-121]. CM methods performing on-line JT monitoring raise great interest in terms of planning maintenance operations because the working conditions of a power converter are extremely unpredictable. In this perspective, the CM methods based on the optical properties of the semiconductor power devices have been studied in depth because they are useful for temperature estimation. These methods are usually based on an optical beam that is reflected or scattered back from the semiconductor lattice. There is an inherent dependence between temperature and the energy-related to the photoemission. More specifically, such energy is a function of the JT, hence, in turn, the energy variation can be used to estimate the temperature of the chip. There are various techniques for a thermal mapping that are based on the use of an IR sensor [122-125], IR microscope [126], the 2-D radiometry [127-128] the laser deflection technique [129-131]. While the optical fibers [132-134] and the IR camera [135-139] can be used both to obtain a thermal mapping or the JT value.

Among the aforementioned techniques, the latter two are the most widespread because of their high accuracy and fast response time. As an example, in [134] an optical fiber sensor has been used to measure the die temperature of an IGBT power module in order to estimate the thermal impedance. The module top lid has been removed because the optical system has to be placed close to the die, while it has not been necessary to remove the dielectric gel. This kind of technique allows exclusively a local temperature measurement. Printing the die and bond-wires to increase the emissivity of the chip is usually preferable, but causing a cost increment and severely limiting the on-line use of the method.



Figure 4:1 Thermal map of a SiC power module in case of current injection in the body diode.

The use of an IR camera [139] allows displaying the thermal map of the whole surface of the power module under test, as shown in Figure 4:1. As expected, the module temperature is not uniformly distributed: the temperature gradient between the center and the edge of the module can be greater than 40 °C. Usually, an IR camera is used to carry out a spatial thermal mapping on the device surface, but it is not able to provide an accurate measure of the device JT. It is worth remembering that some temperature measurement errors can be done using an IR camera due to the surface degradation of materials and the intrinsic low emissivity of aluminum. Even in this case, the IR temperature measure solution that increases the thermal emissivity on the surface. These solutions based on optical quantities have some other drawbacks such as the high cost and the impracticality in high-voltage converters.

The SiC power MOSFETs are more and more diffusing devices since they concurrently enable high switching frequency, high voltage, and high-temperature operations. Therefore, the study of the electroluminescence proprieties of the SiC material for on-line CM has started to attract wide interest. For different reasons, the electroluminescence proprieties have already been studied back in 1907 [140], while, in the last three years, the inadvertent light emission phenomenon in the intrinsic body diode has been drawn attention. While the body diode is in forward conduction mode, the chip glows a visible blue light [141]. The light brightness of the SiC body diode strongly depends both on the injected current magnitude and on the JT. Hence, the measurement of light brightness can be used as a novel CM method for temperature detection, where only a few works have already focused on this topic [142-144]. The first proof of concept of the SiC light emission in a commercial power module has already proved the potentiality of this CM method [142]. An inexpensive passive sensing circuit, such as a silicon photodiode and a resistor, has been adopted and the photodiode output voltage has been correlated to the light emission intensity as a function of the temperature. It is worth noting that the system is small enough that can be easily embedded in the package. Another approach considers a light circuit sensing by using two commercial photodiodes with an active signal conditioning circuit [143]. This approach has been adopted for JT estimation in a real application such as a PWM driven converter. The temperature-dependent changes in the spectrum of the light emission from the body diode of a SiC module have been also investigated [144]. The method has been proved by means of static characterization and dynamic double pulse measurement by using two Silicon Photomultipliers, that are able to detect the peak intensity and consequentially the dependence on temperature. Differently from the previous CM methods, the last one which is based on the light intensity of the SiC body diode enables high-voltage operations, and even more important, the JT can be estimated during on-line operations.

4.2.3 Physical Methods

The early works treating CM methods for the JT measurement have been done by directly contacting the chip surface with a thermo-sensitive material such as a point contact system. In this case, the direct access to the semiconductor chip is necessary and, consequentially, the package must be removed.

Various equipment has been used for the physical contact measurement: thermocouples, thermistors, scanning thermal probes, and multiple contact or blanket coatings, such as liquid crystals and thermographic phosphors [145-151]. The aforementioned equipment relies on the transfer of thermal energy from the DUT to the thermal sensors. In this case, the spatial resolution related to the contact measurements strictly depends on the size and the thermal capacitance of thermo-sensible materials. The ability to provide a temperature map by means of a matrix of thermocouple and a wide spatial resolution (can reach less than 100nm) are the main advantages. Notwithstanding, the physical contact methods are not over widespread in practical applications since the chip of the power module must be accessible to the thermal probe and in this perspective, the on-line measurements and high voltage operations are strongly limited. Furthermore, the measurement of the thermal variation of the power module strictly depends on the time response of the probe that may be considerably slower (few seconds) than the variation of the module JT.

The preliminary methods for the measurement of the JT in a power device have been obtained by using the Thermochromics Liquid Crystals [145] which consist of a thermal imaging tool for mapping surface and spatial temperature distributions. Nowadays, only in a few cases [146-147], the JT of an IGBT module has been experimentally measured during online converter operation. More specifically, the temperature has been determined by using several thermocouples physically connected to the chip. On the other hand, several works use the measurement of some thermocouples as target values used to prove the accuracy and effectiveness of new online junction temperature estimation models [148-151]. For example, the effectiveness of a model carried out, for a three-phase power module IGBT, by considering the transient thermal impedance has been

proved by using several thermocouples [148]. Experimental setup and an on-line control system that includes a microcontroller and a matrix of K-type thermocouples have been built up in order to verify a numerical thermal model for IGBT devices [149]. In [150] an electrical-thermal model has been carried out in terms of both the transient and steady-state responses. To validate the model, an array of thermocouple has been installed on the chip surface. A thermal model based on the Fourier series solution of heat conduction equations has been also validated by using several thermocouples placed both on the surface of the silicon die, on the base plate, and on the heat sink, in order to characterize the transient electrothermal behavior of an IGBT module [151].

4.2.4 Electrical Methods

Electrical methods for the JT measurement are often the preferred choice for the CM of power device because the temperature estimation can be carried out through the measurement of electrical quantities. More specifically, it is worth remembering that the proprieties of the semiconductor materials are temperature dependent and hence, the measurement of the voltage drop or the current that flows into the device can be used as a valid temperature estimator. Among various electrical CM methods, the Thermal Test Chips (TTCs) are directly fabricated on the die surface of the device, and the voltage drop can be used to estimate the temperature variations. On the other hand, the TSEPs are based on the measurement of the voltage drop (or current) during the converter operations. Generally, the measurement of the voltage drop can be carried out by using some voltage probes that are connected to the device terminals. The TSEPs are usually the preferred choice for CM because of their user-friendliness, fast response time to the temperature transients, and good accuracy.

4.2.4.1 Thermal Test Chips

TTCs have been originally developed for the thermal characterization of devices packages [152], and then they have also been used in IGBT power modules. TTCs act as thermal sensors to monitor the JT and they are fabricated on the proximity of the silicon chip.

TTCs can be suitable for online temperature measurements. Various types of TTCs have been realized, such as integrated diodes and resistance temperature detectors (RTDs) [153-161]. Since the forward voltage of the diodes strongly depends on the temperature variation, the measure of the voltage drop can be used for temperature estimation. It is worth remembering that the temperature presents an exponential dependence on the forward voltage. Likewise, RTDs are also used as the temperature-sensitive parameter because the voltage drop is related to the resistance variations. The variable resistance, R_{t_r} can be expressed as:

$$R_t = R_0 (1 + \alpha_0 \Delta T) \tag{96}$$

where R_0 is the value of the resistance at 0 °C, α_0 is the resistance temperature coefficient that strictly depends on the material, and ΔT is the temperature variation.

To use TTCs, a modified IGBT power module layout with an accessible on-chip temperature terminal has been proposed [153]. A string of diodes on the top of the chip has been fabricated and the measurement of the JT has been performed by measuring the forward voltage drop. Instead, in [154] a thin-film RTD placed on the top of the IGBT chip has been realized to measure the average temperature of the die. A similar solution where an NTC thermistor has been embedded in the IGBT power module has been also investigated [155]. Innovative use of a kelvin-emitter resistor, placed directly on the IGBT die surface, as a junction temperature sensor has been also adopted [156]. It provides only a local temperature measurement. While in [157] a chain of integrated diodes has been fabricated on the die surface with the aim to investigate the JT variations during a power cycling test.

The widespread of SiC power module in different power electronics applications has also driven forward the research of innovative control techniques that require real-time monitoring or estimation of the module's JT. In this perspective, several works [158-161] have been focused on the development of electrical models of the devices in which several NTC thermistors have been integrated on the die surface. The measurement of the temperature variation enables the estimation of the aging in a device and consequentially the device model can be continuously updated.

The main drawbacks of the TTCs are the production cost and the manufacturing complexity of the embedded sensors. Indeed, such layout modification complexity of the power module packaging can considerably increase and it also requires additional terminals for the temperature measurements. Furthermore, it is worth remembering that the diodes and RTDs can be affected by degradations along the lifetime of the device that may affect the accuracy of the measurement. These issues have limited the TTCs spread in commercial power devices.

4.2.4.2 Methods using the thermo-sensitive electrical parameters (TSEPs)

The CM methods outlined so far require visual or physical access to the chip. To overcome this limitation, the temperature measurement by thermo-sensitive electrical parameters (TSEPs) has been used as a valid alternative for the estimation of the JT of a power device. The key point consists of correlating the temperature of the semiconductor material with the electrical quantities during the normal operation of the converter. More specifically, the semiconductor devices have an intrinsic dependence on the temperature-related on different parameters, such as the mobility of the carriers $\mu(T)$, intrinsic concentration $n_i(T)$, and the bandgap energy $E_a(T)$. It is worth remembering that the $E_g(T)$ and $n_i(T)$ increase at higher temperatures. While the $\mu(T)$ has a complex dependence with the temperature that is related to the doping concentration and traps in the gate oxide and silicon interface. Therefore, the temperature dependence on the aforementioned parameters may be written as [162]:

$$E_g(T) = E_g(T_0) - \alpha_1 \frac{T^2}{T + \beta_1}$$
(97)

$$n_i(T) = N^{\alpha_2} e^{-\gamma/T}$$
(98)

$$\mu(T) = \mu_0 \frac{\beta_2 \left(\frac{T}{T_0}\right)^{\alpha_3}}{1 + \beta_2 \left(\frac{T}{T_0}\right)^{\alpha_4}}$$
(99)

where α_1 , α_2 , α_3 , β_1 , β_2 , and γ are empirical coefficients, *N* is the number per unit volume of effectively available levels states and T_0 is the room temperature.

Consequentially, the measurement of the electrical quantities measured at the device terminal can be used as a temperature estimator.

Therefore, TSEPs methods use passive voltage or current probes that measure the electrical quantities at the device electrodes, without direct access to the chip device, then the JT is estimated from these measurements. Furthermore, the TSEPs are the preferred approaches to obtain easily JT measurements on packaged devices with a fast time response (less than 100 microseconds). On the other hand, the TSEPs methods do not provide a thermal map of the DUT and hence the JT peak is often hard to evaluate [164]. Such an issue is more severe in multichip devices where the voltage or current measurements only provide a rough temperature of the whole device, without the possibility to know the effective temperature distribution among several paralleled chips [165]. In the following subsection, the main TSEPs methods are briefly discussed.

On state voltage measurement

Among the different TSEPs methods, the on-state voltage measurement under low current injection has been the most used in many industrial and academic applications. In this case, the TSEP is the voltage drop across the device. The advantage of using this CM method relies on the easy calibration procedure and the negligible self-heating of the DUT.

This CM method is widely employed when the devices have a PN junction in their structure. More specifically, bearing in mind a MOSFET power device, the temperature variation can be evaluated as the on-resistance $R_{ds,on}$ fluctuations during the converter operations. For the sake of simplicity, the $R_{ds,on}$ can be approximated as:

$$R_{ds,on} \approx R_{ch} + R_d + R_{sub} + R_{cs} + R_{cd} + R_s + R_d + R_{jfet}$$
(100)

where R_{ch} is the channel resistance, R_d drift region resistance, R_{sub} is the substrate resistance, R_{cs} and R_{ds} are the source and drain contact resistances, R_s and R_d are the source and drain resistances and R_{ifet} is the JFET resistance. Furthermore, the R_{ch} and R_d can be evaluated as [163]:

$$R_{ch} = \frac{L_{ch}}{W_{ch}\mu_{ch}C_{ox}(V_{gs} - V_{th})}$$
(101)

$$R_d = \frac{L_d}{q\mu_d N_d A_d} \tag{102}$$

where L_{ch} and W_{ch} are, respectively, the channel length and width; C_{ox} is the gate capacitance; L_d and A_d are the drift region length and area; N_d is the doping concentration of the drift region; μ_{ch} and μ_d are the channel and drift region mobility. It is worth noting that the R_{ch} decreases at higher temperatures because of both the μ_{ch} and V_{th} decrease at higher temperatures. On the other hand, the R_d acts as a positive temperature coefficient thermistor due to the temperature dependence of μ_d that decreases at higher temperatures.

It is worth to underline that the temperature coefficient of the $R_{ds,on}$ may differ for the power devices from different vendors, that is mainly caused by the different design of the device. Therefore, notwithstanding an easier calibration procedure, the CM method must be calibrated when a different device is adopted.

Firstly, the calibration procedure is mandatory, which is used to find the relationship between the JT and the TSEP. Typically, the calibration step consists in the use of a current source *I*_{*cal*}, in a range from 1 mA to tens of A. It is worth noting that during the calibration procedure, the device temperature can be usually fixed by a temperature-controlled heat sink. Then, the temperature measurement can be carried out during the dissipation stage, where the TSEP is measured in a typical converter application. In this case, a current source, *Id*, feeds the DUT in order to increase its temperature by means of power dissipations. Therefore, the voltage drop across the device, under known electrical conditions, is measured as a function of the temperature.

A simplified schematic of the circuits for the measurement of the voltage under low current is depicted in Figure 4:2 for an IGBT (Figure 4:2(a)) and a MOSFET (Figure 4:2(b)). The measurement can be carried out both for the on-state and-off state voltage. A voltmeter is usually connected in parallel to the DUT for the measurement of the voltage drop. It is worth noting that, the current *Ical* must be at least hundreds of mA to guaran-

tee a linear relationship between the voltage drop and the temperature [166-167]. In literature, many works [166-181] have focused on voltage measurement under low current injection in power diodes during forward polarization [170-173], in IGBT power modules [176-181], and also in power BJTs. Some works [180-181] have focused on the JT estimation in an IGBT power module whose on-state voltage (i.e. collector-emitter voltage, V_{CEon}) has a negative temperature coefficient. The main drawback of this method is the high dependence on the collector current during the measurement of the $V_{CE,on}$. Hence, the load current should be diverted during the measurement and this momentary interruption limits the use of this method in real-time applications.



Figure 4:2 Electrical circuits for the static measurement of the voltage under a low current: (a) IGBT (b) MOSFET.

The principle of operation of the CM methods based on the on-state voltage measurement at high current injection is almost similar to the low current injection methods. The measurement of the $V_{CE,on}$ (or $V_{ds,on}$) voltage drop across the device is used as a TSEP as described for the low current injection mode. The main difference with respect to the previous CM method relies on the calibration procedure. More specifically, a higher current is used for the calibration procedure and it produces a non-negligible selfheating. In this perspective, the relation between the voltage drop on the DUT and the temperature depends also on the value of the injected current.

The experimental setup for the temperature measurement is depicted in Figure 4:3. A high current generator feeds the DUT with a pulsed current, I_{H} , and a voltmeter is connected in parallel to the DUT. It is important to point out that the measurement of the JT can be obtained during the heating process.



Figure 4:3. Electrical circuits of a DUT for the static measurement of the voltage under high current.

The TSEPs are usually the MOSFET on-state drain-source voltage [182], the power diodes forward voltage [176], and the IGBT on-state emitter-collector voltage [176]-[183]. The sensitivity of the aforementioned TSEP is strictly related to the on-state current value regardless of the specific device. The JT estimation is only practicable for current values greater than tens of Ampere [183]. Hence, this method appears to be very useful especially for online JT measurement during the normal converter operation. Several circuit solutions to measure the V_{CEOP} of the power device have been devised [166],[176],[182-185].

This approach also presents some limitations due to the voltage swing between the onstate and off-state of the device. This implies the use of advanced electronic sensing circuits, thus increasing the complexity of the system. Innovative and compact sensing circuit to face these issues have been proposed [182,183].

Another issue is the contact resistances of the voltage probes that cause an undesired voltage drop that may produce an overestimation of the JT measurement [184]. This issue has been partially mitigated with a correction factor based on the layout of the power module [184-185]. Unfortunately, the introduced correction factor has to be calibrated as the device aging progress.

Saturation current

The measurement of the saturation current, I_{sat} , has been also used as TSEP [173],[179],[186-187] in power module with IGBTs or MOSFETs. This current can be measured by using a current probe or a voltage probe (by adding a shunt resistor). The electrical quantities measured provide a JT estimation due to the dependence on the chip temperature of the channel electron mobility, μ_{ch} , of the threshold, V_{th} , and of the PNP transistor current gain β for the IGBT [186]. It is worth remembering that the current I_{sat} shows a complex temperature dependence, but under the assumption that all the devices are at the same temperature and by neglecting the self-heating, the current I_{sat} in a device can be simply approximated as:

$$I_{sat} = \frac{1}{2} \frac{\mu_{ch}(T) W_{ch} C_{ox}}{L_{ch}} \left(V_{GS} - V_{th}(T) \right)^2$$
(103)

The measurement setup consists of a voltage source, V_{GT} , connected between the gateemitter (or gate-source) terminals of the DUT and a DC source voltage, V_D , connected between the drain-source or collector-emitter terminals of the DUT. Figure 4:4 shows the setup for an IGBT device. The voltage value of V_{GT} is usually higher than the threshold voltage V_{th} of the device and a pulsed current is injected into the DUT by controlling the switch T_1 . The saturation current can be measured through the voltage drop on the R_{shunt} . The setup clearly demonstrates that the thermal characterization of the device cannot be performed during the online converter operation.



Figure 4:4 Electrical circuits for the measurement of the saturation current in an IGBT device.

The first procedure is the calibration step, where the DUT is usually placed in a controlled hot plate that overheats the device, and hence, the I_{sat} is measured at varying plate temperature. Then, the measurement procedure (see Figure 4:4) consists to perform a non-destructive short-circuit in order to produce a significant channel temperature variation during a short time. In this perspective, the measurement variation of the current I_{sat} can be associated with a specific temperature value. Moreover, the temperature calibration may not be performed without power losses that influence the device self-heating [186-187]. Furthermore, it has been demonstrated that JT measurement is more accurate only for high temperatures.
Gate Threshold Voltage

The threshold voltage V_{th} is defined as the voltage to be applied to the gate-source terminals to have a given current, which is the minimum current that must flow into the device channel to assume the device turned on. Instead, from the standpoint of power electronic devices, the V_{th} is defined as the level of gate bias needed to observe a transition from weak-inversion to strong-inversion. For a MOS transistor structure, the V_{th} can be approximated as [188] :

$$V_{th} \approx 2\varphi_F(T) - \frac{Q_{SS}}{C_O} + \varphi_{ms}(T) + \sqrt{\frac{2\varepsilon q N_A}{C_O}} \sqrt{2\varphi_F(T)}$$
(104)

where φ_F is the Fermi potential Q_{SS} is the extrinsic change due to surface states, C_O is the gate oxide capacitance, φ_{ms} is the metal-semiconductor work function difference, ε is the oxide dielectric constant, q is the elementary charge unit and N_A is the body doping.

By referring to the (9), it can be demonstrated that the voltage V_{TH} decreases with increasing temperature [188] and it is a TSEP useful for temperature monitoring of MOSFETs [175], [189], and IGBTs [179-180], [190-191]. A potential measurement setup for the calibration procedure and the measurement of the V_{th} as TSEP in the case of an IGBT device is depicted in Figure 4:5.



Figure 4:5 Circuit for the calibration step of the threshold voltage method.

The gate and drain (collector) terminals are short-circuited and a current source, I_{cal} , feeds the DUT while a voltmeter measures the V_{th} . It is worth noting that the calibration step is based on the low current injection method and whereby, the self-heating is negligible.

Some works [179-180] have focused on the temperature dependence of the V_{th} measured by varying the collector-emitter voltage and the current collector value for an IGBT device. The I_{cal} value has to be higher than 5 mA in order to have a correct calibration step for high temperatures and high sensitivity [179-180]. Other works [190-191] have

focused on the temperature measurements after the power dissipation of the device. More specifically, a current source with two different current levels, one for dissipation (high current injection) and the other for JT measurement (low current injection) has been proposed. This CM method is not suitable for online condition monitoring.

Gate–Source or Gate-Emitter Voltage turn ON-OFF

The gate-emitter (source) voltage, V_{ge} (or V_{gs}) is used as a TSEP during the turn-on and turn-off of the switch [194-195]. The high sensitivity and the linear dependence of V_{ge} (or V_{gs}) with the temperature is the strength of this method. Similarly to the threshold voltage method, the V_{ge} (or V_{gs}) TSEP method cannot be used for online JT estimation in a power converter application, because the gate and collector (drain) terminal has to be shorted. The experimental setup of the gate-source or gate-emitter voltage as CM method is very similar to the threshold voltage one (see Figure 4:5). In this case the current injected into the DUT is higher than the current used in the threshold voltage method and consequentially the self-heating is not negligible.

Figure 4:6 depicts the simulation of the gate-emitter voltage V_{ge} of an IGBT during the turn-off at varying the device temperature. It is worth noting that the following analysis can be done by considering the turn-on of a device. The Miller plateau becomes wider as the temperature increases. In other terms, the time shift Δt in the figure is strictly related to the temperature of the chip and it can be detected by using a time counter that triggers from the first falling edge to the second one after the Miller plateau.

The Miller plateau width t_d can be approximated as [194]:

$$t_d = \frac{R_{Gint}(T) \cdot C_{rss}(T) \cdot (V_{DD} - V_{ON})}{\left(\frac{I_{load}(T)}{g_m(T)} + V_{th}(T)\right)}$$
(105)

where R_{Gint} is the internal gate resistance, C_{rss} is the Miller capacitance, V_{DD} is the DC-link voltage, V_{ON} is the on-state voltage, I_{load} is the load current, g_m is the transconductance. Equation (10) shows that t_d is directly proportional to C_{rss} and R_{Gint} . It is worth noting that the impact of temperature variation on V_{ON} and V_{DD} is negligible, while the temperature variation of the terms (I_{load} / g_m) and V_{TH} partly neutralize each other. The internal gate resistance depends on the temperature as the electron mobility decreases at higher temperatures. Therefore, t_d increases at higher temperatures due to the temperature dependence of C_{rss} and R_{Gint} . Therefore, the time interval t_d of the Miller plateau in the V_{ge} (or V_{gs}) voltage can be used as a TSEP to estimate the JT of IGBTs (or MOSFETs).



Figure 4:6 Simulated V_{ge} waveform during the turn off of an IGBT by varying the temperature working operations.

The authors in [194] have proved the temperature independence of the collectoremitter voltage. Instead, the calibration step measurement has been improved in [138] where an auxiliary sensing circuit has been added to the gate driver with the aim to reduce undesirable oscillations during the turn off of the device. While in [195] the linear dependence of t_d respect the temperature of the chip has been demonstrated and also, a parametric analysis by varying the JT, I_{load} , and the DC-link voltage has been performed.

Turn on-off delay time

The switching behavior of the power devices has been also adopted as a CM method [196-202]. In this case, the TSEPs are the voltage and current waveforms during the turn-on and turn-off of the DUT. This method is quite similar to the V_{ge} (or V_{gs}) TSEP method, but the JT monitoring can be performed online during the converter operations. More specifically, the delay, ΔD , at turn-on, between the collector current i_c and the gate-emitter voltage V_{ge} for an IGBT device (see Figure 4:7) is used as a TSEP [196-198], as well as the delay between the drain current i_d and the gate-source voltage V_{gs} for a MOSFET device. Bearing in mind the IGBT devices, the turn-on delay is of great interest because ΔD increases linearly with the temperature [198], it only depends on the dc-link voltage and it is not influenced by the value of i_c . More specifically, during the switching on time interval t_{on} , the gate current charges the gate-emitter capacitance C_{GE}



Figure 4:7 Simplified *i_c* -*V_{ge}* turn ON waveforms of an IGB at different temperature working operations.

that is connected in series with the gate resistance R_{Gint} . Therefore, the zero state waveform of the $v_{ge}(t)$ can be written as [194]:

$$v_{ge}(t) = V_G \cdot \left(1 - e^{t/\tau}\right)$$

$$\tau \approx R_{Gint}(T) \cdot C_{GE}(T)$$

$$t_{ON} \approx \tau(T) \cdot \ln\left(1 - \frac{V_{th}(T)}{V_G}\right)$$
(106)

where V_G is the driver gate-emitter voltage.

The dependence on the temperature of the turn-on delay ΔD can be analyzed by combining both the (9) and (11). V_{th} decreases as the temperature increases and the value of the time constant τ depends on the temperature variations too. It is worth underlining that the gate charge (the intrinsic gate capacitances) has a weak dependence on the temperature while the internal gate-resistance R_{Gint} has a stronger dependence on temperature due to the channel mobility μ that decreases at higher temperatures.

Figure 4:7 depicts the simulation of an ideal IGBT device during the turn-on at varying working operation temperatures (40 °C, 70°C and 100°C). The shift on the right of the waveforms is strictly related to the aforementioned temperature dependence. The previous method is valid also for the MOSFETs.

An advanced sensing circuit (voltage probes, FPGA, ADC) that records the transient evolutions of both V_{ge} and i_c waveforms have been proposed in [197]. The delay is calculated as the time interval between the time instant the rising edge of the V_{ge} is detected, and the rising edge of the current i_c (Figure 4:7). This method allows a sensitivity close to 2 ns/°C. Moreover, since a gate resistor with a large resistance improves the accuracy of the temperature measurements during the switching behavior of the converter but worsen the efficiency, a variable gate resistor has been proposed in order to set a higher value exclusively when the JT is measured [198].

Similarly, the turn-off delay can also be used as a TSEP, reaching a sensitivity level close to the one obtained with the turn-on delay method [199]. Other works have proposed an alternative sensing circuit for the JT estimation during the turn OFF [200]. However, the turn-off delay method does not attract interest because it is not linear at high-temperature operations, and the time delay depends greatly both on the i_c current and on the DC link voltage [201-202].

In general, the turn-on and turn-off TSEPs methods require high bandwidth sensors and an advanced sampling circuit for temperature measurement, that increase considerably the cost of the overall system. Furthermore, these methods usually require an external circuit to trigger a counter for the estimation of the turn-on and turn-off delay time.

Current and voltage change rate

In the last decade, the research has moved from the study of the electrical quantities (such as the voltage and current waveforms) to their derivative functions, which are observed during the device commutation, called dynamic thermo-sensitive electrical parameter (DTSEPs) methods. More specifically, the collector-emitter voltage change rate (dv_{ce}/dt) and the collector current change rate (di_c/dt) have been used as temperature estimators [203-207]. The dependence of both dv_{ce}/dt and di_c/dt with respect to the temperature has been explored theoretically and confirmed also experimentally [205-207].

As an example, the dV_{ce}/dt in an IGBT device can be approximated as [206]:

$$\frac{dv_{ce}}{dt} \approx \frac{1}{\tau_{gc}(T)} \left(\frac{V_{GE,ON} - V_{GE,OFF}}{1 + \binom{C_0}{g_m(T)\tau_{gc}(T)}} \right)$$
(107)

 $\tau_{gc} \approx R_{Gint}\left(T\right) \cdot C_{GC}(T)$

where C_O is the charge extraction capacitance, $V_{GE, ON}$, and $V_{GE, OFF}$ are the on-off gate driver voltages.

It is worth noting that the term dV_{ce}/dt depends on the physical parameters of the IGBT device and the temperature dependence is not easy to obtain. More specifically, the JT affects the dV_{ce}/dt through the MOS channel parameters such as the L_{ch} , W_{ch} , the emitter recombination parameter, the channel mobility, and so on. A detailed discussion of all the temperature parameter dependencies is given in [206]. The dependence of many parameters influencing the derivative quantities on the temperature strongly limits the use of this CM method for online JT measurement in practical power converter applications. A wide investigation of the IGBT maximum dv_{ce}/dt for the JT estimation has revealed the severe limits due to the influence of the control method, the DC link voltage, and the load current [206]. Likewise, the maximum dic/dt during turn-off as TSEPs has been also investigated [207]. Even in this case, the measurement of the current change rate has been performed by using an additional circuit able at capturing the current and voltage transient dynamics, which require both high bandwidth sensors and the use of voltage probes and Rogowski coil probes. Furthermore, this sensing circuit should be designed to avoid any disturbance, and also it has to be insensitive to the temperature variation of the system.

In the last generations of IGBT and SiC high power modules, the Kelvin emitter pin has been introduced. Such an additional pin involves in the package an integrated inherent parasitic inductance L_{eE} between the Kelvin pin and power emitters pin [208-210], as shown in Figure 4:8. The transient collector current characteristic during the turn OFF process has been introduced as a potential DTSEP, called maximum collector current falling rate $-dI_C/dt_{max}$ [210]: the collector current I_C flows in the inductance L_{eE} and the resulting voltage drop enables an easier investigation of the JT measurement.

Also in [207], both the static and dynamic behaviors of the stored carriers in the IGBT collector current during the falling rate have been analyzed. Furthermore, the influences of the physical parameters of the device on the temperature sensitivity of $-dI_c/dt_{max}$ have been fully investigated. However, several drawbacks of these methods are related to the strong dependence of the applied voltage and the gate resistance, and the thermal characterization can be done only offline.



Figure 4:8 IGBT module equivalent circuit.

Peak gate current

An innovative method for JT measurement in IGBTs and MOSFETs, based on the temperature-dependent of the internal gate resistance, has been studied in the last years. Firstly, the measure of the R_{Gint} in a power module has been already faced out by using a standard LCR meter [211], where a common approach is to consider the equivalent series resistance (ESR) of both the gate–emitter and gate–collector capacitance (see Figure 4:9(a)). Another method to estimate the R_{Gint} variation has been related to the measurement of the gate charge during the turn-on of the DUT [212]. Therefore, the peak gate current during the turn-on switching behavior has been assumed as a valid TSEP.

The JT measurement via the peak gate current can be studied during the standard charging cycles of the gate terminal. Considering an IGBT device, the turn-on process starts when the gate driver output voltage changes from a negative value to a positive one. Therefore, the gate current can be computed as the step response of a second-order RLC circuit [214] (see Figure 4:9(a)). The parasitic inductance L_G can be neglected and the peak current can be estimated by simply using the Ohm's law provided that the RLC circuit is overdamped. It is worth noting that the external gate resistance R_{Gext} has not a significant temperature dependence. Therefore, the temperature variation of R_{Gint} can be carried out by the measure of the peak current variation. In other words, the maximum value of the gate current provides a suitable strategy for the measurement of the chip JT. The measurement circuit is shown in Figure 4:9(b). The peak voltage on the external gate resistor during turn-on is measured with a peak detector circuit (a differential amplifier and a peak detector). Then, the acquired data are processed by an Analog to Digital converter to the microcontroller. This measurement circuit can be integrated into the gate driver, and the JT monitoring can be operated during the online operation of the converter. This method does not require calibration steps and more importantly, the voltage peak has a linear relationship with the temperature.

Only a few works have focused on the peak gate current as a TSEPs method. More specifically, the sensing circuit depicted in Figure 4:9(b) has been proposed in [215-216] where the JT has been esteemed in an IGBT power module. The authors have asserted that the proposed method has better accuracy for JT measurement compared to other TSEP methods in the literature. However, this method requires additional complex trigger circuits for the measurement of JT, which may introduce additional disturbance in the system. It is worth remembering that the aging of the power module may affect the internal gate resistances. Hence, a correction factor should be introduced for calibration with the aging of the device.



Figure 4:9 Gate driver RLC network. (a) Peak Detector Schematic to detect peak voltage over the external gate resistor.

4.2.5 Comparison of the CM methods

| Table 4:2 Summarization of different CM methods. |
|--|
|--|

| Legend | Method | Advantages | Disadvantages |
|--------|---|---|---|
| A | OPTICAL - Fiber optic | On line measurements High sensitivity High accuracy | Package modifica- tion High cost |
| В | OPTICAL - Photodiode sensor - internal | Online measurements Contactless | Technology not mature Package modifica- tion |
| С | OPTICAL - Photodiode sensor - external | Online measurements Contactless | Package modifica- tion |
| D | OPTICAL - IR camera | Spatial resolutionContactless | Package modification Poor response time Poor accuracy |
| E | OPTICAL - IR camera print | Spatial resolutionContactless | Package removal Poor response time Poor accuracy |

| _ | | • | On line measure- ments | • | Technology not mature |
|------|---------------------------------|--------------------|---------------------------|----------------|------------------------------|
| F Ac | Acoustic | • | Contactless | • | Noise sensitive |
| | | | | • | High cost |
| | | • | On line measure- ments | • | Package modifica- tion |
| G | Physical | • | Good linearity | • | Poor response time |
| | | • | Good sensitivity | • | Obsolete |
| Н | TTCs - NTC | • | On line measure- ments | • | Layout modifica- tion |
| | | • | Good linearity | • | Aging sensitive |
| I | TTCs - Diode | • | On line measure- ments | • | Layout modifica- tion |
| | | • | Good sensitivity | • | Poor linearity |
| | TSEP - On-state voltage, | • | High sensitivity | ٠ | High-cost sensing |
| J | low current | • | High linearity | • | Offline measure- ments |
| | | • | Easy calibration | | Offling massure |
| к | TSEP - Gate threshold | • | Good sensitivity | · | ments |
| | voitage | • | Good linearity | • | Unplug DUT |
| | | • | On line measure- | • | High-cost sensing |
| L | high current | | ments | • | Aging sensitive |
| | | • High linea | High linearity | • | Load current de- pendence |
| М | TSEP - Saturation current | • High sensitivity | High sensitivity | • | Offline measure- ments |
| | | | • | Poor linearity | |
| | | | | • | Offline measure- ments |
| Ν | TSEP - Gate turn OFF voltage | • | High linearity | • | High-cost sensing |
| | | | | • | Inaccurate meas- urement |
| 0 | TSEP - Turn on-off delay | • | On line measure- ments | • | Aging sensitive |
| | ume | • | High linearity | • | High-cost sensing |

| | | Less parameter- dependent | |
|---|---------------------------------------|---|---|
| Ρ | TSEP - Peak gate current | On line measurements Good linearity Easy to implement | Aging sensitiveHigh-cost sensing |
| Q | TSEP – Voltage-Current change rate | Good sensitivity Good linearity | Offline measurements High-cost sensing Gate resistance dependence |

Table 4:2 summarizes all the CM methods discussed previously. A comparison between the advantages and disadvantages of each approach has been outlined. It has been strongly highlighted the linear dependence with the temperature in the case of the optical, physical, and TTCs methods. Furthermore, the investigation revealed that the aforementioned methods and a few TSEPs methods (on-state voltage high current, turn on-off delay time, and peak gate current) can be adopted for online measurements in a real power converter application.

On the other hand, it has emerged as the TTCs requires a layout modification of the power module, notwithstanding the high linearity and independence from any physical parameters. Although the optical and acoustic methods are able at contactless monitoring the SOH, the technology is not yet well mature to be widespread in commercial solutions.

4.3 Intrusiveness criteria classification

The SOH of power electronic devices and systems can be evaluated with the research of innovative and enhanced strategies of CM methods. Several works have presented different CM strategies aiming at reaching a high level of accuracy and user-friendliness. On the other hand, the estimation of the intrusiveness level of the various CM methods has not yet drawn adequate attention.

In the following, some intrusiveness criteria and a classification that takes into account several factors that may affect the monitoring operations in a power conversion system have been discussed. Firstly, it has been analyzed whether a specific CM method needs to modify the device structure, then the impact of the monitoring on the converter operations as well as any modification on the overall power conversion system has been investigated. In the following, the three different intrusiveness criteria have been described, then the CM methods have been analyzed based on the introduced intrusiveness criteria.

4.3.1 Description of the Intrusiveness criteria

4.3.1.1 Power device modification

The device modification criterion relates to alterations or changes in the structure of the power devices (e.g. package, layout, and so on) due to a selected CM method. More specifically, the monitoring of the SOH needs the measurement of a physical or electrical quantity, the greater the device modifications necessary to perform the measurement the greater the CM method intrusiveness at the power device level.

In some cases, the CM method does not involve any changes in the device, such as the acoustic method [111-119] where an acoustic spectrometer is placed in the proximity of the DUT. Similarly, only the electrical quantities are measured at the device terminals in the case of TSEPs methods. On the other hand, other CM methods have to drill one or several holes on the package structure for the JT measurements. For example, the use of the fiber optical method [120-144] requires the inclusion of probes close to the die. Other CM methods require a particular modification in the packaging structure of the device, such as an integrated photodiode sensor for the JT estimation in a SiC power module. In this case, the sensor has to be integrated inside the device package and two terminal has to be added to the pin layout of the power module. An impressive modification of the internal layout of a power module may occur in the case of TTCs [152-161], where a diode or an RTDs is fabricated on the surface of the die. While in different works the package has to be partially removed or a complete lift-off of the package has to be performed for the measurements, such as the use of an IR camera when a painting solution is used at the device surface.

4.3.1.2 Modification of power conversion system operation

The conversion system operation criterion is based on the level of influence during the common working operation of a power converter. Several CM methods are not able to carry out the JT measurements during the online operation of the converter, and in this perspective, the power conversion system has to be shut off. Therefore, these CM methods are strongly intrusive especially when the power converter has to be unplugged from the application and a dedicated experimental setup has to be used for the monitoring. Most TSEPs methods are strongly invasive such as those based on the measurement of the on-state voltage under low current injection, the saturation current, the gate threshold voltage, the gate turn-off delay, the current change rate, and the voltage change rate [164-200]. On the other hand, the physical [145-151], optical [120-144], and acoustic methods [111-119] can be used during online converter opera-

tions. Furthermore, it must be taken into account also the influence of the voltage probes during the measurements, such as the probe bandwidth or o any contact of the probe with the DUT. For example, the thermocouples [145-151] are directly placed on the die surface, while in a TSEPs method [164-216] the voltage probes are connected in the external connectors.

4.3.1.3 System modification

The overall system can be assumed as the set of power converters, the load, and power supplies. The system modification criterion is based on the level of changes in the systems components and/or their interconnections, that can be also specific for a particular application. This criterion also takes into account the impact of a CM method in terms of the use of cumbersome tools for data acquisition, the need to open the chassis protection of the system, whereas the auxiliary circuitry may be bulky respect the applications and so on.

The system modification criterion strongly depends on the specific application and in this perspective, it has to be evaluated how the CM methods may impact in terms of size and modification on the overall system. For example, various optical methods [120-144] may have a different level of intrusiveness. Bearing in mind the IR camera used after painting the surface of the device asks for the opening of the converter system and therefore, it is extremely intrusive. While in the case of the integrated photodiode the level of intrusiveness is lower since the voltage drop is measured thought two BNC cables connected to an oscilloscope. The intrusiveness of the physical methods, such as the thermocouple, is higher than the previous ones because of an auxiliary circuit is needed in order to correlate the resistance variation with the JT estimation. Furthermore, in the case of several TSEPs methods [164-216], the intrusiveness level is even higher due to the need to use auxiliary switches or auxiliary signal conditioning circuits that are able to detect the JT variation in the power module.

4.3.2 Classification of the CM methods based on the intrusiveness criteria

The adoption of a unique figure of merit (FOM) representing the intrusiveness of a CM is highly desired. In this perspective, it is advantageous to first define an index of intrusiveness for each criterion. Therefore, a triplet of numerical marks, each in the range from 1 (nonintrusive) to 10 (fully intrusive), is assigned to a CM method according to the intrusiveness criteria. More specifically, for a given criterion the following procedure has been adopted to assign the marks to each CM method. A mark equal to 1 has been assigned to the methods fully nonintrusive. A CM method satisfies such a condition when there is not any change after the introduction of the method. After that, the CM methods that involve a slight modification have been analyzed. They are ranked according to the modification degree and a mark in the range 2-4 has been assigned to each of them according to my opinion. Similarly, the other methods involving an intermediate modification have been analyzed and a mark in the range 5-7 has been assigned to them. Finally, a mark between 8 and 10 has been assigned to the CM methods involving a strong modification. Although the assignment of the mark is subjective the rank of the CM methods according to an intrusiveness criterion can be considered almost unbiased.

4.3.2.1 FOM_D: Evaluation of the intrusiveness according to the Device modification criterion

The device modification criterion refers to alterations or changes in the structure of a power device due to a given CM method with respect to the same device without employing any CM. In this perspective, each CM method has been analyzed in-depth to reveal the intrusiveness at the device level and, thus the index of intrusiveness FOM_D has been assigned. The lowest value of FOM_D is assigned when no modification or packaging alterations are done neither by the manufacturer and the user. The FOM_D index increase as the package modifications increase (e.g. drills in the package to insert probes). Furthermore, FOM_D also increases in case of changes in the layout of the power module, and finally, FOM_D is set to the highest value when there is a complete package removal. In the following, the CM methods are ranked according to the device modification intrusiveness criterion, and the marks are assigned by adopting the procedure described before.

TSEPs CM methods present the lowest intrusiveness because the monitoring of the temperature can be carried out by exclusively measuring the electrical quantities on the device terminals. As evident, this approach does not require any modification of the device package or layout. Likewise, the acoustic methods (F) have also the lowest intrusiveness because an acoustic spectrometer records the acoustic emission, and then it is possible to estimate the state of the aging of the power module without any modification. The spectrometer may be placed close to the device, around a few centimeters, without any package modifications. In the field of optical CM methods, only the use of an IR camera (D) shows the lowest FOM_D. In this case, the measurement of the device temperature can be carried out by placing the IR camera until one meter from the DUT. Hence, the FOM_D has been set equal to 1 for all these methods. On the other hand, another optical CM method, that is the one adopting the fiber optic (A), it has been considered a little more intrusive with respect to the previous ones. In this case, a small device package modification has been introduced: two small holes in the package to insert the fiber optic wires close to the device chip. In this perspective, a FOM_D equal to 2 has been assigned. The intrusiveness of the optical CM methods based on photodiode sensor inside the package should be distinguished in two different scenarios:

- the sensor has been fabricated directly (*a priori*) into the package from the device manufacturer (B).

- the photodiode is placed a *posteriori* by a user (C).

In the first case, the overall package can be optimized by the manufacturer in order for the device to perform well meanwhile the modification could be limited. In this perspective, the device modifications are more than a simple hole on the package and hence, it has been assigned a FOM_D equal to 4. In the other scenario, the device package has to be opened and the photodiode placed by a user. Hence, the latter solution has higher intrusiveness with respect to the integrated one, and hence, it has been assigned a FOM_D equal to 5.

Whereas the DUT has been printed on the surface to increase its emissivity in the case of an IR camera (E), FOM_D must be set to a higher value due to the partial opening of the package structure. In this case, a FOM_D value equals to 6 has been assigned. In a similar way, the use of a TTCs such as the RTD (H) and integrated diodes (I) that are usually fabricated on the surface of the die allows the measurement of the JT by measuring the voltage drop. Therefore, the internal layout and the pin layout of the power module has to be redesigned from the original one. In this perspective, the FOM_D value that has been assigned 8 and 7 respectively because they require a substantial redesign of the device layout.

Finally, the CM method using the thermocouple (G) is the most intrusive due to the need to open the device package and to place the thermistor on the chip of the power (FOM_D equals to 9). It is worth remembering that the device package must be drilled to place the BNC cables for the measurements.



Figure 4:10 depicts a bar diagram that represents the values of FOM_D of each CM method.

Figure 4:10 FOM_D assigned to the CM methods

4.3.2.2 FOM_c: Evaluation of the intrusiveness according to the Conversion system operation criterion The conversion system operation criterion refers to the impact of a CM method on the normal working operations of a power converter. In this perspective, each CM method has been analyzed with the aim to fully understand the intrusiveness level, thus the index of intrusiveness FOM_c has been assigned. The lowest value of FOM_c is assigned when on-line monitoring is possible and there is not any physical contact between the conversion system and the measurement system (e.g. the measurement is performed without any probes or wires). It is worth underlining that the electrical probes have several limitations in their use such as limited bandwidth, intrinsic parasitic capacitances, and also, they cannot always be used for high voltage operations. In this perspective, the use of the voltage or current probes may be intrusive for the measurements. In the following, the CM methods are ranked according to the conversion operation intrusiveness criterion, and the marks are assigned by adopting the procedure described before.

It is a matter of fact that all the optical CM methods that are based on the measurement of the light brightness or the measurement of the emissivity change of material present the lowest intrusiveness. As an example, the use of an IR camera (D) is fully nonintrusive since the spatial measurement of the temperature gradient on the power module can be done without any direct contact with the converter. Furthermore, both the use of the optical fibers (A) or the integrated photodiodes (B) or the external photodiodes (C) introduce the lowest value of intrusiveness, where the monitoring of the JT can be carried out without direct contact and without interrupting the converter operations. In a similar way, the acoustic method (F) uses an acoustic spectrometer placed in the proximity of the device, and therefore, the converter is not affected by the use of this method. Hence, the *FOM_C* has been also set equal to 1 to all these CM methods.

The physical methods (G) and the TTCs (H and I), have a slightly higher level of intrusiveness with respect to the previous CM methods. More specifically, to perform the JT estimation, a voltage drop is measured by using a voltage probe on the external connectors of the power module. Therefore, the monitoring of the device SOH could be lightly affected by the sensing equipment that may introduce a little disturbance, thus a FOM_c equal to 2 has been assigned.

The TSEPs Turn off-on delay time (O) and peak gate current (P) CM methods are able to carry out the JT monitoring during online working operations. More specifically, for the turn-off delay, the time delay between the voltage and current waveforms are measured. Therefore, the use of at least two different probes (e.g. current probe and voltage probe) is mandatory. In a similar way, the peak current gate CM method requires the use of two voltage probes, and hence, the level of intrusiveness is comparable. In these cases, the FOM_c has been set equal to 4 for both the CM methods. The On-state voltage at higher current (L) also allows the monitoring of the JT during the online operation, but at cost of a high current injected into the device (30 A is a typical value). Notwithstanding the ability to monitor the JT during online operations, the pulsed current injected into the device must have a time width of hundreds of microseconds, and consequentially the converter operations have to be changed for a short time period from the typical functionalities. Furthermore, the measured TSEP is the on-state voltage, and hence, it requires an active clamping and extremely sensitive circuit to measure the lower onstate voltage, which increases the level of intrusiveness. In applications where the monitoring must be frequently or continuously performed such a CM method become equivalent to the ones that cannot be performed online since none of them is usable in such kinds of application. In this case, the FOM_c has been set equal to 8 due to the alteration in the converter system, which is strongly greater than the previous methods.

On the other hand, the On state voltage under low-level injection (J), the threshold voltage (K), the saturation current (M), the gate emitter voltage turn on-off (N), and the voltage-current change rate (Q) CM methods shows the highest level of intrusiveness. These TSEPs methods need to shut off the converter in order to perform the JT measurement. Moreover, these methods require the unplugging of the device and a dedicated circuit (such as switches, current generator, etc.) has to be used for monitoring. Consequently, a FOM_c equal to 10 as been assigned to them.

Figure 4:11 summarizes the values of FOMC for each specific CM method.



Figure 4:11 FOM_C assigned to the CM methods.

4.3.2.3 FOMs: Evaluation of the intrusiveness according to the System modification criterion The System modification criterion has been defined as the impact of a CM method on the conversion system (such as the chassis removal) as well as the introduction of components or modifications in the final application (e.g. the addition of auxiliary electrical circuits).

Several CM methods have been analyzed with the aim to highlight their intrusiveness level in order to rank them and to assign the index of intrusiveness *FOM*₅ based on the system modification criterion. The lowest value of *FOM*₅ is related to the case when limited modifications or addition of auxiliary electrical circuits are introduced in the final applications with respect to the scenario without CM.

The measurement of the JT by means of an integrated photodiode (B) only requires the measurement of the voltage drop on the resistor. This means that no auxiliary circuit or

external supply voltages are needed for the monitoring. Hence, the use of integrated photodiode has the lowest intrusiveness, thus a FOM_s equal to 1 has been assigned to the method of adopting an integrated photodiode. On the other hand, the adoption of an external photodiode (C) involves a little higher intrusiveness since it usually needs a signal conditioning and external power supply. Similar considerations are valid for TTCs based CM methods. The monitoring of the JT consists of a simple measurement of the voltage variation that can be carried out by using a voltmeter sensing circuit. Therefore, the intrusiveness level in the real system is very low. Then a FOM_s equal to 2 has been assigned to these three methods (C-H-I). Instead, the thermocouples (G) lead to a further slight increment in the level of intrusiveness because a simple and compact auxiliary circuit (e.g. Wheatstone bridge) able to detect the resistance changing with the temperature is usually employed. In a similar way, the turn on-off delay (O) and the peak current (P) CM methods have the same level of intrusiveness. In the first one, the measure of the JT can be carried out by measuring the time delay between the voltage and the current of the DUT. The time interval measurement can be done by using an FPGA or microcontroller that is able to detect the signal variations in a range of microseconds. The sensing circuit may be integrated into the gate driver, otherwise, the auxiliary sensing circuits are placed close to the converter system. Similarly, the peak gate current method uses two voltages probes and a detecting circuit that is able to measure the voltage variation on the gate resistance. A FOM_s equal to 3 has been assigned to them (G-O-P).

The JT monitoring through an IR camera (D) requires that the chassis of the converter system has to be removed, and the IR camera is placed at about one meter from the DUT. Besides, whereas the device surface must be varnished (E), the level of intrusiveness is higher due to this further modification. Hence, a FOM_s equal to 4 and 5 has been assigned in case of the use of an IR camera respectively, without and with the surface varnishing.

The on-state voltage at a low current injection (J) method requires a sensing circuit able at measuring a voltage variation during the switching on-off of the device. Furthermore, an additional active clamping circuit able at measuring a low voltage value. Hence, the auxiliary circuit and sensing circuits lead to intrusiveness comparable with the previous case, thus a *FOM*_S equal to 4 has been assigned. The threshold voltage (K) and gate emitter turn off voltage (N) CM methods require a low current generator for the calibration step. For example, considering an IGBT device, the gate and the collector terminals are shorted and then, the collector-emitter voltage drop is measured through a voltmeter. Hence, the gate driver has to be modified in order to perform the measurement. According to this, the *FOM*_S related to the system modification criterion has been set to 5. Similar reasoning can be carried out in case of the voltage-current change rate TSEP CM method.

The measurement of the JT in case of the on-state voltage at high injected current (L) CM method requires a high DC pulsed current source that feeds the DUT by using sever-

al switches and a voltmeter measures the voltage drop of the device. Furthermore, this CM method does not have high accuracy for the JT measurement, and hence, sophisticated and cumbersome sensing circuits are usually employed. Therefore, the aforementioned method shows a higher level of intrusiveness in the overall system and hence, *FOMs* has been set to 7. Similarly, the saturation current (M) CM method requires a DC source and a switch that can inject into the DUT a current pulse width of hundreds of microseconds. Furthermore, the JT monitoring can be carried out by using an auxiliary circuit that is able to detect the saturation current of the device. The *FOMs* has been set equal to the previous CM method since it involves similar modifications in the system.

It is worth underlining that the CM method based on the optical fiber (A) and the acoustic (F) methods are the most intrusiveness ones. In detail, the first method requires an advanced sensing circuit and a data acquisition station is usually employed. The data acquisition station generally consists of a PC server, several ADC converters, and a dedicated power supply system. It is worth noting that the data acquisition station may result in bulky and the integration with the preexisting applications is not easy. Likewise, the acoustic CM method uses an acoustic spectrometer that requires a specific circuit sensing that may be bulky to be easily integrated into the converter system. Furthermore, the chassis of the system has to be removed to place the acoustic spectrometer in the proximity of the power device. Thus the FOM_s has been set to 8 for these methods.

Figure 4:12 summarizes the *FOMs* value for each CM method for the System modification criterion.



Figure 4:12 FOMs assigned to the CM methods.

4.3.3 Figures of Merit (FOMs)

The adoption of a unique FOM representing the general level of intrusiveness of a CM is highly desired. It is important to underline that the FOM must be able to give a qualitative evaluation of the overall intrusiveness among the CM methods and therefore, to have a fast and practicable impact of intrusiveness. Such a general FOM can be obtained by properly combining the previous levels of intrusiveness, being each of them computed for a given criterion.

In the following, an overall FOM (*FOM*₀) that provides a numerical evaluation of the intrusiveness level for each CM method has been proposed. However, the aforementioned FOM provides a single value accounting for the three intrusiveness criteria, and hence, it may suffer from the biased marks proposed to compare the CM methods. In this perspective, an alternative strategy to estimate the level of intrusiveness of the CM methods has been based on the Pareto optimality [217].

4.3.3.1 Overall and application-based Figures of Merit.

An overall FOM, FOM_O assigned to a specific CM method, m, has been defined as the product of the three levels of intrusiveness FOM_D , FOM_C , and FOM_S . In detail:



$$FOM_O(m) = FOM_D(m) * FOM_C(m) * FOM_S(m)$$
(108)

Figure 4:13 FOM₀ assigned to the CM methods.

Figure 4:13 depicts the value of the FOM_o computed for each CM methods. The most invasive CM method results the TSEP - Saturation current (M). It is worth remembering that the monitor of the JT can be carried out only by switching off the power converter and furthermore, it requires the use of a DC source and a switch that controls the injection of a current pulse width of hundreds of microseconds into the DUT. Also, the JT monitoring can be estimated by using an auxiliary circuit that must be able to detect accurately the saturation current of the device. In this perspective, the high values both of FOM_c and FOM_s lead to an increase in the overall intrusiveness.

Other TSEPs CM methods, such as the On-state voltage under low and high current injection (J and L), Gate threshold voltage (K), Gate turn OFF voltage (N), and Voltage-Current change rate (Q) show also a high level of intrusiveness. These CM methods need to switch off the power converter and consequentially the FOM_c has been set to the maximum value. Furthermore, the FOM_s also has been set at a higher value, because it is usually used cumbersome equipment to measure the JT variation. Among the TSEPs CM methods, the Peak gate current (P) and Turn on-off delay time (O) shows a lower value of intrusiveness with respect to the aforementioned methods. More specifically, both CM methods are able to measure the JT without any converter shut off, and hence, the FOM_c does not impact the overall intrusiveness level. It is worth noting that the Physical CM method (G) shows a higher level of intrusiveness due to the high value of FOM_D , where the device package must be open in order to place the thermocouples.

The less intrusive CM methods are the ones using the integrated photodiode (B) and IR camera (D) according to the FOM₀. More specifically, the use of an integrated photodiode does not introduce a significant device modification, and the JT can be estimated during the online converter operations. Furthermore, system modification is negligible, because the key point of this CM method consists of the measurement of the voltage drop across the integrated photodiode. Similarly, the IR camera is able to measure the temperature of the DUT without any contact with the system. The *FOMs* is the highest contribution due to the need to remove the chassis on the application. The Acoustic CM method (F) also shows a low value of overall intrusiveness. In this case, the device modification is negligible and the SOH can be carried out during the online converter operations. The major contribution that increases the level of intrusiveness is due to system modification, where the acoustic spectrometer is usually cumbersome and it is not easily integrated into a preexistent power application.

Finally, it has been defined as invasiveness FOM f_a , which is application related. More specifically, the FOM takes into account all the intrusiveness criteria, but the importance of each of them in the specific application is considered. For a given CM method (*m*) and a specific application (*a*), the FOM is computed as follows:

$$f_a = w_{a,1} \cdot FOM_D(m) + w_{a,2} \cdot FOM_C(m) + w_{a,3} \cdot FOM_S(m)$$
(109)

where $w_{a,1}$, $w_{a,2}$, and $w_{a,3}$ are weights that strictly depend on the specific application and their sum is equal to 1. For example, in a wind farm, the on-line CM is fundamental. Therefore, in such an application, $w_{a,2}$ is set close to 1 while the others are nearly 0.

4.3.3.2 Intrusiveness comparison based on the Pareto optimality

As said before, although the assignment of the marks is subjective since it is based on my personal opinion, the rank of the CM methods according to an intrusiveness criterion is almost unbiased. Therefore, the previous overall FOM, which provides a single value accounting for the three intrusiveness criteria could suffer biased marks when they are adopted to compare the CM methods. A strategy to compare the methods according to the rank instead of the mark is then important. With this in mind, the Pareto optimality [217] reasoning has been adopted as described in the following.



Figure 4:14 FOM_D Vs FOM_S represents the center of the circle, while the circle radius is set proportionally to FOM_c

Figure 4:14 reports a rough classification based on a simple visual inspection that aims to concurrently compare the CM methods in terms of the three intrusiveness criteria. More specifically, each method has been represented by a circle in the figure. The cou-

ple of values FOM_D - FOM_S represent the center of the circle, while the circle radius is set proportionally to FOM_C . More specifically, whether a CM method is placed on the left, it is better than one in the right, in terms of intrusiveness with respect to the FOM_D (xaxis). But the former is worse in terms of FOM_S (y-axis) if it is placed over the latter, otherwise the former present lower intrusiveness with respect to both criteria. On the other hand, the last condition does not imply that the former is better than the other at all since the FOM_C (circle radius) must be also considered. To formalize the comparison, some concepts are borrowed from the Pareto efficiency analysis. A CM method, Ψ , is said optimum, according to the Pareto optimality when [217]:

In other words, Ψ is said optimum when there is not any CM method better of it. The green circles represent all the optimum CM methods, that is the methods satisfying (15), while the red circles represent the others. It this perspective, all the CM methods that have been represented with a green circle belong to the Pareto front.

Figure 4:14 highlights that Acoustic (F), On state voltage at high current injection (L), Saturation current (M), Gate emitter voltage turn off (N), Voltage current change rate (Q), the Threshold voltage (K), On state voltage at low current injection (J), Turn off delay time (O), IR camera (D) and Peak gate current (P) have the same level of intrusiveness in terms of FOM_D . On the other hand, the methods L, M, N, Q, K, and J are worse than the method O (which is equivalent to D and P) in terms of both the two other criteria. Consequently, L, M, N, Q, K, and J are not optima (red circles). The lowest index of intrusiveness in terms of converter operation criterion (FOM_C) (it is depicted as a smaller radius of the circle) has been assigned to method F. Therefore, the CM method F is better than O from this criterion point of view, although it has a higher value of intrusiveness in terms of the system modification criterion (greater FOM_S). Therefore, both O, D, P, and F belong to the Pareto front and, consequentially, they are optimum in terms of intrusiveness according to the definition in (15).

The circle related to the CM method A presents the same radius of the one representing F, and it is located at the same height, that the same FOM_c and FOM_s values have been assigned to them, but A is not optimum since present a worse value of FOM_D .

The remaining CM methods such as the ones based on the external photodiode (C), the IR camera print (E), the TTCs-diode (I), the TTCs-NTC (H), and the Thermocouple (G) are located in the upper right of the CM method based on the integrated photodiode (B). Moreover, none of them presents a lower FOM_c (i.e. lower radius) than B. Then, they are not optima (red). The circle related to method B is on the right of O and F (i.e. B is worse of them in terms of device intrusiveness). On the other hand, the CM method B

presents the lowest intrusiveness in the conversion system (FOM_s). Therefore, B is an optimum CM method according to 15 (green circle), regardless of the fact it presents a negligible impact on the converter operation (FOM_c).

Chapter 5 Non-invasive SiC MOSFET Temperature Estimation Method based on the unintentional Light Emission from the Intrinsic Diode

5.1 Introduction

As evident from the analysis performed in the previous chapter, online junction temperature (T_j) monitoring has been widely studied in the literature [207],[218],[219]. The TSEPs methods are well-established solutions but they require an adequate calibration and decoupling procedure [220].

Other methods such as acoustic monitoring [117],[119] optical methods [132],[134],[136],[139], and so on guarantee a high accuracy, low complexity, and a fast transient response, without the aforementioned TSEPs drawback. Optical methods are common techniques but are rarely easy enough for the online temperature estimation.

The exploitation of the light emission phenomenon from the body diode of a SiC device may overcome these limitations. The electroluminescence phenomenon refers to the photoemission from a SiC material and it is related to its intrinsic body diode that operates in the third quadrant. The light brightness in the SiC intrinsic body diode depends on the junction temperature and on the forward current that flows into the diode. The light intensity decrease at high junction temperature being equal to the current injected. Thus, the electroluminescence phenomenon can be a suitable indicator of the temperature of the SiC MOSFET chip during a general power converter application.

As widely discussed in the previous chapter, the results of the comparison among various CM methods for power device has highlighted that the use of the Integrated photodiode is one of the best choices since it involves the lowest intrusiveness level. By considering the SiC MOSFET intrinsic diode as an unintentional diode, in this chapter, a new CM method adopting the Integrated photodiode in a SiC MOSFET module has been proposed. In recent years, some works have already investigated methods for a non-intrusive and online junction temperature measurement based on the detection of the emitted light intensity by the intrinsic body diode [143-144],[221]. All these solutions use expensive sensing circuits, such as silicon photomultiplier and active circuit for biasing the photodetector. This involves an increase in the complexity of the online temperature estimation system and the repeatability of the measurements is difficult. In this perspective, the proposed system presents an innovative method to characterize the junction temperature that overcomes the previous limitations. In detail, the gate-source voltage of the SiC MOSFET has been shorted and a pulsed current has been injected into the intrinsic body diode to emulate a potential third quadrant operation. The transient light emission has been acquired by using an inexpensive, simple, and effective passive sensing circuit, such as a commercial photodiode and a resistor connected in parallel. After that, the output voltage values of the photodiode at different temperatures and current pulses have been analyzed offline with the aim to obtain a dependence between the output voltage and the junction temperature. The average of the voltage values in two different time subinterval has been computed. Even if the solution is extremely simple, the proposed method highlights a good ability in the junction temperature estimation. The investigation provides a relation between the transient light emission with the junction temperature and the current magnitude. In the following, the proposed experimental setup has been widely described and several results have been shown.

5.2 Electroluminescence and prior art



Figure 5:1 Luminescence intensity under different forward currents at 120 °C. (spectrum adopted from [221]).

In recent years, the electroluminescence phenomena related to the photoemission from a SiC material has been already used to characterize and monitor the state of the health of a commercial SiC MOSFET. More specifically, the electroluminescence phenomenon is related to the intrinsic SiC body diode that operates in the bipolar conduction mode. There are three kinds of recombination processes in the low doped region: the direct band, Auger, and defect energy level states assisted recombination. The latter is related to the photon emission of the extra energy of the free carriers [222]. Bearing in mind the relationship between the peak wavelength λ related to the emitted photons and the bandgap energy *E* [223]:

$$\lambda = \frac{hc}{E} \tag{111}$$

where *h* is the Planck constant and *c* is the speed of light in a vacuum, it is possible to evaluate the peak wavelength of the emitted light for a 4H-SiC material. The peak wavelength is around 380 nm, which is the visible wavelength range of the blue light. Figure 5:1 shows the spectrum of generated light and the luminescence intensity under different forward currents at 120 °C for a 4H-SiC MOSFET [221].

As widely studied in the literature, the light color emitted by the SiC intrinsic body diode depends on the junction temperature T_j , which affects the wavelength, since the generated light is the result of the radiative recombination process [142], [221]. Moreover, the brightness depends on the magnitude of the forward current that flows into the diode[224-225]. It is worth noting that the light intensity decreases at higher T_j at the same current.

In many practical applications, the intrinsic SiC body diode may be used instead of the antiparallel SiC Schottky barrier diode, by providing a cost-effectiveness solution without sacrificing the conversion efficiency [226]. Thus, the aforementioned electroluminescence phenomenon can be a suitable indicator of the temperature of the SiC MOSFET chip in a generic power converter application. Only a few works in literature have been already focusing on this topic. The authors in [221] have proposed an online T_i estimations by measuring the light brightness of a SiC intrinsic body diode in a power module during the PWM inverter operations. This method has a low sensitivity and the measure of the light intensity requires an injected pulsed current with a large time width of about 15 ms, which is higher than the switching period in actual power converter applications. Furthermore, the method requires an active circuit sensing that may be bulky and consequentially, the monitoring system cannot be integrated easily in the device package. In [144] the temperature-dependent changes in the spectrum of the light emission of a SiC intrinsic body diode have been also studied. The approach has been validated by means of static characterization and dynamic measurements. Although the approach shows good sensitivity and it is able to detect the light variation in case of a pulsed current of about 1µs, the experimental setup is extremely expensive (two Silicon Photomultiplier photon detectors) and it is too bulky to be integrated inside the package of the device for an online T_j measurement.

5.3 Experimental setup

A schematic of the experimental setup is depicted in Figure 5:2. The passive sensing circuit consists of a photodiode from Vishay [227] and a resistor R_s of 500 k Ω connected in parallel (yellow block in Figure 5:2).

The main features of the photodiode have been summarized in Table 5:1. It can operate both in photovoltaic or photoconductive mode. The latter has a higher dark current that varies directly with the photodiode temperature. In other words, the measurement performed by the photodiode in photovoltaic mode is less sensitive to the temperature of the photodiode itself, thus enabling the repeatability of the measurement of the SiC device. Therefore, the photovoltaic mode has been chosen to obtain unbiased operations for ultra-low light level applications.



Figure 5:2 Experimental setup schematic.



Figure 5:3 Picture of the experimental setup.

Table 5:1 Main features of the Silicon PIN photodiode at 25°C.

| Parameters | Value |
|-----------------------------|----------------|
| Reverse dark current | 0.123 nA |
| Range of spectral bandwidth | 430 to 1100 nm |
| Noise equivalent power | 400 pW/VHz |

Table 5:2 Main features of the power module at 25°C.

| | Parameters | Value |
|------------|-------------------------|-----------------|
| | Breakdown Voltage | 1200 V |
| MOSFET | On resistance | 56 mΩ |
| | Drain Current | 40 A |
| | Forward Voltage | 6.4 V |
| BODY DIODE | Forward Resistance | 50 mΩ |
| | Maximum forward current | 160 A (PW≤10µs) |



Figure 5:4 Photodiode voltage output at different T_{J} and different diode forward current values: (a) I_{L} =4,5A (b) I_{L} =7,5 A (c) I_{L} = 11,5 A.

Figure 5:5 Photodiode voltage output at different T_j and different diode forward current values: (a) I_L =4,5A (b) I_L =7,5 A (c) I_L = 11,5 A.

The photodiode absorbs photons and it generates a current proportional to the incident light that, in turn, depends on the SiC devices temperature. Therefore, the SiC devices temperature is translated into an output voltage (*Vo*).

A 1.2 kV/40 A power module from Semikron [228] (main feature in Table 5:2) has been used for the experimental characterization (black dashed block in Figure 5:2). M1 has been chosen as a device under test (DUT).

The gate and source terminals of M1 have been shorted and the source of M2 has been connected to the negative terminal of the DC source. Similarly, M3 and M4 are kept off by short-circuiting the gate to the source to avoid undesired turn on. A freewheeling diode D_F has been connected in parallel to the SiC MOSFET, M1.

The intrinsic body diode has been biased with a current pulse, I_L , by means of a DC source paralleled with a capacitor C_{in} (50 mF). The pulsed current has been controlled by the MOSFET M2 and the gate driver HCPL-3120 [229]229. The function generator produces a square voltage waveform, V_{in} , with a constant period and duty cycle equal to one-third. The gate output control voltage, V_{contr} , is equal to 15V, and it is in series with a resistor R_{out} (10 Ω).

Figure 5:3 displays the experimental setup. The module top lid has been removed and the photodiode has been placed approximately 2 mm over the SiC MOSFET chip to detect the light emission. There is no voltage limitation to the application because the photodiode is not in contact with the SiC MOSFET.

It is worth noting that the measurements of the light brightness have been carried out without removing the metallization in place in the power module. Furthermore, the measurements have been obtained without removing the dielectric gel between the die and the photodiode sensor. Further research would be interesting to deepen whether the use of the dielectric gel can affect the light brightness measurements.

The SiC module has been placed on a temperature-controlled hot plate. For the sake of simplicity, the measurements have been done right after powering up, so that it has been reasonable to assume that T_j is equal to the plate one.

5.4 Experimental results

Figure 5:4 and Figure 5:5 depict the transient waveforms of the sensing output voltage V_0 for three different current amplitudes I_L (4.5 A, 7.5 A, and 11.5 A) at three different case temperatures (40 °C, 70°C and 100°C) by using a pulsed injected current at 333Hz and 3 kHz, respectively.

From the experimental evidence, the measurements confirm that the light output strictly depends on the combination of the current magnitude I_L and T_j . More specifically, the voltage output, V_O , decreases for increasing temperature and it increases at higher current injections.

The measured voltage V_0 fluctuates and it is not constant for each temperature measurement. This is more evident in the case of the highest frequency. The problem is due to the simple and extremely inexpensive sensing circuit that has been employed for the measurement, where no active filters or EMI shielding have been used for the light brightness measurements. Moreover, the commercial photodiode used in the experimental setup has an intrinsic time constant in the range of tens of microseconds, which is not enough for switching frequencies higher than 10 kHz. It is worth to underline that the V_0 value is in the range of tens of mV.

Figure 5:4 shows the results in the case of a 333 Hz pulsed current with an on-time (T_{on}) of 1 millisecond, whereas Figure 5:5 shows the results in the case of 3 kHz, T_{on} = 100 microsecond. It is worth noting that in some cases, linearity is very low and some overlaps appear.



Figure 5:6 Experimental photodiode voltage output at T_j =40 °C with a I_L = 11,5 A at 3 kHz. Both the subinterval Δt_1 and Δt_2 represent the last 30% of the T_{on} and T_{off} time.



Figure 5:7 Dependence of V_{OM} with T_j and I_L at 333 Hz.



Figure 5:8 Dependence of V_{OM} with T_j and I_L at 3 kHz.

For instance, Figure 5:5(c) shows that the waveforms at 100°C (green trace) partially overlaps the one at 70°C (blue trace).

Notwithstanding the aforementioned limitations, the proposed strategy always makes able to distinguish fairly easily the temperature variations in a SiC power module in case of a pulsed current until 10 kHz. The simple key idea is to consider the average of the voltage values of the output voltage V_O during a portion of the T_{on} (i.e. Δt_1 of Figure 5:6) and T_{off} (i.e. Δt_2 of Figure 5:6). As depicted in Figure 5:6, both the subintervals Δt_1 and Δt_2 are measured at the last 30% of the T_{on} and T_{off} periods. We have calculated the average value of V_O during Δt_1 , named V_{OH} (red box of Figure 5:6) and Δt_2 , named V_{OL} (the green box of Figure 5:6).

The difference between V_{OH} and V_{OL} has been calculated and averaged out for fifteen consecutive time periods as follows:

$$V_{O,mean} = \frac{\sum_{i=1}^{15} (V_{OH,i} - V_{OL,i})}{15}$$
(112)

| Current $I_{L}[A]$ | $T_{on} = 1 \text{ ms}$ | | $T_{on} = 1$ | 00 µs |
|--------------------|-------------------------|--------|--------------|--------|
| | (333 kHz) | | (3 kł | Hz) |
| | α [mV/°C] | β [mV] | α [mV/°C] | β [mV] |
| 4.5 | -0.09 | 12.75 | -0.04 | 5.01 |
| 7.5 | -0.07 | 16.64 | -0.09 | 11.61 |
| 11.5 | -0.10 | 20.77 | -0.10 | 15.47 |

Table 5:3 Coefficients of the polynomial functions Equation (113)

Table 5:4 Coefficients of the polynomial functions Equation (114)

| Current IL [A] | <i>T</i> _{on} = 1 ms | <i>T_{on}</i> = 100 μs | |
|----------------|-------------------------------|--------------------------------|--|
| | (333 kHz) | (3 kHz) | |
| 4.5 | ± 4.6 °C | ± 10 °C | |
| 7.5 | ± 5.5 °C | ± 5.04 °C | |
| 11.5 | ± 3.9 °C | ±4 °C | |

Such a quantity has been used as an indicator of the device temperature. This approach has turned out to be extremely effective, especially, in the case of a measure at higher frequencies.

The dependence of V_{OM} on T_j and I_L , both for 333 Hz and 3 kHz and for different currents has been reported in Figure 5:7 and Figure 5:8, where the vertical axis is $V_{O,mean}$ and the horizontal one is T_j . Fitting lines have been added to the figures.

Since V_{OM} varies linearly with T_j , $V_{O,mean}$ can be estimated by means of a linear interpolation of the measured quantities (dashed lines in Figure 5:7 and Figure 5:8). Hence, the $V_{O,est}$ can be written as:

$$V_{O,est}(T_j) = \alpha T_j + \beta \tag{113}$$

where the term α is the slope and β is the intercept of the function. Table 5:3 summarizes the coefficients of the polynomial functions for each test set. Rearranging eq. (113) in terms of T_j yields an estimate $T_{j,est}$ of junction temperature as a function of $V_{O,est}$:

$$T_{j,est}(V_{O,est}) = \frac{V_{O,est} - \beta}{\alpha}$$
(114)

The proposed method can be integrated into a power module for the T_j monitoring during real operations. In detail, the evaluation of $V_{O,mean}$ at a known current enables the estimation of T_j by using (114).

Furthermore, the measuring system is very promising with respect to the measurement uncertainty, where the oscilloscope has a sensitivity of 1 mV and the photodiode has an uncertainty of about 20%, based on the datasheet. More specifically, the measurement uncertainty of the T_j has been calculated by referring (114), where it has been considered the ratio between the uncertainty of the voltage terms, $V_{O,est}$, and β , respect the uncertainty of the slope α . Table 5:4 shows the calculated uncertainties at different currents and frequencies. It is worth noting that the uncertainty related to the $T_{j,est}$ decreases at increasing currents. The worst-case accuracy occurs at 3 kHz and 4.5 A, showing a reasonable uncertainty of ±10°C.

References

- Zubi, G.; Dufo-López, R.; Pasaoglu, G.; Pardo, N. Techno-economic assessment of an off-grid PV system for developing regions to provide electricity for basic domestic needs: A 2020-2040 scenario. *Applied Energy* 2016, 176, 309-319.
- Palacios-Garcia, E.J.; Chen, A.; Santiago, I.; Bellido-Outeiriño, F.J.; Flores-Arias, J.M.; Moreno-Munoz, A. Stochastic model for lighting's electricity consumption in the residential sector. *Energy and Buildings* 2015, 89, 245-259.
- 3. Schmid, G. et al. Numerical analysis of a vertical double-pipe single-flow heat exchanger applied in an active cooling system for high-power LED street lights. *Applied Energy* **2017**, 195, 426-438.
- 4. Xu, X.; Collin, A.; Djokic, S. Z.; Langella, R.; Testa, A.; Drapela, J. Experimental evaluation and classification of LED lamps for typical residential applications. IEEE PES Innovative Smart Grid Technologies Conference Europe, Torino, 2017, pp. 1-6.
- 5. Reames, T.G.; Reiner, M.A.; Stacey, M.B. An incandescent truth: Disparities in energy-efficient lighting availability and prices in an urban U.S. county. *Applied Energy* **2018**, 218, 95-103.
- Chies, L.; Costa, M. A. D.; Bender, V. C. Improved design methodology for LED lamps. IEEE 24th International Symposium on Industrial Electronics, Buzios, 2015, pp. 1196-1201.
- 7. Raciti, A.; Rizzo, S.A.; Susinni, G. Parametric PSpice Circuit of Energy Saving Lamp Emulating Current Waveform. *Applied Sciences* **2019**, *9*, 152-180.
- 8. Ramljak, I.; Amir Tokić, A. Harmonic emission of LED lighting. AIMS Energy 2020, 8, 1–26.
- 9. Nagaraj, C.; Manjunatha Sharma, K. Fuzzy PI controller for bidirectional power flow applications with harmonic current mitigation under unbalanced scenario. *AIMS Energy* **2018**, *6*, 695– 709.
- 10. Drapela, J.; Langella, R.; Slezingr, J. Testa, A. A Tunable Flickermeter to Account for Different Lamp Technologies. *IEEE Transactions on Power Delivery* **2017**, 32, 872-880.
- 11. IEEE-1459-2010 Standard—IEEE Standard Definitions for the Measurements of Electric Power Quantities under Sinusoidal, Nonsinusoidal, Balanced or Unbalanced Conditions; IEEE: New York, NY, USA, 2010.
- Chen, W.; Li, S.N.; Hui, S.Y.R. A comparative study on the circuit topologies for offline passive light-emitting diode (LED) drivers with long lifetime & high efficiency. In Proceedings of the 2010 IEEE Energy Conversion Congress and Exposition (ECCE), Atlanta, GA, USA, 12–16 September 2010; pp. 724–730.
- 13. Lee, B.; Kim, H.; Rim, C.; Robust Passive LED Driver Compatible with Conventional Rapid-Start Ballast. *IEEE Trans. Power Electron.* **2011**, *26*, 3694–3706.
- 14. Hui, S.Y.; Li, S.N.; Tao, X.H.; Chen, W.; Ng, W.M. A Novel Passive Offline LED Driver with Long Lifetime. *IEEE Trans. Power Electron.* **2010**, *25*, 2665–2672.
- 15. Wang, Y.; Alonso, J.M.; Ruan, X. A Review of LED Drivers and Related Technologies. *IEEE Trans. Ind. Electron.* **2017**, 64, 5754–5765.
- Yada, T.; Katamoto, Y.; Yamada, H.; Tanaka, T.; Okamoto, M.; Hanamoto, T. Design and Experimental Verification of 400-W Class LED Driver with Cooperative Control Method for Two-Parallel Connected DC/DC Converters. *Energies* 2018, 11, 2237.

- Pinto, R.A.; Cosetin, M.R.; Marchesan, T.B.; Cervi, M.; Campos, A.; Do Prado, R.N. Compact lamp using high-brightness LEDs. In Proceedings of the IEEE Industry Applications Society Annual Meeting, Edmonton, AB, Canada, 5–9 October 2008; pp. 1–5.
- Li, Y.Y.-C.; Chen, C.-L. C. A novel single-stage high-power-factor AC-to-DC LED driving circuit with leakage inductance energy recycling. *IEEE Trans. Ind. Electron.* 2012, 59, 793–802.
- Alonso, J.M.; Vina, J.; Vaquero, D.G.; Martinez, G.; Osorio, R. Analysis and design of the integrated double buck-boost converter as a high-power-factor driver for power-LED lamps. *IEEE Trans. Ind. Electron.* 2012, 59, 1689–1697.
- Qu, X.; Wong, S.C.; Tse, C.K. Resonance-assisted buck converter for offline driving of power LED replacement lamps. *IEEE Trans. Power Electron.* 2011, 26, 532–540.
- Qin, Y.; Chung, H.; Lin, D.Y.; Hui, S.Y.R. Current source ballast for high power lighting emitting diodes without electrolytic capacitor. In Proceedings of the 34th Annual Conference of IEEE Industrial Electronics, Orlando, FL, USA, 10–13 November 2008; pp. 1968–1973.
- Hwu, K.-I.; Tai, Y.-K.; He, Y.-P. Bridgeless Buck-Boost PFC Rectifier with Positive Output Voltage. *Applied Science* 2019, 9, 3483.
- Ma, H.; Lai, J.; Feng, Q.; Yu, W.; Zheng, C.; Zhao, Z. A Novel Valley-Fill SEPIC-Derived Power Supply without Electrolytic Capacitor for LED Lighting Application. *IEEE Trans. Power Electron.* 2012, 27, 3057–3071.
- Hariprasath, S.; Balamurugan, R. A valley-fill SEPIC-derived power factor correction topology for LED lighting applications using one cycle control technique. In Proceedings of the *International Conference on Computer Communication and Informatics*, Coimbatore, India, 4–6 January 2013; pp. 1–4.
- Wu, X.; Wang, Z.; Zhang, J. Design Considerations for Dual-Output Quasi-Resonant Flyback LED Driver with Current-Sharing Transformer. *IEEE Trans. Power Electron.* 2013, 28, 4820– 4830.
- Shagerdmootaab, A.; Moallem, M. Filter Capacitor Minimization in a Flyback LED Driver Considering Input Current Harmonics and Light Flicker Characteristics. *IEEE Trans. Power Electron.* 2015, 30, 4467–4476.
- Arias, M.; Lamar, D.G.; Linera, F.F.; Balocco, D.; Aguissa Diallo, A.; Sebastián, J. Design of a Soft-Switching Asymmetrical Half-Bridge Converter as Second Stage of an LED Driver for Street Lighting Application. *IEEE Trans. Power Electron.* 2012, *27*, 1608–1621.
- Kim, C.; Kim, E.; Shin, H. A study on the power LEDs drive circuit design with asymmetrical half-bridge resonant converter. In Proceedings of the International Conference on Electrical Machines and Systems, Tokyo, Japan, 15–18 November 2009; pp. 1–4.
- Arias, M.; Diaz, M.F.; Cadierno, J.E.R.; Lamar, D.G.; Sebastián, J. Digital Implementation of the Feedforward Loop of the Asymmetrical Half-Bridge Converter for LED Lighting Applications. *IEEE J. Emerg. Sel. Top. Power Electron.* 2015, 3, 642–653.
- Castro, I.; Martin, K.; Vazquez, A.; Arias, M.; Lamar, D.G.; Sebastian, J. An AC–DC PFC Single-Stage Dual Inductor Current-Fed Push–Pull for HB-LED Lighting Applications. *IEEE J. Emerg. Sel. Top. Power Electron.* 2018, 6, 255–266.
- Mohamadi, A.; Afjei, E. A single-stage high power factor LED driver in continuous conduction mode. In Proceedings of the 6th Power Electronics, Drive Systems & Technologies Conference (PEDSTC2015), Tehran, Iran, 3–4 February 2015; pp. 462–467.
- Wang, L.; Zhang, B.; Qiu, D. A Novel Valley-Fill Single-Stage Boost-Forward Converter with Optimized Performance in Universal-Line Range for Dimmable LED Lighting. *IEEE Trans. Ind. Electron.* 2017, 64, 2770–2778.
- Chen, S.-M.; Liang, T.-J.; Yang, L.-S.; Chen, J.-F. A safety enhanced, high step-up DC-DC converter for AC photovoltaic module application. *IEEE Trans. Power Electron.* 2012, 27, 1809–1817.
- Cheng, C.; Chang, C.; Cheng, H.; Tseng, C. A novel single-stage LED driver with coupled inductors and interleaved PFC. In Proceedings of the 9th International Conference on Power Electronics and ECCE Asia (ICPE-ECCE Asia), Seoul, Korea, 1–5 June 2015; pp. 1240–1245.
- Cheng, C.; Cheng, H.; Chung, T. A novel single-stage high-power-factor LED street-lighting driver with coupled inductors. In Proceedings of the IEEE Industry Applications Society Annual Meeting, Lake Buena Vista, FL, USA, 6–11 October 2013; pp. 1–7.
- Duan, R.-Y.; Lee, J.-D.; High-efficiency bidirectional DC-DC converter with coupled inductor. *IET Power Electron.* 2012, 5, 115–123.
- Vecchia, M.D.; Van den Broeck, G.; Ravyts, S.; Tant, J.; Driesen, J. Modified step-down seriescapacitor buck converter with insertion of a Valley-Fill structure. *IET Power Electron.* 2019, 12, 3306–3314.
- Xu, S.; Shen, W.; Qian, Q.; Zhu, J.; Sun, W.; Li, H. An efficiency optimization method for a high frequency quasi-ZVS controlled resonant flyback converter. *In Proceedings of the* IEEE Applied Power Electronics Conference and Exposition (APEC), Anaheim, CA, USA, 17–21 March 2019; pp. 2957–2961.
- 39. Tabisz, W.A.; Gradzki, P.M.; Lee, F.C.Y. Zero-voltage-switched quasi-resonant buck and flyback converters-experimental results at 10 MHz. *IEEE Trans. Power Electron.* **1989**, 4, 194–204.
- 40. Gritti, G.; Adragna, C. Analysis, design and performance evaluation of an LED driver with unity power factor and constant-current primary sensing regulation. *AIMS Energy* **2019**, *7*, 579–599.
- Zhang, S.; Liu, X.; Guan, Y.; Yao, Y.; Alonso, J.M. Modified zero-voltage-switching singlestage LED driver based on Class E converter with constant frequency control method. *IET Power Electron.* 2018, *11*, 2010.
- 42. Jeng, L.S.; Peng, T.M.; Hsu, Y.C.; Chieng, H.W.; Shu, P.; J. TI. Quasi-Resonant Flyback DC/DC Converter Using GaN Power Transistors. *World Electr. Veh. J.* **2012**, *5*, 567–573.
- 43. Wang, Z.; Wu, X.; Chen, M.; Zhang, J. Optimal design methodology for the current-sharing transformer in a quasi-resonant (QR) flyback LED driver. In Proceedings of the Twenty-Seventh Annual IEEE Applied Power Electronics Conference and Exposition (APEC), Orlando, FL, USA, 5–9 February 2012; pp. 2372–2378.
- Li, J.; Liang, T.; Chen, K.; Lu, Y.; Li, J. Primary-side controller IC design for quasi-resonant flyback LED driver. In Proceedings of the IEEE Energy Conversion Congress and Exposition (ECCE), Montreal, QC, Canada, 20–24 September 2015; pp. 5308–5315.
- Zhang, J.; Zeng, H.L.; Wu, X.K. An Adaptive Blanking Time Control Scheme for an Audible Noise-Free Quasi-Resonant Flyback Converter. *IEEE Trans. Power Electron.* 2011, 26, 2735– 2742.
- Chen, K.; Chiu, C.C.; Lin, M.; Yeh, C.P.; Lin, J.M.; Chen, K.H. Quasi-Resonant Control with Dynamic Frequency Selector and Constant Current Startup Technique for 92% Peak Efficiency and 85% Light-load Efficiency Flyback Converter. *IEEE Trans. Power Electron.* 2013, 29, 4959– 4969.
- Bodetto, M.; Aroudi, A.E.; Cid-Pastor, A.; Al-Numay, M.S. Improving the Dimming Performance of Low-Power Single-Stage AC–DC HBLED Drivers. *IEEE Transactions on Industrial Electronics* 2017, 64, 5797-5806.
- Li, S.; Chen, H.; Tan, S.; Hui, S. Y.; Waffenschmidt, E. Power Flow Analysis and Critical Design Issues of Retrofit Light-Emitting Diode (LED). *IEEE Transactions on Power Electronics* 2015, 30, 3830-3840.
- 49. Ribas, J. et al. LED Series Current Regulator Based on a Modified Class-E Resonant Inverter. *IEEE Transactions on Industrial Electronics* **2018**, *65*, 9488-9497.
- Heidari, M. et al. Analysis of the energy efficiency potential of household lighting in Switzerland using a stock model. *Energy and Buildings* 2018, 158, 536-548.
- 51. Yeh, E.; Venkata, S. S.; Sumic, Z. Improved distribution system planning using computational evolution. *IEEE Transactions on Power Systems* **1996**, 11, 668-674.

- 52. Hao, R. et al. Active chromatically consistent hybrid light illumination system with natural light and light emitting diode for indoor lighting. *Energy and Buildings* **2019**, 12-23.
- 53. Molina, J.; Mesas, J. J.; Mesbahi, N. Sainz, L. LED lamp modelling for harmonic studies in distribution systems. *IET Generation, Transmission & Distribution* **2017**, 11, 1063-1071.
- Miceli, R.; Schettino, G.; Viola, F. A Novel Computational Approach for Harmonic Mitigation in PV Systems with Single-Phase Five-Level CHBMI. *Energies* 2018, 11, 21-35.
- Giglia, G.; Ala, G.; Di Piazza, M.; Giaconia, G.; Luna, M.; Vitale, G.; Zanchetta, P. Automatic EMI Filter Design for Power Electronic Converters Oriented to High Power Density. *Electronics* 2018, 7, 9-20.
- Watson, N. R. et al. Implications for Distribution Networks of High Penetration of Compact Fluorescent Lamps. *IEEE Transactions on Power Delivery* 2009, 24, 1521-1528.
- 57. Sichirollo, F.; Alonso, J. M.; Spiazzi, G. A Novel Double Integrated Buck Offline Power Supply for Solid-State Lighting Applications. *IEEE Trans. on Ind. Appl.* **2015**, 51, 1268-1276.
- Saxena, R.; Nikum, K. Comparative study of different residential illumination appliances based on power quality. IEEE 5th India International Conference on Power Electronics. Dec. 2012, pp. 1-15
- Rönnberg, S. K.; Wahlberg, M. Bollen, M. H. J. Harmonic emission before and after changing to LED and CFL — Part II: Field measurements for a hotel. Proceedings of 14th International Conf. on Harmonics and Quality of Power, Bergamo, 2010, pp. 1-6.
- Uddin, S. et al. Power quality performance of energy-efficient low-wattage LED lamps, *Meas-urement* 2013, 46, 3783–3795.
- Rönnberg, S. K.; Wahlberg, M.; Bollen, M. H. J. Harmonic emission before and after changing to LED lamps — Field measurements for an urban area. IEEE 15th ICHQP, Hong Kong, 2012, pp. 552-557.
- Shabbir, H.; Rehman, M.; Rehman, S.A.; Sheikh, S.K.; Zaffar, N. Assessment of harmonic pollution by LED lamps in power systems. Power Systems Conference, Clemson University, 11-14 March 2014.
- Rönnberg, S. K. Harmonic emission before and after changing to LED and CFL Part I: Laboratory measurements for a domestic customer. Proceedings of 14th ICHQP, Bergamo, 2010, pp. 1-7.
- Uddin, S., Shareef, H., Mohamed, A., et al. An analysis of harmonics from LED lamps. Proc. Asia-Pacific Symp. on Electromagnetic Compatibility, May 2012, pp. 837–840.
- Blanco, A.M., Parra, E.E. Effects of high penetration of CFLs and LEDs on the distribution networks. Proc. 14th IEEE Int. Conf. on Harmonics and Quality of Power, September 2010, pp. 1–5.
- Gabbar, H.A.; Zidan, A. Optimal scheduling of interconnected micro energy grids with multiple fuel options. *Sustainable Energy* 2016, 7, 80-89.
- Attanayaka, A.; Karunadasa, Hemapala, J. Estimation of state of charge for lithium-ion batteries - A Review. *AIMS Energy* 2019, 7, 186-210.
- Khadem, S.K.; Basu, M.; Conlon, M.F. Capacity enhancement and flexible operation of unified power quality conditioner in smart and microgrid network. *AIMS Energy* 2018, 6, 49-69.
- Dheer, D.K.; Doolla, S.; Bandyopadhyay, S.; Guerrero, J.M. Effect of placement of droop based generators in distribution network on small signal stability margin and network loss. *International Journal of Electrical Power and Energy Systems* 2017, 88, 108-118.
- Dong, X.; Wu, Z.; Song, G.; Ji, H.; Li, P.; Wang, C. A hybrid optimization algorithm for distribution network coordinated operation with SNOP based on simulated annealing and conic programming. IEEE Power and Energy Society General Meeting (PESGM), Boston, MA, 2016, pp. 1-5.
- Quevedo, P.M.; Contreras, J.; Mazza, A.; Chicco, G.; Porumb, R. Reliability Assessment of Microgrids With Local and Mobile Generation, Time-Dependent Profiles, and Intraday Reconfiguration. *IEEE Transactions on Industry Applications* 2018, 54, pp. 61-72.

- Mansoor, A.; Grady, W.M.; Chowdhury, A.H.; Samotyi, M.J. An investigation of harmonics attenuation and diversity among distributed single-phase power electronic loads. *IEEE Transactions on Power Delivery* 1995, 10, 467-473.
- Molina, J.; Sainz, L. Compact Fluorescent Lamp Modeling for Large-Scale Harmonic Penetration Studies. *IEEE Transactions on Power Delivery* 2015, 30, 1523-1531.
- Nassif, A. B.; Xu, W. Characterizing the Harmonic Attenuation Effect of Compact Fluorescent Lamps. *IEEE Transactions on Power Delivery* 2009, 24, 1748-1749.
- Korovesis, P. N. et al. Influence of large-scale installation of energy saving lamps on the line Voltage distortion of a weak network supplied by photovoltaic station. *IEEE Transactions on Power Delivery* 2004, 19, 1787-1793.
- Dwyer, R. et al. Evaluation of harmonic impacts from compact fluorescent lights on distribution systems. *IEEE Transactions on Power Systems* 1995, 10, 4, 1772-1779.
- Nassif, A. B.; Yong, J.; Xu, W.; Measurement-based approach for constructing harmonic models of electronic home appliances, *IET Generation*, *Transmission & Distribution*, 2010, 4, 363-375.
- Braga, D. S.; Jota, P. R. S.; Prediction of total harmonic distortion based on harmonic modeling of nonlinear loads using measured data for parameter estimation, 17th International Conference on Harmonics and Quality of Power (ICHQP), Belo Horizonte, Brasil, 2016, 454-459.
- Yong, J.; Chen, L.; Nassif, A. B.; Xu, W.; A Frequency-Domain Harmonic Model for Compact Fluorescent Lamps, *IEEE Transactions on Power Delivery* 2010, 25, 1182-1189.
- Molina, J.; Sainz, L.; Model of Electronic Ballast Compact Fluorescent Lamps, *IEEE Transac*tions on Power Delivery 2014, 29, 1363-1371.
- Molina, J.; Mesas, J. J.; Sainz, L.; Parameter estimation procedure for the equivalent circuit model of compact fluorescent lamps, *Electric Power Systems Research* 2014, 116, 128-135.
- Souza, R. R. N. et al.; Nonlinear Loads Parameters Estimation and Modeling, IEEE International Symposium on Industrial Electronics, Vigo, Spain, 2007, 937-942.
- Mesas, J.; Sainz, L.; Molina, J.; Parameter Estimation Procedure for Models of Single-Phase Uncontrolled Rectifiers, *IEEE Transactions on Power Delivery* 2011, 26, 1911-1919.
- Ringel, M.; Schlomann, B.; Krail, M.; Rohde, C. Towards a green economy in Germany? The role of energy efficiency policies. *Appl. Energy* 2016, 179, 1293–1303.
- Erickson, R.; Madigan, M.; Singer, S. Design of a simple high-power-factor rectifier based on the flyback converter. In Proceedings of the Fifth Annual Proceedings on Applied Power Electronics Conference and Exposition, Los Angeles, CA, USA, 11–16 March 1990; pp. 792–801.
- Huber, L.; Irving, B.T.; Jovanovic, M.M. Line current distortions of DCM/CCM boundary boost PFC converter. In Proceedings of the Twenty-Third Annual IEEE Applied Power Electronics Conference and Exposition, Austin, TX, USA, 24–28 February 2008; pp. 702–708.
- Kim, J.W.; Choi, S.M.; Kim, K.T. Variable On-time Control of the Critical Conduction Mode Boost Power Factor Correction Converter to Improve Zero-crossing Distortion. In Proceedings of the International Conference on Power Electronics and Drives Systems, Kuala Lumpur, Malaysia, December 2005; pp. 1542–1546.
- Guo, Z.; Ren, X.; Wu, Y.; Zhang, Z.; Chen, Q. A novel simplified variable on-time method for CRM boost PFC converter. In Proceedings of the IEEE Applied Power Electronics Conference and Exposition (APEC), Tampa, FL, USA, 26–30 March 2017; pp. 1778–1784.
- Di Mauro, S.; Raciti, A.; Rizzo, S. A.; Susinni, G. LED lamp PSpice circuit emulating first harmonic current absorption as a function of the main voltage. In Proceedings of the conference SPEEDAM, Amalfi, Italy, 2018, 1101-1106.
- Raciti, A.; Rizzo, S. A.; Susinni, G. Steady-State Electrical Modeling of LED and CF Bulb Lamps Under Variable Voltage on the Main, EEEIC / I&CPS Europe, Palermo, Italy, 2018, 1-6.
- Alobeidli, K.; Syed, M.; El Moursi, M.; Zeineldin, H. Novel Coordinated Voltage Control for Hybrid Micro-Grid with islanding capability, Power & Energy Society General Meeting, Denver, CO, 2015, 1-1.

- Pasha, A. M.; Zeineldin, H.; Saad, A.; Sumaiti, A.; El Moursi, M. S.; El Sadaany, E. F. Conservation Voltage Reduction for Autonomous Microgrids Based on V–I Droop Characteristics, *IEEE Transactions on Sustainable Energy* 2017, 8, 1076-1085.
- Tech, H.; Leber, D.; Schwarzmeier, A. A novel approach to efficient inrush current limitation for LED power applications, PCIM Europe 2014; International Exhibition and Conference for Power Electronics, Intelligent Motion, Renewable Energy and Energy Management, Nuremberg, Germany, 2014, pp. 1-8.
- Blaabjerg, F.; Dragicevic, T.; Davari, P. Applications of Power Electronics. *Electronics* 2019, 8, 465.
- Abadi, I.; Imron, C.; Bachrowi, M.M.; Fitriyanah, D.N. Design and implementation of battery charging system on solar tracker based standalone PV using fuzzy modified particle swarm optimization. *AIMS Energy* 2020, 8, 142-155.
- García Salvador, R.M.; Novas, N.; Alcayde, A.; El Khaled, D.; Montoya, F.G. Electronics and Its Worldwide Research. *Electronics* 2020, 9, 977-987.
- Musumeci, S.; Bojoi, R.; Armando, E.; Borlo, S.; Mandrile, F. Three-Legs Interleaved Boost Power Factor Corrector for High-Power LED Lighting Application. *Energies* 2020 13, 17-28.
- Yilmaz M.; Krein, P.T. Review of the Impact of Vehicle-to-Grid Technologies on Distribution Systems and Utility Interfaces. *IEEE Transactions on Power Electronics* 2013, 28, 5673-5689.
- Konara, K. M. S. Y.; Kolhe, M.L.; Sharma. A. Power dispatching techniques as a finite state machine for a standalone photovoltaic system with a hybrid energy storage. *AIMS Energy* 2020, 8, 214-230.
- Faraci, G.; Raciti, A.; Rizzo, S.A.; Schembra, G. Green Wireless Power Transfer System for a Drone Fleet Managed by Reinforcement Learning in Smart Industry, *Applied Energy* 2020, 259-276.
- Ding, Y.; Loh, P. C.; Tan, K. K.; Wang, P.; Gao, F. Reliability evaluation of three-level inverters. In Proc. 25th Annu. IEEE Appl. Power Electron. Conf. Expo., 21–25 Feb. 2010; pp. 1555–1560.
- 102. De Leon-Aldaco, S. E.; Calleja, H.; Chan, F.; Jimenez-Grajales, H. R. Effect of the mission profile on the reliability of a power converter aimed at photovoltaic applications-a case study. *IEEE Trans. Power Electron.* 2013, 28, 2998–3007.
- Alam, M. K.; Khan, F. H.; Reliability analysis and performance degradation of a boost converter. In Proc. IEEE Energy Convers. Congr. Expo., 15–19 Sep., 2013, pp. 5592–5597.
- Tajfar, A.; Mazumder, S. K.; A fault-tolerant switching scheme for a photovoltaic highfrequency-link inverter, In Proc. 27th Annu. IEEE Appl. Power Electron. Conf. Expo., 5–9 Feb., 2012, pp. 2087–2094.
- 105. Madhukar Rao, A.; Umesh, B. S.; Sivakumar, K.; A fault tolerant dual inverter configuration for islanded mode photovoltaic generation system. In Proc. 1st Int. Future Energy Electron. Conf., 3–6 Nov., 2013, pp. 816–821.
- 106. Haumann, S.; Becker, M.; Rudzki, J.; Eisele, R.; Osterwald, F.; Novel bonding and joining technology for power electronics - enabler for improved lifetime, reliability, cost and power density. In Proc. 28th Annu. IEEE Appl. Power Electron. Conf. Expo., 17–21 Mar., 2013, pp. 622–626.
- 107. Krebs, T.; Duch, S.; Schmitt, W.; Kotter, S.; Prenosil, P.; Thomas, S.; A breakthrough in power electronics reliability–new die attach and wire bonding materials. In Proc. IEEE 63rd Electron. Compon. Technol. Conf., 28–31 May, 2013, pp. 1746–1752.
- Mohagheghi, S.; Harley, R. G.; Habetler, T. G.; Divan, D. Condition monitoring of power electronic circuits using artificial neural networks, *IEEE Trans. Power Electron.* 2009, 24, 2363–2367.
- 109. Xiang, D.; Ran, L.; Tavner, P.; Bryant, A.; Yang, S.; Mawby, P. Monitoring solder fatigue in a power module using case-above-ambient temperature rise. *IEEE Trans. Ind. Appl.* **2011**, 47, 2578–2591.
- 110. Li, D.; Li, X. Study of degradation in switching mode power supply based on the theory of PoF. In Proc. Int. Conf. Comput. Sci. Serv. Syst., 11–13 Aug., 2012, pp. 1976–1980.

- Levikari, S.; Kärkkäinen, T. J.; Andersson, C.; Tammminen J.; Silventoinen, P. Acoustic Detection of Cracks and Delamination in Multilayer Ceramic Capacitors. *IEEE Transactions on Industry Applications* 2019, 55, 1787-1794.
- 112. Cheraghi, M.; Karimi, M.; Booin, M. B. An investigation on acoustic noise emitted by induction motors due to magnetic sources. 9th Annual Power Electronics, Drives Systems and Technologies Conference (PEDSTC), Tehran, 2018, pp. 104-109.
- Smulko, J.; Józwiak, k.; Olesz, M.; Hasse, L. Acoustic emission for detecting deterioration of capacitors under aging, *Microelectronics Reliability* 2011, 51, 624-627.
- 114. Kärkkäinen, T. J.; Talvitie, J. P.; Kuisma, M.; Silventoinen, P.; Mengotti, E. Measurement challenges in acoustic emission research of semiconductors. 17th European Conference on Power Electronics and Applications (EPE'15 ECCE-Europe), Geneva, 2015, pp. 1-6.
- 115. Kärkkäinen, T. J.; Talvitie, J. P.; Kuisma, M.; Silventoinen, P.; Mengotti, E. Acoustic emission caused by the failure of a power transistor, IEEE Applied Power Electronics Conference and Exposition (APEC), Charlotte, NC, 2015, pp. 2481-2484.
- 116. Kärkkäinen, T. J.; Talvitie, J. P.; Ikonen, O.; Kuisma, M.; Silventoinen, P.; Mengotti, E. Sounds from semiconductors - Acoustic emission experiment with a power module. 16th European Conference on Power Electronics and Applications, Lappeenranta, 2014, pp. 1-6.
- Kärkkäinen, T. J. et al. Acoustic Emission in Power Semiconductor Modules—First Observations. In IEEE Transactions on Power Electronics 2014, 29, 6081-6086.
- Müller, S.; Drechsler, C.; Heinkel, U.; Herold, C. Acoustic emission for state-of-health determination in power modules. 13th International Multi-Conference on Systems, Signals & Devices (SSD), Leipzig, 2016, pp. 468-471.
- 119. Davari, P.; Kristensen, O.; Iannuzzo, F. Investigation of acoustic emission as a non-invasive method for detection of power semiconductor aging, *Microelectronics Reliability* **2018**, 88–90, 545-549.
- Oh, H.; Han, B.; McCluskey, P.; Han, C.; B. D. Youn. Physics-of failure, condition monitoring, and prognostics of insulated gate bipolar transistor modules: A review. *IEEE Trans. Power Electron.* 2015, 30, 2413–2426.
- Luo, H.; Wang, X.; Zhu, C.; Li, W.; He, X. Investigation and emulation of junction temperature for high-power IGBT modules considering grid codes. *IEEE J. Emerg. Sel. Topics Power Electron*. 2018, 6, 930–940.
- Scheuermann, U.; Schuler, S. Power cycling results for different control strategies. *Microelectron. Reliability* 2010, 50, 1203-1209.
- 123. Zarebski, J.; Gorecki, K. The electro thermal large-signal model of power MOS transistors for SPICE. *IEEE Trans. Power Electron.* **2010**, 25, pp. 1265-1274.
- 124. Breglio, G.; Irace, A.; Spirito, P.; Letor, R.; Russo, S. Fast transient infrared thermal analysis of smart Power MOSFETS in permanent short circuit operation. In Proc. 18th Int. Symp. Power Semicond. Devices IC's, 2006, pp. 1-4.
- 125. Hunger, T.; Schilling, O. Numerical investigation on thermal crosstalk of silicon dies in high voltage IGBT modules. In Proc. PCIM, 2008.
- Hillkirk, L.-M.; Dynamic surface temperature measurements in SiC epitaxial power diodes performed under single-pulse self-heating conditions. *Solid-State Electron.* 2004, 48, 2181-2189.
- 127. Breglio, G.; Rinaldi, N.; Spirito, P. Thermal mapping and 3D numerical simulation of new cellular power MOS affected by electro-thermal instability. *Microelectronics Journal*, 2000, 31, 741-746.
- 128. Spirito, P.; Breglio, G.; D'Alessandro, V.; Rinaldi, N. Thermal instabilities in high current power MOS devices: Experimental evidence electro-thermal simulations and analytical modeling. In Proc. 23rd Int. Conf. Microelectron., May 12-15 2002, pp. 23-30.
- 129. Thalhammer, R.-K.; Wachutka, G.; Physically rigorous modeling of internal laser-probing techniques for micro structured semiconductor devices. *IEEE Trans. Comput.-Aided Design Integr. Circuits Syst.* 2004, 23, 60-70.

- 130. Furbock, C.; Thalhammer, R.; Litzenberger, M.; Seliger, N.; Pogany, D.; Gornik, E.; Wachutka, G. A differential backside laser probing technique for the investigation of the lateral temperature distribution in power devices. In Proc. 11th Int. Symp. Power Semicond. Devices ICs, May 26-28 1999, pp. 193-196.
- Werber, D.; Wachutka, G.; Interpretation of laser absorption measurements on 4 H-SiC bipolar diodes by numerical simulation. In Proc. Int. Conf. Simul. Semicond. Process. Devices, Sep. 9-11 2008. pp. 89-92.
- 132. Hamidi, A.; Coquery, G.; Lallemand, R.; Vales, P.; Dorkel, J. M. Temperature measurements and thermal modeling of high power IGBT multichip modules for reliability investigations in traction applications. *Microelectron. Reliabil.* **1998**, 38, 1353-1359.
- Carubelli, S.; Khatir, Z.; Experimental validation of a thermal modelling method dedicated to multichip power modules in operating conditions. *Microelectron. J.* 2003, 34, 1143-1151.
- Khatir, Z.; Carubelli, S.; Lecoq, F. Real-time computation of thermal constraints in multichip power electronic devices. *IEEE Transactions on Components and Packaging Technologies* 2004, 27, 337-344.
- 135. Schmidt, R.; Scheuermann, U. Using the chip as a temperature sensor—The influence of steep lateral temperature gradients on the Vce(T)-measurement. In Proc. 13th Eur. Conf. Power Electron. Appl., Sep. 8-10 2009, pp. 1-9.
- 136. Brckner, T.; Bernet, S. Estimation and measurement of junction temperatures in a three-level voltage source converter. *IEEE Trans. Power Electron.* **2007**, 22, 3-12.
- Mermet-Guyennet, M.; Perpina, X.; Piton, M. Revisiting power cycling test for better life-time prediction in traction. *Microelectron. Reliabil.* 2007, 47, 1690-1695.
- 138. Avenas, Y.; Dupont, L. Comparison of junction temperature evaluations in a power IGBTs module using an IR camera and three thermo-sensitive electrical parameters. Presented at the Appl. Power Electron. Conf. Expo., Feb. 5-9, 2012.
- Dupont, L.; Avenas, Y.; Jeannin, P.-O. Comparison of junction temperature evaluations in a power IGBT module using an IR camera and three thermosensitive electrical parameters. *IEEE Transactions on Industry Applications* 2013, 49, 1599-1608.
- 140. Round, H.J. A note on carborundum, *Electrical World* 1907, 309.
- Winkler, J.; Homoth, J.; Kallfass, I. Utilization of parasitic luminescence from power semiconductor devices for current sensing. In Proc. PCIM Eur., Nuremberg, Germany, May 2018, pp. 1–8.
- Ceccarelli, L.; Luo, H.; Iannuzzo, F. Investigating SiC MOSFET body diode's light emission as temperature-sensitive electrical parameter. *Microelectronics Reliability* 2018, 88–90, 627-630.
- 143. Li, C.; Luo, H.; Li, C.; Li, W.; Yang, H.; He, X. Online Junction Temperature Extraction of SiC Power MOSFET with Temperature Sensitive Optic Parameter (TSOP) Approach. *IEEE Transactions on Power Electronics* 2019, 34, 10143-10152.
- 144. Winkler, J.; Homoth, J.; Kallfass, I.; Electroluminescence-Based Junction Temperature Measurement Approach for SiC Power MOSFETs. *IEEE Transactions on Power Electronics* **2020**, 35, 2990-2998.
- 145. Parsley, M. The use of thermochromics liquid crystals in research applications, thermal mapping and non-destructive testing. In Proceedings, Seventh IEEE Semiconductor Thermal Measurement and Management Symposium, Phoenix, AZ, USA, 1991, pp. 53-58.
- Brekel, W.; Duetemeyer, T.; Puk, G.; Schilling, O. Time Resolved In Situ Tvj Measurements of 6.5 kV IGBTs during Inverter Operation. *Bodo's Power* 2010.
- 147. Salem, T. E.; Ibitayo, D.; Geil, B. R. A Technique for Die Surface Temperature Measurement of High-Voltage Power Electronic Components using Coated Thermocouple Probes. IEEE Instrumentation and Measurement Technology Conference Proceedings, Sorrento, 2006, pp. 651-654.

- 148. Sathik, M. H. M.; Prasanth, S.; Sasongko, F.; Padmanabhan, S. K.; Pou, J.; Simanjorang, R. Online junction temperature for off-the-shelf power converters. IEEE Applied Power Electronics Conference and Exposition (APEC), San Antonio, TX, 2018, pp. 2769-2774.
- 149. Ikonen, M.; Häsä, H.; Rauma, K. Silventoinen, P. A system for thermal model verification of a power switch. 37th IEEE Power Electronics Specialists Conference, Jeju, 2006, pp. 1-4.
- Bonsbaine, A.; Trigkidis, G. Benamrouche, N. An integrated electro-thermal model of IGBT devices (experimental validation). 44th International Universities Power Engineering Conference (UPEC), Glasgow, 2009, pp. 1-5.
- Du, B.; Hudgins, J. L.; Santi, E.; Bryant, A. T.; Palmer, P. R.; Mantooth, H. A. Transient Electrothermal Simulation of Power Semiconductor Devices. *IEEE Transactions on Power Electronics* 2010, 25, 237-248.
- 152. Claassen, A. Shaukatullah, H. Comparison of diodes and resistors for measuring chip temperature during thermal characterization of electronic packages using thermal test chips. Thirteenth Annual IEEE. Semiconductor Thermal Measurement and Management Symposium, Austin, TX, USA, 1997, pp. 198-209.
- Motto, E. R.; Donlon, J. F. IGBT module with user accessible on-chip current and temperature sensors. Twenty-Seventh Annual IEEE Applied Power Electronics Conference and Exposition (APEC), Orlando, FL, 2012, pp. 176-181.
- 154. Ka et al., Instrumented chip dedicated to semiconductor temperature measurements in power electronic converters. IEEE Energy Conversion Congress and Exposition (ECCE), Milwaukee, WI, 2016, pp. 1-8.
- 155. Y. Zhou et al., Dynamic junction temperature estimation via built-in negative thermal coefficient (NTC) thermistor in high power IGBT modules. IEEE Applied Power Electronics Conference and Exposition (APEC), Tampa, FL, 2017, pp. 772-775.
- 156. Baker, N.; Iannuzzo, F.; Beczkowski, S. Kristensen, P. K. Proof-of-Concept for a Kelvin-Emitter On-Chip Temperature Sensor for Power Semiconductors. 21st European Conference on Power Electronics and Applications (EPE '19 ECCE Europe), Genova, Italy, 2019, pp. P.1-P.8.
- 157. Kempiak, C.; Lindemann, A.; Thal, E. Idaka, S. Investigation of the usage of a chip integrated sensor to determine junction temperature during power cycling tests. CIPS 2018; 10th International Conference on Integrated Power Electronics Systems, Stuttgart, Germany, 2018, pp. 1-6.
- 158. Baker, N.; Iannuzzo, F. Li, H. Impact of Kelvin-Source Resistors on Current Sharing and Failure Detection in Multichip Power Modules. 20th European Conference on Power Electronics and Applications (EPE'18 ECCE Europe), Riga, 2018, pp. 1-7.
- Kempiak, C.; Lindemann, A.; Idaka, S.; Thal, E. Investigation of an Integrated Sensor to Determine Junction Temperature of SiC MOSFETs During Power Cycling Tests. 10th International Conference on Power Electronics and ECCE Asia (ICPE 2019 - ECCE Asia), Busan, Korea (South), 2019, pp. 3084-3089.
- Liu, P.; Chen, C.; Zhang, X.; Huang, S. Online junction temperature estimation method for SiC modules with built-in NTC sensor. *CPSS Transactions on Power Electronics and Applications* 2019, 4, 94-99.
- Liu, P.; Zhang, X.; Yin, S.; Tu, C.; Huang, S. Simplified Junction Temperature Estimation using Integrated NTC Sensor for SiC Modules. IEEE International Power Electronics and Application Conference and Exposition (PEAC), Shenzhen, 2018, pp. 1-4.
- O' Donnell, K. P; Chen, X. Temperature dependence of semiconductor band gaps. *Applied Physics Letters* 1991, 58, 2924-2926.
- Baliga, B. J. Fundamentals of power semiconductor devices. Springer International Publishing, 2013.
- Blackburn, D.-L.; An electrical technique for the measurement of the peak junction temperature of power transistors. In Proc. 13th Annu. Reliabil. Phys. Symp., Las Vegas, NV, Apr. 1–3 1975, pp. 143–150.

- 165. Farjah, E.; Perret, R. Application and analysis of thermosensitive parameters in the case of hybrid power modules. In Proc. IEEE Ind. Appl. Soc. Annu. Meet., Orlando, FL, Oct. 8–12 1994, pp. 1284–1289.
- Choi, U.; Blaabjerg, F.; Iannuzzo, F.; Jørgensen, S. Junction temperature estimation method for a 600V, 30A IGBT module during converter operation. *Microelectronics Reliability* 2015, 55, 2022-2026.
- 167. Oettinger, F. F.; Blackburn, D. L.; Rubin, S. Thermal characterization of power transistors. *IEEE Transactions on electron devices* **1976**, 23, 831-838.
- Held, M.; Jacob, P.; Nicoletti, G.; Scacco, P.; Poech, M.-H. Fast power cycling test for insulated gate bipolar transistor modules in traction application. *International journal of electronics* 1999, 86, 1193-1204.
- Blackburn, D. L. A review of thermal characterization of power transistors. In Proc. Semiconductor Thermal and Temperature Measurement Symposium, SEMI-THERM IV., Fourth Annual IEEE, Feb. 1988, pp. 1-7.
- Khatir, Z.; Dupont, L.; Ibrahim, A. Investigations on junction temperature estimation based on junction voltage measurements. *Microelectron. Reliabil.* 2010, 50, 1506–1510.
- 171. Barnes, C.-M.; Tuma, P.-E. Practical considerations relating to immersion cooling of power electronics in traction systems. *IEEE Trans. Power Electron.* **2010**, 25, 2478–2485.
- 172. Sofia, J.-W.; Electrical measurement using semiconductors. Electron. Cool. 1997, 3, pp. 22–25.
- 173. Nowak, M.; Rabkowski, J.; Barlik, R. Measurement of temperature sensitive parameter characteristics of semiconductor silicon and silicon carbide power devices. In Proc. 13th Power Electron. Motion Control Conf., Poznan, Poland, Sep. 1–3 2008, pp. 84–87.
- Blackburn, D.-L.; Berning, D.-W. Power MOSFET temperature measurements. In Proc. Annu. Power Electron. Spec. Conf., Cambridge, MA, Jun. 14–17 1982, pp. 400–407.
- Jakopovic, Z.; Bencic, Z.; Kolonic, F.; Important properties of transient thermal impedance for MOS-gated power semiconductors. In Proc. IEEE Int. Symp. Ind. Electron., Bled, Slovenia, Jul. 12–16 1999, pp. 574–578.
- 176. Perpiñà, X.; Serviere, J.-F.; Saiz, J.; Barlini, D.; Mermet-Guyennet, M.; Millán, J. Temperature measurement on series resistance and devices in power packs based on on-state voltage drop monitoring at high current. *Microelectronics Reliability* **2006**, 46, 1834-1839.
- Meysenc, L.; Saludjian, L.; Bricard, A.; Rael, S.; Schaeffer, C. A high heat flux IGBT micro exchanger setup. *IEEE Trans. Compon. Packag. Manuf. Technol. A* 1997, 20, 334–341.
- 178. Cova, P.; Ciappa, M.; Franceschini, G.; Malberti, P.; Fantini, F. Thermal characterization of IGBT power modules. *Microelectron. Reliabil.* **1997**, 37, 1731–1734.
- 179. Ammous, A.; Allard, B.; Morel, H. Transient temperature measurements and modeling of IGBT's under short circuit. *IEEE Trans. Power Electron.* **1998**, 13, 12–25.
- Duong, S.; Rael, S.; Schaeffer, C.; De Palma, J. F. Short circuit behaviour for PT and NPT IGBT devices—Protection against explosion of the case by fuse. In Proc. Eur. Conf. Power Electron. Appl., Seville, Spain, Sep. 19–21 1995, pp. 249–254.
- Forest, F.; Rashed, A.; Huselstein, J.-J.; Martiré, T.; Enrici, P. Fast power cycling protocols implemented in an automated test bench dedicated to IGBT module ageing. *Microelectronics Reliability* 2015, 55, 81-92.
- Koenig, A.; Plum, T.; Fidler, P.; De Doncker, R.-W. On-line junction temperature measurement of CoolMOS devices. In Proc. 7th Int. Conf. Power Electron. Drive Syst., 2007, pp. 90–95.
- Kim, Y.-S; Sul, S.-K.; On-line estimation of IGBT junction temperature using on-state voltage drop. In Proc. 1998 IEEE Ind. Appl. Conf., St Louis, MO, Oct. 12–15 1998, pp. 853–859.
- 184. Dupont, L.; Avenas, Y. Evaluation of thermo-sensitive electrical parameters based on the forward voltage for on-line chip temperature measurements of IGBT devices. In Energy Conversion Congress and Exposition (ECCE-2014), pp. 4028-4035.

- Ghimire, P.; Pedersen, K. B.; Trintis, I.; Munk-Nielsen, S. Online chip temperature monitoring using Vce-load current and IR thermography. In Proc. Energy Conversion Congress and Exposition (ECCE-2015), pp. 6602- 6609.
- Castellazzi, A.; Wachutka, G. Low-voltage Power MOSFETs used as dissipative elements: Electrothermal analysis and characterization. In Proc. 37th IEEE Power Electron. Spec. Conf., Jeju, Korea, Jun. 18–22 2006, pp. 1–7.
- 187. Ayadi, M.; Fakhfakh, M.; Moez, G.; Neji, R. Electro-Thermal Simulation of a Three Phase Inverter with Cooling System. *Journal of Modelling and Simulation of Systems* **2010**, 1, 163-170.
- 188. Baliga, J. Power Semiconductor Devices. International Thomson Publishing, Boston, 1996.
- 189. Chen, H.; Pickert, V.; Atkinson, D. J.; Pritchard, L. S. On-line monitoring of the MOSFET device junction temperature by computation of the threshold voltage. In Proc. 3rd IET Int. Conf. Power Electron. Mach. Drives, Dublin, Ireland, Apr. 4–6 2006, pp. 440–444.
- Cao, X.; Wang, T.; Lu, G.-Q.; Ngo, K. D. T. Characterization of lead-free solder and sintered nano-silver die-attach layers using thermal impedance. In Proc. Int. Power Electron. Conf., Sapporo, Japan, Jun. 21–24 2010, pp. 546–552.
- Huang, X. Y.; Lu, C.; Xie, X.; Fan, Y.; Zhang, J.; Meng, X. A study of test system for thermal resistance of IGBT. In Proc. 2010 Asia Pacif. Conf. Postgraduate Res. Microelectron. Electron. (PrimeAsia), Shanghai, China, Sep. 22–24 2010, pp. 312–315.
- 192. Strauss, B.; Lindemann, A. Indirect measurement of junction temperature for condition monitoring of power semiconductor devices during operation. In Proceedings of PCIM Europe, International Exhibition and Conference for Power Electronics, Intelligent Motion, Renewable Energy and Energy Management, Nuremberg, Germany, 2015, pp. 1-6.
- 193. Butron Ccoa, J. A.; Strauss, B.; Mitic, G.; Lindemann, A. Investigation of Temperature Sensitive Electrical Parameters for Power Semiconductors (IGBT) in Real-Time Applications. PCIM Europe; International Exhibition and Conference for Power Electronics, Intelligent Motion, Renewable Energy and Energy Management, Nuremberg, Germany, 2014, pp. 1-9.
- 194. Berning, D.; Reichl, J.; Hefner, A.; Hernandez, M.; Ellenwood, C.; Lai, J.-S. High speed IGBT module transient thermal response measurements for model validation. In Proc. 38th IAS Annu. Meet. Ind. Appl. Conf., Salt Lake City, UT, Oct. 12–16 2003, pp. 1826–1832.
- 195. Sundaramoorthy, V.; Bianda, E.; Bloch, R.; Nistor, I.; Knapp, G. Heinemann, A. Online estimation of IGBT junction temperature (Tj) using gate-emitter voltage (Vge) at turn-off. 15th European Conference on Power Electronics and Applications (EPE), Lille, 2013, pp. 1-10.
- Kuhn, H.; Mertens, A.; On-line junction temperature measurement of IGBTs based on temperature sensitive electrical parameters. In Proc.13th Eur. Conf. Power Electron. Appl., Barcelona, Spain, Sep. 8–10 2009, pp. 1–10.
- 197. Barlini, D.; Ciappa, M.; Castellazzi, A.; Mermet-Guyennet, M.; Fichtner, W. New technique for the measurement of the static and of the transient junction temperature in IGBT devices under operating conditions. *Microelectron. Reliabil.* 2006, 46, 1772–1777.
- 198. Du, M.; Xin, J.; Wang, H.; Ouyang, Z.; Wei, K. Estimating Junction Temperature of SiC MOSFET Using Its Drain Current During Turn-On Transient. *IEEE Transactions on Electron Devices* 2020, 67, 1911-1918.
- Li, L.; Ning, P.; Wen, X.; Li, Y.; Ge, Q.; Zhang, D. et al. A turn-off delay time measurement and junction temperature estimation method for IGBT. Applied Power Electronics Conference and Exposition (APEC-2017), pp. 2290-2296.
- 200. Zhang, Z.; Wang, F.; Costinett, D. J.; Tolbert, L. M.; Blalock, B. J.; Wu, X. Online junction temperature monitoring using turn-off delay time for silicon carbide power devices. IEEE Energy Conversion Congress and Exposition (ECCE), Milwaukee, WI, 2016, pp. 1-7.
- Luo, H.; Chen, Y.; Sun, P.; Li, W.; He, X. Junction Temperature Extraction Approach with Turn-Off Delay Time for High-Voltage High-Power IGBT Modules. *IEEE Transactions on Pow*er Electronics 2016, 31, 5122-5132.

- Zhang, J.; Du, M.; Jing, L.; Wei, K.; Hurley, W. G. IGBT Junction Temperature Measurements: Inclusive of Dynamic Thermal Parameters. *IEEE Transactions on Device and Materials Reliability* 2019, 19, 333-340.
- Luo, H.; Chen, Y.; Li, W.; He, X. Online high-power P-i-N diode junction temperature extraction with reverse recovery fall storage charge. *IEEE Trans. Power Electron.* 2016, 580-618.
- Luo, H.; Li, W.; He, X. Online high power Pin diode chip temperature extraction and prediction method with maximum recovery current di/dt. *IEEE Trans. Power Electron.* 2015, 30, 2395– 2404.
- Xiang, D.; Ran, L.; Tavner, P.; Yang, S. Condition monitoring power module solder fatigue using inverter harmonic identification. *IEEE Trans. Power Electron.* 2012, 27, 235–247.
- Bryant, A.; Yang, S.; Mawby, P.; Xiang, D. Investigation into IGBT dV/dt during turn-off and its temperature dependence. *IEEE Trans. Power Electron.* 2011, 26, 3019–3031.
- 207. Chen, Y.; Luo, H.; Li, W.; He, X.; Iannuzzo, F.; Blaabjerg, F. Analytical and Experimental Investigation on A Dynamic Thermo-Sensitive Electrical Parameter with Maximum \$dI_{C}/dt\$ During Turn-off for High Power Trench Gate/Field-Stop IGBT Modules. *IEEE Transactions on Power Electronics* 2017, 32, 6394-6404.
- Zheng, R.; Haoge, X., Chengmin, L.; Wuhua L.; Xiangning, H.; Luo, H.; and Li, D.; Online Aging Parameter Extraction with Induced Voltage *dveL/dt* between Kelvin and Power Emitter in Turn-off Progress for IGBT Modules. IEEE Energy Conversion Congress and Exposition (ECCE), 2018, pp. 362-366.
- Zeng, Z.; Li, X.; Zhang, X.; Cao, L. Comparative Evaluation of Kelvin Connection for Current Sharing of Multi-Chip Power Modules. IEEE Energy Conversion Congress and Exposition (ECCE), 2018, pp. 4664-4670.
- 210. Chen, H. L.; Li, W.; He, X. A thermo-sensitive electrical parameter with maximum dIC/dt during turn-off for high power trench/field-stop IGBT modules. In Proc. 31th Annu. IEEE Appl. Power Electron. Conf. Expo., Long Beach, CA, USA, May 2016, pp. 499–504.
- Mautry, P. G.; Trager, J. Investigation of self-heating in VLSI and ULSI MOSFETs. In Proc. Int. Conf. Microelectron., Mar. 5–7 1990, pp. 221–226.
- Baker, N.; Munk-Nielsen, S.; Liserre, M.; Iannuzzo, F. Online junction temperature measurement via internal gate resistance during turn-on. In Proc. 16th Eur. Conf. Power Electron. Appl., Aug. 26–28 2014, pp. 1–10.
- Lembeye, Y.; Schanen, J. L.; Keradec, J. P.; Experimental characterization of insulated gate power components: Capacitive aspects. In Proc. 32nd IAS Annu. Meeting Ind. Appl. Conf., Oct. 5–9 1997, pp. 983–988.
- Zhou, S.; Zhou, L.; Sun, P. Monitoring potential defects in an IGBT module based on dynamic changes of the gate current. *IEEE Trans. Power Electron.* 2013, 28, 1479–1487.
- Baker, N.; Munk-Nielsen, S.; Iannuzzo, F.; Liserre, M. IGBT Junction Temperature Measurement via Peak Gate Current. *IEEE Transactions on Power Electronics* 2016, 31, 3784-3793.
- Baker, N.; Dupont, L.; Munk-Nielsen, S.; Iannuzzo, F.; Liserre, M. IR Camera Validation of IGBT Junction Temperature Measurement via Peak Gate Current. *IEEE Transactions on Power Electronics* 2017, 32, 3099-3111.
- 217. Dilettoso, E.; Rizzo, S.A.; Salerno N. A weakly Pareto compliant quality indicator. *Mathematical and Computational Applications* 2017, 22, 25-42.
- Gonzalez, J. O.; Alatise, O.; Hu, J.; Ran, L.; Mawby, P. A. An Investigation of Temperature-Sensitive Electrical Parameters for SiC Power MOSFETs. *IEEE Transactions on Power Electronics* 2017, 32, 7954-7966.
- Luo, H.; Li, W.; Iannuzzo, F.; He, X.; Blaabjerg, F. A via collector-side thermo-sensitive electrical parameters through emitter stray inductance in high-power IGBT modules, *IEEE Trans. Ind. Electron.* 2018, 65, 4724–4738.

- Avenas, Y.; Dupont, L.; Khatir, Z. Temperature Measurement of Power Semiconductor Devices by Thermo-Sensitive Electrical Parameters—A Review. *IEEE Transactions on Power Electronics* 2012, 27, 3081-3092.
- 221. Li, C.; Lu, Z.; Wu, H.; Li, W.; He, X.; Li, S. Junction Temperature Measurement Based on Electroluminescence Effect in Body Diode of SiC Power MOSFET. IEEE Applied Power Electronics Conference and Exposition (APEC), Anaheim, CA, USA, 2019, 338-343.
- 222. Kimoto, T. Material science and device physics in SiC technology for high-voltage power devices, *Jpn. J. Appl. Phys.* 2015, 54, 40-103.
- 223. Neamen, D. A. Semiconductor Physics and Devices: Basic Principles, 3rd ed. New York, NY, USA: McGraw-Hill, 2003.
- 224. Lasance, C.J.M.; Poppe, A. Thermal Management for LED Applications, Vol. 2 Springer New York, New York, NY, 2014.
- 225. Shalish, I.; Altfeder, I.B.; Narayanamurti, V. Observations of conduction-band structure of 4H and 6 H SiC. *Phys. Rev* 2002, 65.
- 226. Yin, S.; Liu, Y.; Liu, Y.; Tseng, K. J.; Pou, J.; Simanjorang, R. Comparison of SiC voltage source inverters using synchronous rectification and freewheeling diode, *IEEE Trans. Ind. Electron.* 2018, 65, 1051-1061.
- 227. Vishay VBPW34S Silicon PIN Photodiode datasheet https://www.vishay.com/docs/81128/vbpw34s.pdf.
- 228. SEMIKRON SK45MH120TSCp SiC power module datasheet https://www.semikron.com/products/product-classes/sic/full-sic/detail/sk45mh120tscp-24920411.html.
- 229. BROADCOM HCPL-3120 gate driver optocoupler, https://www.broadcom.com/products/optocouplers/industrial-plastic/isolated-gate-driveoptocouplers/gate-drives/hcpl-3120.